

On some periodic solutions of discrete vibro-impact systems with a unilateral contact condition

Huong Le Thi

▶ To cite this version:

Huong Le Thi. On some periodic solutions of discrete vibro-impact systems with a unilateral contact condition. General Mathematics [math.GM]. Université Côte d'Azur, 2017. English. <NNT: 2017AZUR4033>. <tel-01665818>

HAL Id: tel-01665818 https://tel.archives-ouvertes.fr/tel-01665818

Submitted on 17 Dec 2017 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Université Côte d'Azur

Doctoral School **Sciences Fondamentales et Appliquées** Laboratory **Laboratoire de Mathématiques J. A. Dieudonné**

Thesis defended by Huong LE THI

Defended on 16th June, 2017

In order to become Doctor from Université Côte d'Azur

Academic Field Mathematics

On some periodic solutions of discrete vibro-impact oscillators with a unique unilateral contact condition

Thesis supervised by Stéphane JUNCA

Committee members

Referees	Guillaume James Remco Leine	Professor at INP Professor at University of Stuttgart
Examiners	Gérard Iooss Mike Jeffrey Claude-Henri Lamarque Mathias Legrand Cyril Touzé	Professor at University of Nice Senior Lecturer at University of Bristol Professor at ENTPE Assistant Professor at McGill University Professor at ENSTA
Guests	René Lozi Bernard Rousselet	Professor at University of Nice Professor at University of Nice
Supervisor	Stéphane Junca	Associate Professor at University of Nice



Université Côte d'Azur

École doctorale Sciences Fondamentales et Appliquées Unité de recherche Laboratoire de Mathématiques J. A. Dieudonné

Thèse présentée par Huong LE THI

Soutenue le 16 juin 2017

En vue de l'obtention du grade de docteur de l'Université Côte d'Azur

Discipline Mathématiques

Sur des solutions périodiques d'oscillateurs discrets à vibro-impact avec une unique condition de contact unilatéral

Thèse dirigée par Stéphane JUNCA

Composition du jury

Rapporteurs	Guillaume James Remco Leine	professeur à l'INP professeur à l'Université de Stuttgart
Examinateurs	Gérard Iooss Mike Jeffrey Claude-Henri Lamarque Mathias Legrand Cyril Touzé	professeur à l'Université de Nice chargé de cours à l'Université de Bristol professeur à l'ENTPE professeur assistant à l'Université McGill professeur à l'ENSTA
Invités	René Lozi Bernard Rousselet	professeur à l'Université de Nice professeur à l'Université de Nice
Directeur de thèse	Stéphane Junca	мсғ à l'Université de Nice

This thesis has been prepared at

Laboratoire de Mathématiques J. A. Dieudonné

UMR 7351 CNRS UNS 06108 Nice Cedex 02 France

☎(33)(0)4 92 07 62 29☑(33)(0)4 93 51 79 74Web Sitehttp://math.unice.fr/



Abstract

The purpose of this thesis is to investigate *N* degree-of-freedom vibro-impact oscillators with an unilateral contact. The dynamics is linear in the absence of contact; it is governed by an impact law otherwise. Trajectories that display a sticking phase are identified. The First Return Map is a fundamental tool to explore such periodic solutions. The Poincaré section is tangent to grazing orbits and thus yields the well-known square-root singularity, as already reported in Mechanics, which is here revisited in a rigorous mathematical framework. Another important singularity is exhibited: the discontinuity of the First Return Time. Finally, the square-root dynamics near the linear grazing orbits which may lead to the instability of these linear grazing orbits is studied. It is found that the square-root dynamics emerges from the square-root singularity of the First Return Time if one of the coefficients related to a linear map does not vanish. Under a generic condition of the matrix of eigenvectors, the square-root dynamics near a linear grazing orbit is proven to exist.

Keywords: nonsmooth analysis, vibro-impact systems, unilateral contact, periodic solutions, sticking phase, linear grazing orbit, poincaré map

Sur des solutions périodiques d'oscillateurs discrets à vibro-impact avec une unique condition de contact unilatéral

Résumé

Le but de cette thèse est d'étudier des oscillateurs à vibro-impact à *N* degrés de liberté avec une condition de contact unilatéral. La dynamique est linéaire en l'absence de contact ; elle est régie par une loi d'impact autrement. Des trajectoires présentant une phase de contact collant sont identifiées. L'application de premier retour de Poincaré est un outil fondamental pour étudier la dynamique près de solutions périodiques. La section de Poincaré est tangente aux orbites rasantes et conduit à une singularité en « racine carrée », déjà connue en Mécanique, singularité revisitée dans un cadre mathématique rigoureux. Elle implique la discontinuité du temps de premier retour. Enfin, la dynamique en racine carrée près des modes linéaires rasants, qui peut conduire à l'instabilité de ces modes, est abordée. On constate que la dynamique en racine carrée émerge de la singularité en racine carrée du temps de premier retour si l'un des coefficients liés à une carte linéaire ne s'annule pas. Pour une condition générique de la matrice des vecteurs propres, la dynamique en racine carrée près d'une mode linéaire rasants est prouvée.

Mots clés : analyse non lisse, systèmes discrets à vibro-impact, contact unilatéral, solutions périodiques, mode linéaire rasant, l'application de poincaré

Acknowledgments

I would like to express my deepest gratitude to my supervisor Stéphane Junca for his never-ending patience and generosity. Working with him taught me to be confident and to face complicated situations by simplifying them. Above all I have learned from him that the idyllic beauty of mathematics springs from the most basic concepts. His innovativeness never ceases to amaze me. I would also like to acknowledge Mathias Legrand for the many invaluable discussions with him during my doctoral studies. It was a pleasure to be able to work and write articles with him.

I would like to thank Bernard Rousselet for the helpful discussions and for spending time to answer my questions. My gratitude also goes to Remco Leine, Guillaume James, Gérard Iooss, Mike Jeffrey, Claude-Henri Lamarque, Cyril Touzé, and René Lozi, for accepting to be a member of the jury for my defence and for their insightful questions and comments that lead to the improvement of my work.

I am greatly indebted to the Eramus Mundus Mobility with Asia for providing the scholarship for my doctoral studies. Thanks must also be accorded to the international relations department of Thang Long University for facilitating the scholarship procedures. I also would like to convey my sincere thanks to the laboratory J.A. Dieudonné for hosting my stay in Nice and providing opportunities to attend and present my work in scientific conferences.

In particular, I would like to recognize the support and thoughtful concern of Marc and Francine Diener, and the EMMA team, especially Julie. I am also grateful to Jean Marc, Roland, Chiara, Julia, Angelique for their unfailing support and assistance in technical and administrative processes in Laboratoire Dieudonné.

For the scientific discussions, companionship and their advice, I thank my colleagues: anh Bien, ka Lam, chi Yen, chi Phuong Anh, Carole, Melisande. For sharing the joyful moments of life as well as the hard times in Nice, I thank to my friends especially chi Nga and Yen. I also thank my other Vietnamese friends in Nice for their help and the Vietnamese Student Association for organising events that keep us feel at home even if we are far away from Viet Nam. For the happy and relaxed moments trying new food and discussing almost everything under the sun, I thank the Filipino EMMA group and Win.

My journey would not have been complete and interesting without the presence, infinite patience and infectious joy of Emmanuel. I take this opportunity to thank him for journeying with me in both my scientific and personal life for almost the whole duration of my PhD. With his guidance and encouragement, I was able to face all challenges.

Lastly, to my family: bo me, chi Thuy, Duc, Natasha and Misha, thank you for never giving up on me and always encouraging me when I have any trouble. Thinking about how hard my parents work so that we, their children, can go to school and have a better life motivated me to work harder and do my best for this thesis. The cheerful smiles of Natasha and Misha never fail to refresh me from my fatigue. I also thank my older sister, my closest friend for always supporting me. Credit is due to my younger brother for being ready any time I need him, especially for his help in computer-related tasks.

It is impossible to list and thank in this acknowledgement all the people whom I have met and shared wonderful time with during my stay in Nice. To everyone, thank you very much! LE THI Huong

Table of contents

Abstract	vii
Acknowledgments	ix
Table of contents	xi
Introduction	1
1 On solutions to vibro-impact oscillators	5
2 Periodic solutions with sticking phases of a 2-dof oscillator	13
3 On the first return time of <i>N</i> -dof vibro-impact oscillators	49
Conclusions	93
Bibliography	95
A Power-root singularity for a mass-spring chain	99
B 1-IPP dynamics with internal resonances near grazing orbits	101
Contents	105

Introduction

Undergraduate and graduate students who learn physics and differential equations learn how to solve analytically the dynamics of a single mass attached to a wall by a spring. Such oscillating systems form the foundation for more complex behaviours as found in real applications. For instance, various models have been proposed to model the motion of blades in turbomachinery applications, see Figure 1(a). A simplified continuous model for the blade could be a thin longitudinal elastic rod attached to a wall and which stretches until it interacts with a constraint, which plays the role of the casing, see Figure 1(b). In the framework of the continuous model, there exists a methodology to perform modal analysis of elastic structures [24]. Moreover, numerical procedures capable of performing nonsmooth modal analysis of an elastic bar subject to a unilateral contact constraint are proposed in the literature [52, 51]. Another approach is to use discrete models since they can exhibit a variety of behaviors that bring important insight on nonsmoothness [6, 12, 20]. Hence, the present work focuses on a discrete model where the rod is replaced by a vibrating chain of masses connected by springs, one of which being attached to a wall. Depending on the initial conditions, this chain may hit the constraint. This model is commonly referred to as a vibro-impact system with a unilateral constraint; the unilateral constraint comes from the presence of a rigid foundation.

It is known from linear theory that the modes of vibration of a discrete or continuous system can be used to decouple the equations of motion, and that any arbitrary oscillation can be expressed as superposition of the modal responses. Any forced resonances of the system under external harmonic forces always occur in neighborhoods of normal modes. Moreover, linear normal modes are neutrally stable. A mode is unstable if a small perturbation of the initial conditions that are required for the realization of the modal motion leads to the elimination of the mode oscillation [48]. These properties are however not necessarily enjoyed by the so-called nonlinear normal modes arising from the study of nonlinear systems [43, 44]. Nonlinear modes are defined as continuous one-parameter families of periodic orbits [22, 49, 39]. For the discrete *N*-dof system undergoing an energy-preserving impact law on one of its masses, nonlinear modes of vibration with non-grazing impact are explored in [38, 35, 19, 23, 25, 32, 46, 45]. It was shown in [25] that the families of periodic orbits featuring one impact per period exist and lie on two-dimensional invariant manifolds in the state-space. The stability of the such periodic orbits



(-)

FIGURE 1 - (a) Turbomachinery blade and casing. (b) Thin rod with a unilateral constraint

lying on the invariant manifolds is also established. Further research on stability properties of non-smooth dynamical systems is found in [27].

In the same aspiration of seeking for nonlinear modes with non-grazing impacts, the author identifies the nonlinear modes that display long "sticking" contact phases. The term "sticking" here means that the impacting mass will rest on the rigid foundation for a finite or infinite amount of time during its motion even though there is no "sticking force" arising from the rigid foundation and acting on the impacting mass. Results indicate that the periodic solutions involving sticking phases are isolated, which is different from the nonlinear modes involving non-zero velocity impacts.

Another main task of the thesis is to investigate and understand the dynamics via the First Return Map (FRM) [18]. In smooth analysis, this map is defined in a so-called Poincaré section which is naturally chosen to be transverse to the flow of the system. The first return map then associates the points in Poincaré section with the first return of the flow to this Poincaré section. The impact hyperplane is the natural choice for the Poincaré section in the study of impacts with non-zero pre-impact velocity [25, 21]. While treating the grazing trajectory, the transversality condition is lost. Nordmark et al [7, 17, 33] introduced a discontinuity mapping on a suitable Poincaré section lying outside the contact interface to keep the transversality condition. For a one-dof forced oscillator that impacts a rigid obstacle, Chillingworth [9, 10] used an alternative geometric approach which replaces the picture of curved orbits intersecting a flat obstacle with and equivalent picture of straightened out orbits intersecting a curved obstacle: the impact surface. Particular interest in bifurcations is investigated near the grazing trajectory [3, 7, 17,

29, 30, 26, 50, 11, 15]. In the present work, the FRM, which is the fundamental tool to explore periodic solutions, is extended to the investigation of other potential solutions in order to improve the understanding of the global dynamics. Since the Poincaré section is a subset of the contact interface in the phase-space, it can be tangent to orbits which yields the well-known square-root singularity [7, 33]. This singularity is here revisited in a rigorous mathematical framework. Moreover, the study of this singularity implies a more important singularity: the discontinuity of the First Return Time (FRT).

The thesis is organized as follows:

- Chapter 1 is devoted to proving the existence of the solution to the N-dof vibro-impact system that is based on the fundamental existence result given by Schatzman [40]. The well-posedness for a general problem in Rⁿ without friction and under strong restrictions on the regularity of the data was shown by Ballard [5]. In this chapter, the difficulty in giving a simpler proof on the uniqueness of the solution is also proposed.
- Chapter 2 concerns a class of periodic solutions which involve sticking phases. As a minimal model, periodic solutions with one sticking phase-per-period (1-SPP) of a two-dof vibro impact system are considered. A method to obtain such periodic solutions is proposed: it provides conditions on the existence of 1-SPP as well as closed-form expressions of these periodic solutions. In particular, the set of 1-SPP is shown to be at most a countable set of isolated periodic orbits. These solutions are then compared to the nonlinear modes of vibration involving one impact per period (1-IPP), where it is shown that there is an equivalence between 1-SPP and a special set of isolated 1-IPP. It is also demonstrated that the pre-stressed system features sticking phases of infinite duration.
- The first return map is explored in Chapter 3 in a rigorous mathematical framework to describe the square-root singularity for a linear vibro-impact system with a unilateral contact. The Poincaré section is chosen at the contact interface at the cost of losing of the transversality condition. New singular behaviors are observed. In particular, the discontinuity of the First Return Time is displayed along grazing orbits. This discontinuity is largely undesirable because it may cause the instability of the periodic solutions with grazing contacts. Finally, a general condition is given to verify the square-root dynamics in the vicinity of the linear grazing modes (LGM), the periodic solutions with one grazing contact per period, which are known to be the source of many branches of nonlinear modes (e.g. *k*-impact-per-period [25]). From this square-root dynamics might emanate the instability of these linear grazing modes.

Conclusions and remaining open questions are listed in the Conclusion. Appendices, which contain alternative proofs or detailed calculations, follow.

Chapter 1

On solutions to vibro-impact oscillators

Summary The aim of this chapter is to give a simple proof for the existence and discuss around the uniqueness of the solution to a vibro-impact system with a unilateral constraint and without source terms. It is also confirmed that the impacts are isolated, which is known from [5].

1.1 Introduction

In this chapter, the existence of the solutions of the *N*-degree-of-freedom (dof) vibro-impact system is given. There is no such theorem like the Cauchy-Lipschitz theorem in the case of a unilateral constraint [5]. The first investigation on the existence and uniqueness of solutions for such systems is that of Schatzman [40] and the Italian school. The study was limited on the configuration space \mathbb{R}^d and to the case of an elastic impact. The proof on the existence of solution is based on the Yosida regularization and compactness arguments. The uniqueness is however proved only for a very specific case. She also gave examples where the loss of regularity can affect the uniqueness. For further investigation on the uniqueness of a solution, Percivale is the first to introduce the analyticity hypothesis [36]. However, the results only apply to a very specific case. The problem with completely inelastic impacts has been studied by Moreau. The one-degree-of-freedom problem with arbitrary impact constitutive law is studied by Schatzman. The uniqueness is proved under the analyticity of the data. Ballard [5] gave a complete proof on the existence and uniqueness of solution in a general setting for *N*-dof system where it is required that all the data are analytic to ensure the existence and uniqueness.

By Proposition 19 in [5, pp. 248-249], it is proven that infinitely many impacts can accumulate at the left of a given instant but *not at the right*. This is a specific feature of the analytical setting that is lost in the C^{∞} -setting. Moreover, in the case of purely elastic impacts, the instants of impact are *isolated*.

The system of interest is a second-order system

$$\mathbf{M}\ddot{\mathbf{u}} + \mathbf{K}\mathbf{u} = \mathbf{r} \tag{1.1a}$$

$$\mathbf{u}(0) = \mathbf{u}_0, \quad \dot{\mathbf{u}}(0) = \dot{\mathbf{u}}_0 \tag{1.1b}$$

$$u_N(t) \le d, \quad R(t) \le 0, \quad (u_N(t) - d) R(t) = 0$$
 (1.1c)

$$\dot{\mathbf{u}}^{+}(t)^{\mathsf{T}}\mathbf{M}\dot{\mathbf{u}}^{+}(t) + \mathbf{u}^{\mathsf{T}}(t)\mathbf{K}\mathbf{u}(t) = E(\mathbf{u}(t), \dot{\mathbf{u}}^{+}(t)) = E(\mathbf{u}(0), \dot{\mathbf{u}}(0)),$$
(1.1d)

with

$$\mathbf{M} = \mathbf{diag}(m_j)_{j=1}^N; \quad \mathbf{K} = (k_{ij})_{i,j=1}^N; \quad \mathbf{u}(t) = (u_j)_{j=1}^N; \quad \mathbf{r}(t) = (0, \dots, 0, R(t))$$
(1.2)

where u_j , \dot{u}_j and \ddot{u}_j represent the displacement from its equilibrium, the velocity and acceleration of the mass j, j = 1, ..., N, respectively. Condition (1.1c) says that mass N is constrained on the right side by a rigid obstacle at a distance d > 0 from its equilibrium. The N^{th} mass is assumed to be unilaterally constrained by the presence of a rigid foundation. The other masses are not constrained in any way. The quantity R(t) is the reaction force induced by the obstacle on mass N at the time of gap closure. In general R(t) is a measure, but for solutions with sticking phases, it is a Lipschitz function that is as regular as \ddot{u}_N [28].

Matrices **M** and **K** are symmetric positive definite. There is then a matrix **P** of eigenmodes that simultaneously diagonalizes both of them, that is $\mathbf{P}^{\mathsf{T}}\mathbf{M}\mathbf{P} = \mathbf{I}$ and $\mathbf{P}^{\mathsf{T}}\mathbf{K}\mathbf{P} = \mathbf{\Omega}^2 = \mathbf{diag}(\omega_i^2)|_{i=1}^N$ where **I** is the $N \times N$ identity matrix and ω_i^2 , i = 1, ..., N are the eigenfrequencies of the linear system without unilateral contact condition.

Equation (1.1d) reflects the conservative nature of the system. The energy is preserved during a motion. Equation (1.1d) commonly implies the existence of a perfectly elastic impact law of the form $\dot{u}_N^+ = -e\dot{u}_N^-$ with e = 1 where \dot{u}_N^- and \dot{u}_N^+ respectively stand for the pre- and post-impact velocities of the Nth mass.

It is known [5] and will be proven in the next section that there exists a solution **u** to System (1.1) which belongs to the space of motions with measure acceleration (MMA). For our system in which there is only one potential contact point between the N^{th} mass and the obstacle, there is more regularity for u_1, \ldots, u_{N-1} than u_N . In the two-dof vibro-impact system modelling a chain of two masses, investigated in Chapter 2, it is shown that u_1 belongs to BV⁴ while $u_2 \in \text{MMA}$ (or BV²) where the space BVⁿ is defined as follows: $\mathbf{u} \in \text{BV}^n$ means $\frac{d^n u}{dt^n}$ is a measure (or $\frac{d^{n-1}u}{dt^{n-1}} \in \text{BV}$), i.e it is a function of bounded variation for every integer $n \ge 1$.

This chapter is organized as follows. The proof of the solution existence to the solutions of the N-dof vibro-impact system is stated in Section 1.2. In Section 1.3, a discussion on the uniqueness is given.

1.2 On the existence

We now give the theorem on the existence of solutions to System (1.1). At the time t when $u_N(t) = d$, the total energy $E(\mathbf{u}(t), \dot{\mathbf{u}}(t))$ is conserved at t+ and t-. Hence, we can assume $E(\mathbf{u}(t), \dot{\mathbf{u}}^-(t)) = E(\mathbf{u}(0), \dot{\mathbf{u}}(0)), \forall t$.

Theorem 1.1. Consider System (1.1). There exists a solution $\mathbf{u} \in \text{MMA}([0; \infty[, \mathbb{R}^N) \text{ if } u_N(0) \le d$. Moreover, $\dot{u}_N \in \text{BV}_{\text{loc}}([0; +\infty[, \mathbb{R}^N).$

Besides, the other components u_k , $k \neq N$, are more regular than u_N .

Proof. To show existence, Yosida regularization is used. For $\epsilon > 0$, let us consider the system with the initial data satisfying the constraint condition

$$\mathbf{M}\ddot{\mathbf{u}} + \mathbf{K}\mathbf{u} = \mathbf{r}_{\epsilon} \tag{1.3}$$

$$(\mathbf{u}(0), \dot{\mathbf{u}}(0)) = (\mathbf{u}_0, \dot{\mathbf{u}}_0) \in \mathbb{R}^N \times \mathbb{R}^N$$
(1.4)

$$u_N(0) \le d \tag{1.5}$$

where $\mathbf{r}_{\epsilon} = [0, ..., 0, -(u_N(t) - d)_+/\epsilon]^{\top}$ is a Lipschitz function, with $x_+ = \max(0, x)$. Via the Cauchy-Lipschitz theorem, there exists a unique solution $\mathbf{u}_{\epsilon} \in C^2([0; \infty[)$. Since the initial data satisfies (1.5), the total energy of the system reads

$$E_{\epsilon}(\mathbf{u}_{\epsilon}(t), \dot{\mathbf{u}}_{\epsilon}(t)) = E(\mathbf{u}_{\epsilon}(t), \dot{\mathbf{u}}_{\epsilon}(t)) + \frac{1}{\epsilon} \left((\mathbf{u}_{\epsilon N}(t) - d)_{+} \right)^{2}$$
(1.6)

$$= E_{\epsilon}(\mathbf{u}_{\epsilon}(t), \dot{\mathbf{u}}_{\epsilon}(t)) = E(\mathbf{u}_{0}, \dot{\mathbf{u}}_{0})$$
(1.7)

where $\mathbf{u}_{\epsilon N}$ is the N^{th} coordinate of \mathbf{u}_{ϵ} . Then the total energy is constant and independent of ϵ . In particular, $E(\mathbf{u}_{\epsilon}(t), \dot{\mathbf{u}}_{\epsilon}(t)) \leq E(\mathbf{u}_{0}, \dot{\mathbf{u}}_{0})$ and $((\mathbf{u}_{\epsilon N}(t) - d)_{+})^{2} \leq \epsilon E(\mathbf{u}_{0}, \dot{\mathbf{u}}_{0})$ and thus

$$\mathbf{u}_{\epsilon N}(t) \le d + \sqrt{\epsilon E(\mathbf{u}_0, \dot{\mathbf{u}}_0)}.$$
(1.8)

Since the energy $E_{\epsilon}(\mathbf{u}(t), \dot{\mathbf{u}}_{\epsilon}(t))$ is conserved, for $\mu > 0$ and $\kappa > 0$, we get

$$E(\mathbf{u}_0, \dot{\mathbf{u}}_0) \ge E_{\epsilon}(\mathbf{u}_{\epsilon}(t), \dot{\mathbf{u}}_{\epsilon}(t)) \ge \mathbf{u}_{\epsilon}^{\top}(t) \mathbf{K} \mathbf{u}_{\epsilon}(t) \ge \kappa \|\mathbf{u}_{\epsilon}(t)\|,$$
(1.9)

$$E(\mathbf{u}_0, \dot{\mathbf{u}}_0) \ge E_{\epsilon}(\mathbf{u}_{\epsilon}(t), \dot{\mathbf{u}}_{\epsilon}(t)) \ge \dot{\mathbf{u}}_{\epsilon}^{\top}(t)\mathbf{M}\dot{\mathbf{u}}_{\epsilon}(t) \ge \mu \|\dot{\mathbf{u}}_{\epsilon}(t)\|.$$
(1.10)

Accordingly, $\mathbf{u}_{\epsilon}(t)$ is bounded in $C^{1}([0; +\infty[, \mathbb{R}^{N}))$. Using Ascoli's theorem to the family of $\{\mathbf{u}_{\epsilon}\}_{\epsilon>0}$ defined on [0; T[with T > 0, we obtain that $\{\mathbf{u}_{\epsilon}\}_{\epsilon>0}$ is relatively compact in $C^{0}([0; T[, \mathbb{R}^{N}))$. As a consequence, there exists a subsequence $\{\mathbf{u}_{n}\} := \{\mathbf{u}_{\epsilon_{n}}\}_{n}$ that uniformly converges to $\mathbf{u} \in \operatorname{Lip}([0; T])$.

We now verify that $\mathbf{u}(t)$ satisfies the conditions of the original system (1.1).

- $u_N(t) \le d, \forall t$ From Inequality (1.8), by taking the limit as $\epsilon_n \to 0$, we have $u_N(t) \le d$.
- $R(t) \le 0$ Let $R_{\epsilon}(t) = -(u_{\epsilon N}(t) d)_{+}/\epsilon$, it follows that $R_{\epsilon}(t) \le 0$. Integrating the system (1.1a) from 0 to t with 0 < t < T we have

$$\int_0^t \left(\mathbf{M}\ddot{\mathbf{u}}(t) + \mathbf{K}\mathbf{u}(t)\right) dt = \int_0^t \mathbf{r}_\epsilon(t) dt.$$
(1.11)

Consider the last equation

$$m_N \dot{u}_N \Big|_0^t + \sum_{i=1}^N k_{Ni} \int_0^t u_i(s) ds = \int_0^t R_\epsilon(s) ds, \qquad (1.12)$$

that is

$$O(1) + O(T) = \int_0^t R_{\epsilon}(s) ds.$$
 (1.13)

It follows that $\int_0^t |R_{\epsilon}(s)| ds \leq O(1+T)$. Therefore, R_{ϵ} is uniformly bounded in $L^1([0;T[)$ and consequently in $\mathcal{M}^1([0;T[))$. Hence R_{ϵ} weakly converges to $R \in \mathcal{M}^1([0;T[))$. Thus $R(t) \leq 0$.

- $(u_N(t) d) R(t) = 0$, i.e. supp $R(t) \subset \{t : u_N(t) = d\}$ If $u_N(t) = d$ then $(u_N(t) d) R(t) = 0$. Otherwise, if $u_N(t_0) < d$, for $t_0 \in I = [t_{-1}; t_{+1}]$ (an arbitrary closed interval). Since u_N is continuous, there exists $\delta > 0$ such that $u_N(t) < d \delta$, $\forall t \in I$. For $t \in I$, $u_{\epsilon_n N} \to u_N$ uniformly in L^{∞} . Therefore, $\exists n_0$ such that $\forall n > n_0$, $u_{\epsilon_n N}(t) < d \delta/2$ on I. It follows that $-(u_{\epsilon_n N}(t) d)_+/\epsilon_n = 0$, for $n \ge n_0$ and $t \in I$. In other words, $R_{\epsilon_n}(t) = 0$, for all $n \ge n_0$, and $t \in I$. Hence, $R(t) = \lim_{\epsilon_n \downarrow 0} R_{\epsilon_n}(t) = 0$, $\forall t \in I$. This shows that $(u_N(t) d)R(t) = 0$.
- The energy is conserved Set $\alpha_{\epsilon}(t) = ((u_{\epsilon N}(t) d)_{+})^{2} / \epsilon \leq E(\mathbf{u}_{0}, \dot{\mathbf{u}}_{0})$. We will show that $\alpha_{n} := \alpha_{\epsilon_{n}}$ converges to 0 as *n* tends to infinity. The definition of R_{ϵ} and α_{ϵ} yields

$$|R_{\epsilon}(t)| = \frac{(u_{\epsilon N}(t) - d)_{+}}{\epsilon}, \qquad (1.14)$$

$$0 \le \alpha_{\epsilon}(t) = |R_{\epsilon}(t)|(u_{\epsilon N}(t) - d)_{+}, \qquad (1.15)$$

$$0 \le \int_0^T \alpha_{\epsilon}(t)dt \le \left(\int_0^T |R_{\epsilon}(t)|dt\right) \sup_{0 \le t \le T} (u_{\epsilon N}(t) - d)_+ \to 0, \tag{1.16}$$

since $\int_0^T |R_{\epsilon}(t)| dt$ is uniformly bounded, and $u_{\epsilon N}$ uniformly converges to d. Thus $\alpha_{\epsilon_n} \to 0$ in L^1_{loc} . It is up to a subsequence $\alpha_{\epsilon_{n_{n'}}} = \alpha_{n'}(t) \to 0$ for almost all $t \in [0; T[$. From (1.6), it follows that, when ϵ tends to 0, $E_{\epsilon}(\mathbf{u}(t), \dot{\mathbf{u}}(t))$ converges to a constant, which is exactly the energy $E(\mathbf{u}(t), \dot{\mathbf{u}}(t))$ of (1.1). We will now show that E is conserved. For all time t outside the impact time, i.e. for all t such that $u_N(t) < d$, total energy is conserved: $E(\mathbf{u}(t), \dot{\mathbf{u}}(t)) = E(\mathbf{u}_0, \dot{\mathbf{u}}_0)$. We will now prove that for any impact time t,

 $E(\mathbf{u}(t), \dot{\mathbf{u}}^{-}(t)) = E(\mathbf{u}_0, \dot{\mathbf{u}}_0) = E(\mathbf{u}(t), \dot{\mathbf{u}}^{+}(t))$, where $E(\mathbf{u}(t), \dot{\mathbf{u}}^{-}(t))$ and $E(\mathbf{u}(t), \dot{\mathbf{u}}^{+}(t))$ are the pre- and post-impact energies of the system. From the last equation

$$m_N \ddot{u}_N + \sum_{i=1}^N k_{Ni} u_i = R(t), \qquad (1.17)$$

and knowing that $\mathbf{u} \in C^0([0;T[) \subset L^1([0;T[) \subset \mathcal{M}^1([0;T[), \text{ and } R(t) \in \mathcal{M}^1([0;T[), \text{ it follows that } \ddot{u}_N \in \mathcal{M}^1([0;T[), \text{ i.e. } \dot{u}_N \in BV([0;T[) \text{ where BV is the space of functions with bounded variations. Therefore, <math>\dot{u}_N^-(t)$ and $\dot{u}_N^+(t)$ exist for all time *t*. These are the only factors that change and affect the energy of the system. Recall that the set of impacts, i.e. $\{t : \dot{u}_N^-(t) \neq \dot{u}_N^+(t)\}$ is countable since \dot{u}_N is a BV function. Hence, $E(\mathbf{u}(t), \dot{\mathbf{u}}^-(t)) = E(\mathbf{u}(t), \dot{\mathbf{u}}^+(t))$, for all *t*.

1.3 On the uniqueness

This section is devoted to a brief discussion on the uniqueness of the solution of (1.1). A long proof is provided in [5] for a general framework and under the assumption of analyticity of all data. In our specific framework with linear stiffness, a simpler proof is expected. Nevertheless, the uniqueness of the solution in the presence of sticking contact is not easy to prove. Let us explain further this difficulty. Given an initial data (1.1b), the solution of (1.1) is unique as long as there is no contact. The question on the uniqueness arises when contacts are taken into account.

The local uniqueness of the solutions is addressed in the right neighborhood of an impact which and without loss of generality, is assumed to be at t = 0. It is proven that there exists a solution $\mathbf{U}_a = [\mathbf{u}_a, \dot{\mathbf{u}}_a]$ in MMA([0; t_0 [) of (1.1), where $t_0 > 0$.

If the impact is with non-zero velocity, i.e. $\dot{u}_{aN}^-(0) > 0$, then by (1.1d), the energy of the system is conserved right before and right after the impact time t = 0:

$$E(\mathbf{u}(0), \dot{\mathbf{u}}^{+}(0)) = E(\mathbf{u}(0), \dot{\mathbf{u}}^{-}(0)), \qquad (1.18)$$

$$\dot{\mathbf{u}}_{a}^{+}(0)^{\mathsf{T}}\mathbf{M}\dot{\mathbf{u}}_{a}^{+}(0) + \mathbf{u}_{a}(0)^{\mathsf{T}}\mathbf{K}\mathbf{u}_{a}(0) = \dot{\mathbf{u}}_{a}^{-}(0)^{\mathsf{T}}\mathbf{M}\dot{\mathbf{u}}_{a}^{-}(0) + \mathbf{u}_{a}(0)^{\mathsf{T}}\mathbf{K}\mathbf{u}_{a}(0).$$
(1.19)

This gives $(\dot{u}_{aN}^+(0))^2 = (\dot{u}_{aN}^-(0))^2$. It follows that the velocity of mass *N* after the impact is $\dot{u}_{aN}^+(0) = -\dot{u}_{aN}^-(0) < 0$. Otherwise, the mass penetrates the obstacle. Indeed, mass *N* bounces back immediately since

$$u_{aN}(t) = u_{aN}(0) + \int_0^t \dot{u}_{aN}(s)ds = u_{aN}(0) + t \,\dot{u}_{aN}^+(0) + o(t)$$
(1.20)

$$= d + t(\dot{u}_{aN}^+(0) + o(1)) < d \text{ for } t \gtrsim 0$$
(1.21)

where t > 0, $\dot{u}_{aN}^+(0) < 0$. After the impact, the *N* masses are driven by linear dynamics similar to (1.1a) where $\mathbf{r} = 0$ with the new initial data at 0: $[\mathbf{u}_a(0), \dot{\mathbf{u}}_a^+(0)]$. Via the Cauchy theorem, uniqueness is guaranteed for $t \in [0, \epsilon]$ where $\epsilon > 0$ is small enough so that there is no contact between the *N*th mass and the obstacle.

If the impact is with zero velocity, then $\dot{u}_{aN}^+(0) = -\dot{u}_{aN}^-(0) = 0$. Let **z** be the solution of the linear system with the initial data $[\mathbf{z}(0), \dot{\mathbf{z}}(0)] = [\mathbf{u}_a(0), \dot{\mathbf{u}}_a^+(0)]$. Let us divide this into two cases:

Case 1: Suppose that there is no sticking phase at t = 0, i.e. $u_{aN}(t) < d$ for all $t \in [0; \delta[$, where $\delta > 0$ is small enough so that there is no other contact between t = 0 and $t = \delta$. Hence, for all $t \in [0; \delta[$, \mathbf{u}_a is analytic and satisfies

$$\mathbf{M}\ddot{\mathbf{u}}_a + \mathbf{K}\mathbf{u}_a = \mathbf{0}.\tag{1.22}$$

Assume that there is another solution $\mathbf{U}(t) = [\mathbf{u}(t), \dot{\mathbf{u}}(t)]$ of (1.1) for $t \in [0; \delta]$. In Section 1.2, it is proven that $\dot{\mathbf{u}} \in BV(]0; \delta[$). Let $\mathbf{v}(t) = \mathbf{u}(t) - \mathbf{u}_a(t), t \in [0; \delta]$, it follows that $\dot{\mathbf{v}} \in BV(]0; \delta[$) and in the interval $[0; \delta]$, $\mathbf{v}(t)$ satisfies

$$\mathbf{M}\ddot{\mathbf{v}} + \mathbf{K}\mathbf{v} = \mathbf{r} \tag{1.23}$$

together with $\mathbf{v}(0) = \mathbf{0}$ and $\dot{\mathbf{v}}(0) = \mathbf{0}$. Let us define the quantity $\bar{u}_N = (\dot{u}_N^+ + \dot{u}_N^-)/2$. Take note that the notation \bar{u}_N coincides with the classical notation of derivative of u_N at any time away from impacts. Since the energy of the system is conserved, it follows that $\dot{u}_N^+(t) = -\dot{u}_N^-(t)$ where t is the impact time and

$$\bar{u}_N(t) = \begin{cases} 0 & \text{if } u_N(t) = d, \\ \dot{u}_N(t) & \text{if } u_N(t) < d. \end{cases}$$
(1.24)

Accordingly, $\bar{v}_N(t) = \bar{u}_N - \dot{u}_{aN}$ satisfies

$$\bar{\dot{v}}_N(t) = \begin{cases} 0 & \text{if } u_N(t) = d \\ \dot{v}_N(t) & \text{if } u_N(t) < d. \end{cases}$$
(1.25)

Pre-multiplying both sides of (2.1a) by $\mathbf{\bar{v}} = [\dot{v}_1, \dots, \dot{v}_{N-1}, \mathbf{\bar{v}}_N]^\top$ yields

$$\bar{\mathbf{v}}^{\mathsf{T}}\mathbf{M}\bar{\mathbf{v}} + \bar{\mathbf{v}}^{\mathsf{T}}\mathbf{K}\mathbf{v} = \bar{\dot{v}}_{N}(t)R(t), \tag{1.26}$$

$$\dot{\mathbf{v}}^{\mathsf{T}}\mathbf{M}\dot{\mathbf{v}} + \mathbf{v}^{\mathsf{T}}\mathbf{K}\mathbf{v} = \int_{0}^{t} \bar{\dot{v}}_{N}(s)R(s)ds.$$
(1.27)

By denoting $E(\mathbf{v}(t), \dot{\mathbf{v}}(t)) = \dot{\mathbf{v}}^{\top}(t)\mathbf{M}\dot{\mathbf{v}}(t) + \mathbf{v}^{\top}(t)\mathbf{K}\mathbf{v}(t)$, it is seen that $E(\mathbf{v}(0), \dot{\mathbf{v}}(0)) = 0$.

Hence,

$$E(\mathbf{v}(t), \dot{\mathbf{v}}(t)) = E(\mathbf{v}(t), \dot{\mathbf{v}}(t)) - E(\mathbf{v}(0), \dot{\mathbf{v}}(0)) = \int_0^t \bar{\dot{\mathbf{v}}}_N(s) R(s) ds.$$
(1.28)

From (1.1c) and (1.25), it follows that $\bar{v}_N(s)R(s) = 0$, $\forall s \in [0; \delta]$. Therefore, $E(\mathbf{v}(t)) = 0$, $\forall t \in]0; \delta[$. It follows that $\mathbf{v}(t) = 0, \forall t \in]0; \delta[$, or $\mathbf{v} \equiv \mathbf{0}$. In other words, $\mathbf{u} \equiv \mathbf{u}_a$, which shows the uniqueness of the solution of (1.1) on the right neighborhood of the impact when there is no sticking phase.

Case 2: Suppose that there is a sticking phase starting at t = 0, i.e. $u_{aN}(t) = d$, $\forall t \in [0; \tau]$, where τ is the end of the sticking phase. For all $t \in [0; \delta[, 0 < \delta < \tau, \text{let } \mathbf{U}(t)$ be another solution of (1.1) on $[0; \delta]$. Then $\mathbf{u}(t)$ and $\mathbf{u}_a(t)$ satisfy (1.1a), (1.1b). The difference $\mathbf{v}(t) = \mathbf{u}(t) - \mathbf{u}_a(t)$ then satisfies

$$\mathbf{M}\ddot{\mathbf{v}} + \mathbf{K}\mathbf{v} = \mathbf{r} - \mathbf{r}_a,\tag{1.29}$$

with $\mathbf{v}(0) = \mathbf{0}$ and $\dot{\mathbf{v}}(0) = \mathbf{0}$ where $\mathbf{r}_a(t) = [0, \dots, 0, R_a(t)]^{\top}$, with $R_a(t) < 0$ as the reaction of the wall during the sticking phase. Premultiplying both sides of (1.1a) by $\mathbf{\bar{v}}$ leads to

$$\bar{\mathbf{v}}^{\top}(t)\mathbf{M}\bar{\mathbf{v}}(t) + \bar{\mathbf{v}}^{\top}(t)\mathbf{K}\mathbf{v}(t) = \bar{\mathbf{v}}_{N}(t)(R(t) - R_{a}(t)),$$
(1.30)

$$E(\mathbf{v}(t), \dot{\mathbf{v}}(t)) = \dot{\mathbf{v}}^{\top}(t)\mathbf{M}\dot{\mathbf{v}}(t) + \mathbf{v}^{\top}(t)\mathbf{K}\mathbf{v}(t) = \int_0^t \bar{\dot{v}}_N(s)\left(R(s) - R_a(s)\right)ds.$$
(1.31)

Notice that $\bar{v}_N(s) = \bar{u}_N(s) - \dot{u}_{aN}(s) = \bar{u}_N(s)$ for all $s \in [0; \delta[$ since $\dot{u}_{aN}(s) = 0$. Hence, $\bar{v}_N(s) = 0$ if there is an impact at the time *s*. As before, $\bar{v}_N(s)R(s) = 0$ since the mean velocity vanishes when $u_N = d$ and the measure *R* is also concentrated when $u_N = d$. Thus,

$$E(\mathbf{v}(t), \dot{\mathbf{v}}(t)) = -\int_0^t \bar{u}_N(s) R_a(s) ds.$$
(1.32)

Notice that $R_a \leq 0$. To show the local uniqueness in this case it suffices to prove that $\overline{\dot{u}}_N(s) \geq 0$. This, however, is not trivial and remains yet to be proven.

Chapter 2

Periodic solutions with sticking phases of a two-degree-of-freedom oscillator

Summary This chapter explores the free dynamics of a simple two-degree-of-freedom vibroimpact oscillator. One degree-of-freedom is limited by the presence of a rigid obstacle and periodic solutions involving one sticking phase per period (1-SPP) are targeted. A method to obtain such orbits is proposed: it provides conditions on the existence of 1-SPP as well as closed-form solutions. It is shown that 1-SPP might not exist for a given combination of masses and stiffnesses. The set of 1-SPP is at most a countable set of isolated periodic orbits. The construction of 1-SPP requires numerical developments that are illustrated on a few relevant examples. Comparison with nonlinear modes of vibration involving one impact per period (1-IPP) is also considered. Interestingly, an equivalence between 1-SPP and a special set of isolated 1-IPP is established. It is also demonstrated that the prestressed system features sticking phases of infinite duration.

2.1 Introduction

In the context of vibration and modal analysis of vibro-impact oscillators, nonlinear modes of vibration with non-grazing impact are explored in [38, 35, 19, 25, 46, 45]. In these works, only one degree-of-freedom (dof) is unilaterally constrained by the presence of a rigid foundation: the dynamics is purely linear when the contact constraint is not active, and governed by an impact law otherwise. Accordingly, the contact force arising when the system interacts with the foundation is a periodic distribution of Dirac deltas.

Instead, the present work pays attention to periodic solutions involving long "sticking" contact phases, thus discarding impulse-driven dynamics reported in the previous works, during which the contacting mass rests against the obstacle for a finite amount of time. Let us clarify the terminology now: as explained later, "sticking" here means that the impacting mass will rest on

the rigid foundation for a finite or infinite amount of time during its motion even though there is no "sticking force" arising from the rigid foundation and acting on the impacting mass. An equivalent terminology would be "lasting non-impulsive" contact phases. This investigation is motivated by the fact that in a continuous setting in space and time, unilateral contact forces are known to be discontinuous functions at most [16] while they become impulsive after a semi-discretization in space via the Finite Element technique. The question is then: are there non-impulsive solutions in the Finite Element framework and alike?



FIGURE 2.1 – Two degree-of-freedom vibro-impact system, d > 0

The system of interest is an oscillator with two masses m_1 and m_2 linearly connected through two springs of stiffness k_1 and k_2 respectively, as depicted in Figure 2.1. The dynamics of interest reads

$$\mathbf{M}\ddot{\mathbf{u}} + \mathbf{K}\mathbf{u} = \mathbf{r} \tag{2.1a}$$

$$\mathbf{u}(0) = \mathbf{u}_0, \quad \dot{\mathbf{u}}(0) = \dot{\mathbf{u}}_0 \tag{2.1b}$$

$$u_2(t) \le d, \quad R(t) \le 0, \quad (u_2(t) - d)R(t) = 0, \quad \forall t$$
 (2.1c)

$$\dot{\mathbf{u}}^{+}(t)^{\mathsf{T}}\mathbf{M}\dot{\mathbf{u}}^{+}(t) + \mathbf{u}^{\mathsf{T}}(t)\mathbf{K}\mathbf{u}(t) = E(\mathbf{u}(t), \dot{\mathbf{u}}^{+}(t)) = E(\mathbf{u}(0), \dot{\mathbf{u}}(0)),$$
(2.1d)

where

$$\mathbf{M} = \begin{bmatrix} m_1 & 0\\ 0 & m_2 \end{bmatrix} \quad ; \quad \mathbf{K} = \begin{bmatrix} k_1 + k_2 & -k_2\\ -k_2 & k_2 \end{bmatrix} \quad ; \quad \mathbf{u}(t) = \begin{pmatrix} u_1(t)\\ u_2(t) \end{pmatrix} \quad ; \quad \mathbf{r}(t) = \begin{pmatrix} 0\\ R(t) \end{pmatrix}. \quad (2.2)$$

Above, u_j , \dot{u}_j , and \ddot{u}_j represent the displacement, velocity, and acceleration of mass j, j = 1, 2, respectively. The gap d is defined between the obstacle and the equilibrium position of the second mass. It is the algebraic *distance* between mass 2 at rest without any external force and the rigid wall and might thus be negative in the prestressed case. The quantity R(t) is the reaction force of the wall on mass 2.

Matrices **M** and **K** are symmetric positive definite so that there is a matrix **P** of eigenmodes which simultaneously diagonalizes both of them, that is $\mathbf{P}^{\top}\mathbf{M}\mathbf{P} = \mathbf{I}$ and $\mathbf{P}^{\top}\mathbf{K}\mathbf{P} = \mathbf{\Omega}^2 =$ $\mathbf{diag}(\omega_i^2)|_{i=1,2}$ where **I** is the 2 × 2 identity matrix and ω_i^2 , i = 1, 2 are the eigenfrequencies of the linear system without unilateral contact.

Equation (2.1d) reflects the conservative nature of the system. The energy is preserved during a motion. Equation (2.1d) implies the existence of a perfectly elastic impact law of the form $\dot{u}_2^+ = -e\dot{u}_2^-$ with e = 1 where \dot{u}_2^- and \dot{u}_2^+ respectively stand for the pre- and post-impact

velocities of mass 2. As detailed later, 1-SPP orbits are defined to be independent of the restitution coefficient *e*. Thus 1-SPP still exist for $e \in [0, 1]$. However, the framework e = 1 is chosen for comparison with another class of periodic solutions with nonzero velocity at the impact. Moreover, the structure of a general solution is simpler when e = 1. For the well-posedness of the initial-value problem with conserved energy, the reader is referred to [36, 5].

The sticking phase is known to appear as a limit of a chattering sequence [4, 7, 8, 34]. Here, there is no source term but sticking periodic solutions can still occur.

The chapter is organized as follows. In Section 2.2, the "sticking phase" definition is provided and the conditions on its occurrence are stated. Then, necessary conditions satisfied by periodic solutions with one sticking phase per period (1-SPP) are given on the period through the free flight duration s which appears to be a key parameter as well as a root of an explicit nonlinear function. Furthermore, when s is known, the corresponding 1-SPP is expressed in closed form. The method and numerical examples are described in Section 2.3 in order to find all 1-SPP.

Mathematical proofs and comments are detailed in Section 2.4, 2.5 and 2.6. More precisely, Section 2.4 deals with the structure of the solution space with a sticking phase. Section 2.5 is devoted to prove Theorem 2.17 on 1-SPP. The existence result of an infinite set of admissible initial data satisfying the constraint $u_2 \le d$ near the sticking phase is proven in Section 2.6. The existence of 1-SPP satisfying the constraint $u_2 \le d$ during the whole period remains an open problem. The prestressed case with $d \le 0$ is discussed in Section 2.7. Section 2.10 concludes the chapter. In this work, all numerical simulations are performed using the parameters $m_1 = m_2 = 1$ and $k_1 = k_2 = 1$, unless indicated otherwise.

2.2 Main results

The occurrence of a sticking phase is first defined with necessary and sufficient conditions in Section 2.2.1. Then, in Section 2.2.2, periodic solutions with one sticking phase per period, the so-called 1-SPP, are characterized through necessary conditions to exhibit them all. Throughout the current work, internal resonances are discarded, i.e. the linear periods satisfy $\bigcap_{j=1}^{2} T_{j}\mathbb{N} = \emptyset$, unless stated otherwise.

2.2.1 Sticking phase

A sticking phase occurs when the mass number 2 stays at $u_2 = d$ on a proper time interval.

Definition 2.1 — Sticking phase and its duration. Let **u** be a solution to System (2.1). A *sticking phase* arises if there exists $t_0 \in \mathbb{R}$ and $\tau > 0$ such that

$$u_2(t) = d, \quad \forall t \in [t_0; t_0 + \tau].$$
(2.3)

Moreover, when there exists $0 < \delta \ll 1$ such that $\forall t \in [0; \delta[$

$$u_2(t_0 - t) < d \text{ and } u_2(t_0 + \tau + t) < d,$$
 (2.4)

then t_0 is the starting time and τ is the duration of the sticking phase.

The central reference [5] is used throughout the chapter: existence is recalled, uniqueness and continuous dependence to the initial data is proved. Moreover, in the conservative case, it is shown that there is no impact accumulation. Accordingly, before and after the sticking phase, condition (2.97) above is sufficient to properly define the beginning and the end of a sticking phase of finite duration. More precisely, it is proven in Proposition 19 of [5] that the impact times are isolated for the perfectly elastic impact law. It is not the same when 0 < e < 1 since chattering might occur [7].

In the present work, the finite duration of the sticking phase for d > 0 is a consequence of Theorem 2.17. This is not always true, as for example with the prestressed case $d \le 0$ detailed in Section 2.7. Such conditions are well known to be related to the sign of the acceleration [7] just before impact and are precisely established for our two-dof system as in [19].

Theorem 2.2 — Sticking contact. Assume d > 0. There exists a sticking phase exactly starting at time t_0 and persisting on its right neighbourhood if and only if:

1.
$$u_2(t_0) = d$$
, $\dot{u}_2(t_0) = 0$, $u_1(t_0) > d$, or

2.
$$u_2(t_0) = d$$
, $\dot{u}_2(t_0) = 0$, $u_1(t_0) = d$, $\dot{u}_1(t_0) > 0$.

The second case where $u_1(t_0) = d$ and $\dot{u}_1(t_0) > 0$ corresponds to the beginning of the sticking phase. The duration of the sticking phase τ then only depends on

$$v = \dot{u}_1(t_0), \text{ and}$$

 $\tau(v) = \frac{2}{\omega} \arctan(\xi v) \text{ where } \omega = \sqrt{\frac{k_1 + k_2}{m_1}} \text{ and } \xi = \frac{\sqrt{(k_1 + k_2)m_1}}{k_1 d}.$ (2.5)

The state of system at the end of the sticking phase is:

$$u_2(t_0 + \tau) = d, \quad \dot{u}_2^-(t_0 + \tau) = 0, \quad u_1(t_0 + \tau) = d, \quad \dot{u}_1(t_0 + \tau) = -\nu.$$
(2.6)

Moreover, the regularity of the curve $\{(u_2(t), \dot{u}_2(t)), t \in \mathbb{R}\}$ is $C^{1.5}$ at the point $(u_2, \dot{u}_2) = (d, 0)$ which corresponds to the time interval $[t_0, t_0 + \tau]$.

The starting of the sticking phase corresponding to the second case stated in Theorem 2.17 is only true for e = 1 since there is the uniqueness of solution in the past for e = 1 [5]. For e < 1, the beginning of the sticking phase can be for $u_1(t_0) > d$. This is easy to see for e = 0. The proof of this theorem is written in Section 2.4. The space C^s is the Sobolev space $W^{s,\infty}$ when *s* is a fractional number [2] which is a generalization of Hölder spaces for s > 1. It should also be highlighted that there is no unilateral constraint on mass 1. The expression $u_1 > d$ sometimes used below does not mean that the mass 1 hits wall since u_1 is the displacement and not the position of mass 1. In the sequel, we do not restrain u_1 .

The loss of regularity at the sticking point is quite clear in Figure 2.2.



FIGURE 2.2 – Admissible 1-SPP with a singularity 1.5 at the intersection between red line and wall, d = 1 and initial data: U(0) = [1, 1, 5.8624394, 0]. (a) Orbits $\{(u_i(t), \dot{u}_i(t)), t \in \mathbb{R}\}, i = 1, 2$. (b) Displacements $t \mapsto u_i(t), i = 1, 2$. (c) Reaction of the wall $t \mapsto R(t)$

This is the least smooth point on the curve $\{(u_2(t), \dot{u}_2(t)), t \in \mathbb{R}\}$. Locally, the orbit is very similar to the graph of $t \mapsto (d - |t|^{1.5}, t)$. Elsewhere, the curve is analytic.

Incidentally, the proof of Theorem 2.17 leads to an explicit classification of all possible contact patterns. There is no accumulation of impacts and only three distinct contact configurations arise. At $t = t_0$, let us assume $u_2(t_0) = d$, then contact is

- 1. an impact if $\dot{u}_{2}(t_{0}) > 0$;
- 2. a grazing contact if $\dot{u}_2(t_0) = 0$ and R(t) = 0 for all $t \approx t_0$;
- 3. a sticking contact if $\dot{u}_2(t_0) = 0$ and R(t) < 0 for some $t \approx t_0$.

Since the velocity of mass 2 shall be discontinuous, $\dot{u}_2^{\pm}(t_0)$ denotes its left/right limit. Due to energy conservation, $\dot{u}_2^{+}(t_0) = -\dot{u}_2^{-}(t_0)$ in our model; in particular, when the incoming velocity

vanishes, that is $\dot{u}_2(t_0) = \dot{u}_2(t_0) = \dot{u}_2(t_0) = 0$, the velocity is continuous.

Sticking contact and grazing contact are the two possible contact occurrences [7] for an impact with zero velocity. In general, it is challenging to know whether a zero pre-velocity impact generates a reaction of the wall without further information. For the considered two-dof system, simple criteria are given on the position and velocity of mass 1 to distinguish a grazing contact from a sticking contact:

- 1. An impact yields an instantaneous rebound with $\dot{u}_2^+(t_0) = -\dot{u}_2^-(t_0) < 0$,
- 2. A grazing contact means that the trajectory would not change irrespective of the presence of the wall. As a corollary of Theorem 2.17, the data at time t_0 is either $u_1(t_0) < d$, $u_2(t_0) = d$, and $\dot{u}_2(t_0) = 0$ or $u_1(t_0) = d$, $\dot{u}_1(t_0) \le 0$, $u_2(t_0) = d$, and $\dot{u}_2(t_0) = 0$. Moreover, there exists $\varepsilon > 0$ such that R(t) = 0 for all $t \in [t_0 \varepsilon; t_0 + \varepsilon[$;
- 3. A sticking contact can be divided into three sequential steps:
 - (a) the beginning of a sticking contact: $u_1(t_0) = d$, $\dot{u}_1(t_0) > 0$, $u_2(t_0) = d$, and $\dot{u}_2(t_0) = 0$. There exists $\varepsilon > 0$ such that R(t) = 0 for all $t \in]t_0 - \varepsilon$; $t_0[$ and R(t) < 0 for all $t \in]t_0$; $t_0 + \varepsilon[$;
 - (b) the resting phase of a sticking contact: $u_1(t_0) > d$, $u_2(t_0) = d$, and $\dot{u}_2(t_0) = 0$. There exists $\varepsilon > 0$ such that R(t) < 0 for all $t \in]t_0 \varepsilon$; $t_0 + \varepsilon[$;
 - (c) the end of a sticking contact: $u_1(t_0) = d$, $\dot{u}_1(t_0) < 0$, $u_2(t_0) = d$, and $\dot{u}_2(t_0) = 0$. There exists $\varepsilon > 0$ such that R(t) < 0 for all $t \in]t_0 - \varepsilon$; $t_0[$ and R(t) = 0 for all $t \in]t_0$; $t_0 + \varepsilon[$.

2.2.2 Periodic solutions with one sticking phase per period (1-SPP)

The main results of this chapter is concerned with the possible existence and computation of 1-SPP.

Definition 2.3 — One sticking phase per period solution. A function **u** is called a 1-SPP, if there exists $0 < \tau < T$ such that **u** is a *T*-periodic solution to (2.1) with one sticking phase per period, that is to say up to a time translation that

- 1. $u_2 = d$ on $[0; \tau]$,
- 2. $u_2 < d$ on $]\tau; T[$, and
- 3. $\mathbf{u}(T) = \mathbf{u}(0)$ and $\dot{\mathbf{u}}^{-}(T) = \dot{\mathbf{u}}^{-}(0)$.

Condition (2) above can be relaxed to $u_2(t) \le d$ on $]\tau; T[$ only. This yields admissible periodic solutions with potentially many grazing contacts and sticking phases.

In order to find and characterize all 1-SPP, the following notations are needed:

$$\mathbf{P} = \begin{bmatrix} P_{11} & P_{12} \\ P_{21} & P_{22} \end{bmatrix}, \qquad \mathbf{B} = \mathbf{P}^{-1} = \begin{bmatrix} B_{11} & B_{12} \\ B_{21} & B_{22} \end{bmatrix}, \qquad (2.7)$$

$$\Phi_j(s) = \frac{\sin(\omega_j s)}{\omega_j (1 - \cos(\omega_j s))} = \frac{1}{\omega_j} \cot\left(\frac{\omega_j s}{2}\right),\tag{2.8}$$

$$a_{kj} = -P_{kj}B_{j1}, \quad b_{kj} = \frac{a_{kj}}{\omega_j}, \quad \alpha_j = b_{1j} - b_{2j}, \quad \beta_j = b_{1j},$$
 (2.9)

$$w_k(s) = \sum_{j=1}^2 a_{kj} \Phi_j(s) = \sum_{j=1}^2 b_{kj} \cot\left(\frac{\omega_j s}{2}\right),$$
(2.10)

with j = 1, 2 and k = 1, 2. Note that the interaction coefficients a_{kj} in [25] and in this work have opposite sign. If a 1-SPP exists, then there is only one control parameter, the duration of the free flight $s = T - \tau$, which uniquely determines the 1-SPP through Theorem 2.4. The initial data and the period *T* are functions of *s*. Conversely, such initial data may generate a ghost solution [38] if u_2 becomes greater than *d* during the free flight.

The natural way to obtain periodic solutions for System (2.1) is to look for the fixed points of the associated first return map (FRM). From a quasi-explicit expression of the FRM, it appears that closed forms can be obtained except for an unknown parameter: the free flight time which is a root of an explicit function h. The roots of h are carefully explored to obtain a countably infinite set of initial data which yields 1-SPP if and only if the constraint $u_2 < d$ is satisfied during the free flight. This approach generalizes to the prestressed case $d \le 0$.

Theorem 2.4 — 1-SPP characterization. Assume u is a 1-SPP of System (2.1), then:

1. The duration of the free flight s > 0 is necessarily a root of:

$$h(s) = w_1(s) - w_2(s) = \sum_{j=1}^{2} \alpha_j \cot\left(\frac{\omega_j s}{2}\right) = 0,$$
(2.11)

2. the solution **u** corresponds to the initial data

$$[u_1(0), u_2(0), \dot{u}_1(0), \dot{u}_2(0)] = [d, d, v, 0], \text{ where } v = v(s) = d/w_1(s).$$
 (2.12)

- 3. The period T of **u** is a function of s: $T(s) = s + \tau(v(s))$, where τ is defined in (2.5).
- 4. The orbit is symmetric: $\mathbf{u}(\theta + t) = \mathbf{u}(\theta t)$, $\forall t$, where $\theta := \tau + s/2$.

The proof of this theorem is given in Sections 2.5.1 and 2.5.2.

Remark 2.5. Instead of *s*, the parameter characterizing a 1-SPP could be the velocity v of the first mass at the beginning of the sticking phase. Choosing v specifies all the initial data (2.12) at the beginning of the sticking phase for a 1-SPP and the 1-SPP for all time. Accordingly, there is

a one-to-one correspondence between the set of 1-SPP and the set of initial velocity $v = \dot{u}_1(t_0)$ which yields a 1-SPP. However, the 1-SPP closed-form expressions are simpler with *s* which is kept in the remainder.

Remark 2.6. The set of 1-SPP is at most countable and corresponds to a subset of roots of the analytic function $h(\cdot)$ defined in Equation (2.11).

The roots of the quasi-periodic function $h(\cdot)$ are the first quantities to be carefully investigated to find 1-SPP. In addition, the velocity of the first mass at the beginning of the sticking phase has to be positive, see Theorem 2.17. The sign of this velocity is governed by the sign of $w_1(s)$.

The sticking phase can now be exactly computed. Without loss of generality, assume $t_0 = 0$. The end of the sticking phase is the beginning of the free flight. By denoted $\underline{\mathbf{u}}$ the free flight portion of the solution with $\tau = \tau(s)$ leads to

$$(\underline{\mathbf{u}},\underline{\dot{\mathbf{u}}})(\tau) := (\mathbf{u},\underline{\dot{\mathbf{u}}})(\tau)$$
(2.13)

 $\mathbf{M}\mathbf{\underline{\ddot{u}}}(t) + \mathbf{K}\mathbf{\underline{u}}(t) = 0, \quad \forall t \in]\tau; T[.$ (2.14)

A solution to Equations (2.13)-(2.14) is a physically admissible solution to System (2.1) if it satisfies the constraint :

$$\underline{u}_2(t) < d, \quad \tau < t < T.$$
 (2.15)

If condition (2.15) is violated, then the 1-SPP is not admissible: this is a "ghost" solution [38]. Hence, introducing the following sets:

$$Z = \{s > 0, h(s) = 0\},$$
(2.16)

$$Z^{-} = \{ s \in Z \text{ and } w_1(s) < 0 \},$$
(2.17)

$$Z^{0} = \{s \in Z \text{ and } w_{1}(s) = 0\},$$
(2.18)

$$Z^{+} = \{ s \in Z \text{ and } w_{1}(s) > 0 \},$$
(2.19)

$$Z^{\text{ad}} = \{s \in Z^+ \text{ and } (2.15) \text{ is satisfied}\} \subset Z^+$$

$$(2.20)$$

and $Z = Z^+ \cup Z^0 \cup Z^-$, the admissible free flight times *s* belongs to Z^+ which also corresponds to the "admissible" initial data. Furthermore, from the admissible initial data, the set of admissible 1-SPP has a one-to-one correspondence with Z^{ad} . Is Z^{ad} empty or not? Answering is not straightforward due to the global constraint (2.15) during the full free flight. However, we can quantify the size of Z^+ which leads to solutions satisfying (2.15) at least near the sticking phase. The following assumption 2.7 is needed to avoid that $Z^+ = \emptyset$, see Section 2.6 below.

Assumption 2.7. det
$$\begin{pmatrix} \alpha_1 & \alpha_2 \\ \beta_1 & \beta_2 \end{pmatrix} \neq 0.$$

Notice that if Assumption 2.7 is violated and $(\alpha_1, \alpha_2) \neq (0, 0), (\beta_1, \beta_2) \neq (0, 0)$ then $h(\cdot)$ and $w_1(\cdot)$ have the same roots, i.e. $Z = Z^0$, and $Z^+ = \emptyset$ and there is no 1-SPP.

Theorem 2.8 — Countable infinity of Z^+ . If $\omega_1/\omega_2 \notin \mathbb{Q}$ then Z is countably infinite. Moreover, if Assumption 2.7 holds, Z^+ is also countably infinite.

The proof of this theorem is exposed in Section 2.6. It is straightforward to show that Z is countably infinite when $\omega_1/\omega_2 \notin \mathbb{Q}$ since $h(\cdot)$ is quasi-periodic with many vertical asymptotes. The challenging part in Theorem 2.8 is to prove that Z^+ is also infinite. Incidentally, it turns out that Z^- is also infinite and more precisely that $\operatorname{card}(Z^+ \cap [0; A]) \sim \operatorname{card}(Z^- \cap [0; A])$ for large A. It means that many roots of h do not correspond to 1-SPP. Not only a 1-SPP is a rare object but among the roots of the function h, only a few correspond to an admissible 1-SPP.

In the next Section, the procedure to find 1-SPP is detailed.

2.3 Examples

To construct 1-SPP, Theorems 2.17 and 2.4 are interpreted as follows: let s > 0 satisfy h(s) = 0 and $w_1(s) > 0$. Such *s* is a candidate to construct a 1-SPP **u** to System (2.1) corresponding to the initial data

$$[u_1(0), u_2(0), \dot{u}_1(0), \dot{u}_2(0)]^{\top} = [d, d, +v, 0]^{\top} \text{ where } v = d/w_1(s),$$
(2.21)

with a sticking phase on $[0; \tau]$ and then a free-flight on $[\tau; \tau + s]$ with $\tau = \tau(s)$; more precisely:

- sticking phase for t ∈ [0; τ]: mass 2 sticks to the wall and mass 1 behaves as a 1-dof linear oscillator.
- free flight for $t \in]\tau; \tau + s[$: System (2.1a) is solved with "initial" data at time $\tau = \tau(s)$:

$$[u_1(\tau), u_2(\tau), \dot{u}_1(\tau), \dot{u}_2(\tau)]^{\top} = [d, d, -v, 0]^{\top}.$$
(2.22)

The condition $u_2(t) < d$ is to be checked on the interval $]\tau; \tau + s[$ to obtain a real solution to System (2.1). Otherwise, an impact emerges before $\tau + s$ and the assumption of a free flight is violated on $]\tau; \tau + s[$ so that the corresponding $\mathbf{u}(t)$ is not a 1-SPP.

Accordingly, building a 1-SPP requires two numerical steps:

- 1. Compute the roots of $h(\cdot)$: Figure 2.3 depicts the set of roots as the intersections of $h(\cdot)$ and the *x*-axis.
- Check the admissibility of the associated solution: check if v > 0 and if u₂(t) < d for all t ∈]τ(s); τ(s) + s[. From the symmetry of the solution during the free flight, it is sufficient to check u₂(t) < d for all t ∈]τ; τ + s/2[.



FIGURE 2.3 – $s \in Z = \{h(s) = 0\}$: red point: $s \in Z^{ad}$; blue points: $s \in Z^+$ but $s \notin Z^{ad}$; green points: $s \in Z^-$. The set of *s* corresponding to the admissible initial data are points in blue or red but only one point correspond to a 1-SPP: the red point.

First numerical examples are provided with $m_1 = m_2 = 1$ kg, $k_1 = k_2 = 1$ N/m. Hence, the two natural periods of the unconstrained linear System (2.1a) are $T_1 \approx 10.17$ s and $T_2 \approx 3.88$ s.

Figure 2.2 shows the simplest 1-SPP one can find: only one loop for the orbit of the second mass. This orbit is very smooth except at one point corresponding to the whole sticking phase. At this *sticking point*, only a $C^{1.5}$ -regularity is achieved as discussed in Section 2.4.3. Various examples featuring other responses are introduced in Figures 2.4 and 2.5. Many roots of *h*



FIGURE 2.4 – Admissible 1-SPP with $k_1 = k_2 = 1$; $m_1 = 100$; $m_2 = 1$, initial data U(0) = [1, 1, 0.7070682, 0]: $s \approx 34.412$ s and $\tau \approx 20.804$ s. (a) Orbits. (b) Displacements

belonging to Z^+ do not correspond to 1-SPP. For instance, for $s \approx 17.97 \in Z^+$, the free-flight is not acceptable since the second mass penetrates the rigid obstacle, as pictured in Figure 2.6. The condition $s \in Z^+$ only stipulates that the non-penetration constraint (2.15) is satisfied near the sticking phase. Although Z^+ is countably infinite, it is challenging to find the set $Z^{ad} \subset Z^+$



FIGURE 2.5 – 1-SPP Orbits. (a) $k_1 = 1$, $k_2 = 10k_1$ and $m_1 = m_2 = 1$, initial data U(0) = [1, 1, 0.6525913, 0]. (b) $k_1 = 1, k_2 = 100k_1$ and $m_2 = 1, m_1 = 100m_2$, initial data U(0) = [1, 1, 0.6409175, 0].



FIGURE 2.6 – Non-admissible 1-SPP with initial data U(0) = [1, 1, 1.4447006, 0]: mass 2 penetrates the wall during the free flight. (a) Orbits. (b) Displacements.

yielding 1-SPP. At first, for large *s*, the free-flight lasts a long period of time and the possibility that u_2 exceeds *d* seems to increase. Nevertheless, 1-SPP with large *s* are found in Figures 2.4-2.5.

2.4 Sticking contact

This section is devoted to the mathematical proof of Theorem 2.17 concerned with the necessary and sufficient conditions on the occurrence of a sticking phase¹. The theory for such systems with impacts can be found in [1, 5, 7, 6, 40].

In order to experience a sticking phase, the first necessary condition is a zero-velocity of the contacting mass when the gap is being closed, see below or [5, 7]. Then, the sticking phase holds whenever there is a positive force generated from mass 1 on mass 2. This force is explicit through

¹This theorem was also stated in [19].
the last equation of (2.1a). Next, the sticking system is presented. Via Lemma 3.16, it is clear that the energy of the unconstrained linear free flight system is conserved by the sticking system which becomes simply a 1-dof problem: the closed-form solution as well as the explicit duration of sticking phase are obtained.

The solution is analytic away from the beginning and the end of the sticking phase [5]. It can be seen from Figure 2.2(c) that the contact force is only Lipschitz at the beginning and at the end of the sticking phase. Thus, \ddot{u}_2 is a Lipschitz function, so the function u_2 belongs to the Sobolev space $W^{3,\infty}$. The smoother function u_1 belongs to $C^4 \cap W^{5,\infty}$: for both functions u_1 and u_2 , the singularity is located on the boundary of the sticking phase. The orbit { $(u_2(t), \dot{u}_2(t)), t \in \mathbb{R}$ } has only a $C^{1.5}$ -regularity at the sticking point. The singularity $C^{1.5}$ is visible in Figure 2.2 and is caused by the zero velocity and zero acceleration of mass 2 exactly when the sticking phase starts and ends. This loss of regularity is explored at the end of this Section. The prestressed case $d \leq 0$ is covered in Section 2.7 with the occurrence of sticking phases of infinite duration.

2.4.1 Occurrence of a sticking phase

The proof for the necessary and sufficient conditions for a sticking phase to occur as stated in Theorem 2.17 is now given for d > 0. The case d < 0 is explained just after this proof.

Proof. The right and left analyticity of the solution for the perfect elastic rebound is used (Proposition 19 in [5]). Notice that the condition of a closed contact, *i.e.* $u_2(0) = d$, with zero velocity $\dot{u}_2^-(0) = 0$ is mandatory. Otherwise $\dot{u}_2^-(0) > 0$, $\dot{u}_2^+(0) = -\dot{u}_2^-(0) < 0$ and the mass immediately leaves the wall, that is $u_2(t) < d$ for t > 0 and $t \approx 0$ such that there is no sticking phase. The second equation of System (2.1a) is rewritten with the aforementioned initial data for mass 2 only:

$$m_2 \ddot{u}_2(t) = k_2(u_1(t) - u_2(t)) + R(t),$$

$$u_2(0) = d, \quad \dot{u}_2^-(0) = 0, \quad R(t) \le 0.$$
(2.23)

During a sticking phase $u_2(t) \equiv d$ so $\ddot{u}_2(t) \equiv 0$ and Equation (2.23) yields the relation between the reaction R(t) and the displacement $u_1(t)$:

$$R(t) = k_2(d - u_1(t)), (2.24)$$

which is non-positive if and only if

$$u_1(t) \ge d. \tag{2.25}$$

Let us emphasize that Condition (2.25) is important in this work even though there is no unilateral

constraint on u_1 in the formulation of System (2.1). During the sticking phase, Inequality (2.25) is satisfied. As a consequence, $\dot{u}_2(0) = 0$ is not a sufficient condition to ensure the existence of a sticking phase starting at time t = 0. Various situations depending on the state $(u_1(0), \dot{u}_1(0))$ should be scrutinized:

 $u_1(0) < d$ The left-hand side of (2.23) is strictly negative and so is $\ddot{u}_2^+(0)$. There is no sticking phase. More precisely, u_2 is a piecewise analytic function [5] and its Taylor series in the right neighbourhood of 0 is:

$$u_2(t) = d + t^2 \frac{\ddot{u}_2^+(0)}{2} + O(t^3) < d.$$
(2.26)

- $u_1(0) > d$ Since u_1 is continuous, it remains larger than d in a right neighbourhood $[0; \varepsilon_2[$ of t = 0. Thus, there is a positive force $F(t) = k_2(u_1(t) - u_2(t))$ acting on mass 2, and by Newton's third law, there exists a reaction R(t) such that R(t) = -F(t). Substitution into (2.23) yields $\ddot{u}_2(t) = 0$, $\forall t \in [0; \varepsilon_2[$. Hence $u_2(t) = d$, $\forall t \in [0; \varepsilon_2[$, *i.e.* a sticking phase emerges.
- $u_1(0) = d$ In this case one has $\ddot{u}_2^+(0) = 0$ and there are three possibilities for the velocity of mass 1:
 - 1. If $\dot{u}_1(0) > 0$, $u_1(t)$ becomes immediately larger than *d* for t > 0 small enough. This is similar to the previous case where a sticking phase occurs.
 - 2. If $\dot{u}_1(0) < 0$ then $u_1(t)$ becomes immediately smaller than *d* and no sticking phase occurs. More precisely from Equation (2.23), $m_2\ddot{u}_2^+(0) = k_2(\dot{u}_1(0) \dot{u}_2(0)) < 0$ and the Taylor series of u_2 in the right neighbourhood of 0 is

$$u_2(t) = u_2(0) + t\dot{u}_2(0) + \frac{t^2}{2}\ddot{u}_2^+(0) + \frac{t^3}{6}\ddot{u}_2^+(0) + O(t^4)$$
(2.27)

$$= d + 0 + 0 + t^{3} \frac{\ddot{u}_{2}^{+}(0)}{6} + O(t^{4}) < d.$$
(2.28)

3. If $\dot{u}_1(0) = 0$, then $\ddot{u}_1^+(0) = -k_1 d/m_1 < 0$. Thus $m_2 \ddot{u}_2^+(0) = k_2 (\dot{u}_1(0) - \dot{u}_2(0)) = 0$ and $m_2 u_2^{(4)+}(0) = k_2 (\ddot{u}_1^+(0) - \ddot{u}_2^+(0)) < 0$. Similarly, a Taylor series of u_2 in the right neighbourhood of 0 shows that $u_2(t) < d$ for t > 0 and $t \approx 0$. Thus, there is no sticking phase.

Note that only the last case $u_1(0) = d$ and $\dot{u}_1(0) = 0$ crucially depends on the sign of d. It is further discussed in Section 2.7 when $d \le 0$. Moreover, all piecewise analytic solutions presented above preserve energy. It is clear for the grazing case since $R \equiv 0$. When sticking occurs, energy conservation is a consequence of Lemma 3.16. In conclusion, every introduced case corresponds to the unique solution preserving energy [5].

Remark 2.9. The duration of the sticking phase directly relates to the reaction R(t). The sign of R(t) is given by the sign of $d - u_1(t)$ via (2.24). Thus, for d < 0, an infinite sticking phase

appears when the solution u_1 of the sticking equation (2.31) satisfies $u_1(t) \ge d = -|d|$ for all time, which is possible as detailed in Section 2.7.

2.4.2 Sticking system

It is now shown that the solution with a sticking phase expounded in Section 2.4.1 is the unique solution which preserves the total energy, see [5].

The notion of sticking system is explained in [7]. It is the reduced system during the sticking phase. Since the last mass is at rest, the system loses one degree of freedom. From the previous developments, the sticking system complemented by the initial data at the beginning of a sticking phase is explicitly derived as

$$m_1\ddot{u}_1 + (k_1 + k_2)u_1 - k_2u_2 = 0, \ u_1(0) = d, \ \dot{u}_1(0) = v > 0,$$
(2.29)

$$m_2 \ddot{u}_2 = 0, \ u_2(0) = d, \ \dot{u}_2(0) = 0.$$
 (2.30)

The sticking system (3.23), (3.24) becomes simply a sticking equation (2.31). The initial data for mass 1 has to be clarified. If $u_1(0) > d$ then this inequality is also valid locally in the past, and the sticking phase exists before t = 0. If $u_1(0) = d$ and $\dot{u}_1(0) > 0$, then there exists $\eta > 0$ such that $u_1(t) < d$ for $-\eta < t < 0$ so there is no sticking phase just before t = 0, in other words, t = 0 is the beginning of the sticking phase. The grazing contact case $\dot{u}_1(0) = 0$ and the case where constraint (2.15) is violated, $\dot{u}_1(0) < 0$, do not have to be considered.

During free-flight, there is a total energy (2.1d) for the symmetric system (2.14). However, during the sticking phase, system (3.23-3.24) is not symmetric. Accordingly, the question of conservation of energy during the sticking phase is not obvious. It is proven in the next lemma.

Lemma 2.10. The solution to Equations (3.23-3.24) conserves the energy E (2.1d).

Proof. Assume that t = 0 is the beginning of a sticking phase and $t = \tau$, the end. During this sticking phase on the interval $[0; \tau]$, the governing equations are

$$m_1\ddot{u}_1 + (k_1 + k_2)u_1 = k_2d, \tag{2.31}$$

$$u_2 = d. \tag{2.32}$$

The first equation conserves the energy around the new equilibrium $\overline{u}_1 = k_2 d/(k_1 + k_2)$:

$$\mathbf{E}_{1}(u_{1}(t), \dot{u}_{1}(t)) = m_{1}\dot{u}_{1}^{2}(t) + (k_{1} + k_{2})(u_{1}(t) - \overline{u}_{1})^{2} = \mathbf{E}_{1}(u_{1}(0), \dot{u}_{1}(0)).$$
(2.33)

Moreover, since $u_1(t) = (u_1(t) - \overline{u}_1) + \overline{u}_1$ and $u_2(t) = d$, an easy computation yields:

$$(k_1 + k_2)u_1^2(t) = (k_1 + k_2)(u_1(t) - \overline{u}_1)^2 + 2k_2u_1(t)u_2(t) + C,$$
(2.34)

and $C = -(k_1 + k_2)\overline{u}_1^2$. The energy of System (2.1) can be calculated. Since $\dot{\mathbf{u}}$ is continuous along a sticking phase, the exponents \pm are dropped:

$$E(\mathbf{u}(t), \dot{\mathbf{u}}(t)) = \dot{\mathbf{u}}^{\mathsf{T}}(t)\mathbf{M}\dot{\mathbf{u}}(t) + \mathbf{u}^{\mathsf{T}}(t)\mathbf{K}\mathbf{u}(t)$$
(2.35)

$$= m_1 \dot{u}_1^2(t) + (k_1 + k_2)u_1^2(t) + m_2 \dot{u}_2^2(t) + k_2 u_2^2(t) - 2k_2 u_1(t)u_2(t)$$
(2.36)

$$= m_1 \dot{u}_1^2(t) + (k_1 + k_2)(u_1(t) - \overline{u}_1)^2 + 2k_2 u_1(t)u_2(t) + C + k_2 d^2 - 2k_2 u_1(t)u_2(t)$$
(2.37)

$$= \mathbf{E}_{1}(u_{1}(t), \dot{u}_{1}(t)) + C + k_{2}d^{2} = \mathbf{E}_{1}(u_{1}(0), \dot{u}_{1}(0)) + C + k_{2}d^{2} = E(\mathbf{u}(0), \dot{\mathbf{u}}(0)).$$
(2.38)

This ends the proof: the total energy of the system is constant for all 1-SPP.

The sticking system is now solved and the sticking time is explicitly exhibited: this is an interesting feature of our 2-dof mechanical system. The 1-dof linear oscillator problem with a constant force (2.31) has the explicit solution

$$u_1(t) = A\cos(\omega t + \phi) + \frac{k_2}{k_1 + k_2}d$$
 where $\omega = \sqrt{\frac{k_1 + k_2}{m_1}}.$ (2.39)

The expression of the constants A and ϕ stems from the initial condition $[u_1(0), \dot{u}_1(0)]^{\top} = [d, v]^{\top}$ as follows

$$A = \frac{k_1 d}{(k_1 + k_2)\cos(\phi)} \quad \text{and} \quad \phi = -\arctan(\xi v) \quad \text{with} \quad \xi = \frac{\sqrt{(k_1 + k_2)m_1}}{k_1 d}$$
(2.40)

and τ is the first positive time satisfying $u_1(\tau) = d$, that is $\tau = 2 \arctan(\xi v)/\omega$. This is due to the symmetry of the solution to the 1-dof Problem (2.31) with respect to the u_1 axis in the plane $(u_1, \dot{u}_1), u_1(\tau) = d$ and $\dot{u}_1(\tau) = -v$ which also means that τ is the end of the sticking phase via Theorem 2.17.

2.4.3 1.5-singularity at the sticking point

The following Proposition states precisely the regularity near a sticking phase, essentially C^2 and almost C^3 . The lower $C^{1.5}$ -regularity of the orbit is obtained at the end of the Section.

Proposition 2.11 — **Regularity of solutions.** Assume $\mathbf{u}(\cdot)$ is a solution of System (2.1) on $[T_0; T_1]$ with only a sticking phase on $[0; \tau]$ and a free flight elsewhere with $T_0 < 0 < \tau < T_1$. Then $u_1 \in C^4([T_0; T_1]) \cap W^{5,\infty}([T_0; T_1])$ and $u_2 \in C^2([T_0; T_1]) \cap W^{3,\infty}([T_0; T_1])$.

Proof. Away from the strict beginning and end of the sticking phase, the solution is regular: analytic outside $[0; \tau]$, u_1 is analytic and u_2 is constant inside $]0; \tau[$. The solution regularity at

t = 0 and $t = \tau$ is of higher interest. Only the case t = 0 is considered since the other case $t = \tau$ is quite similar. The initial data at t = 0 is $[\mathbf{u}(0)^{\top}, \dot{\mathbf{u}}(0)^{\top}]^{\top} = [d, d, v, 0]^{\top}$. The second Equation within (2.1a) is

$$m_2 \ddot{u}_2(t) = k_2(u_1(t) - u_2(t)) + R(t).$$
(2.41)

During the sticking phase, $0 < t < \tau$, $u_2(t) = d$ so $\ddot{u}_2(t) = 0$ and $\ddot{u}_2^+(0) = 0$ and before the sticking phase, t < 0, since $u_2(t) < d$, R(t) = 0 and $\lim_{0 > t \to 0} u_2(t) = d = \lim_{0 > t \to 0} u_1(t)$ so from Equation (2.41) $\ddot{u}_2^-(0) = 0$, thus \ddot{u}_2 is continuous at time t = 0 and $\ddot{u}_2(0) = 0$. However, the third derivative of u_2 on the left of t = 0 does not vanish since $m_2\ddot{u}_2^-(0) = k_2(\dot{u}_1(0) - \dot{u}_2(0)) = k_2v > 0$ and \ddot{u}_2 is then bounded. Hence, $u_2 \in C^2([0;T]) \cap W^{3,\infty}([0;T])$.

The regularity of u_1 is investigated from the first Equation of (2.1a) which reads

$$m_1\ddot{u}_1 + (k_1 + k_2)u_1 = k_2u_2, \tag{2.42}$$

and shows that \ddot{u}_1 and u_2 have the same regularity. Accordingly, $\ddot{u}_1 \in C^2([0;T]) \cap W^{3,\infty}([0;T])$ that is $u_1 \in C^4([0;T]) \cap W^{5,\infty}([0;T])$.

We now prove the $C^{1.5}$ -regularity of the orbit without using explicit formula.

Proof. The $C^{1.5}$ -smoothness of the projection of the orbit on the last component, more precisely the regularity of the set

$$\Gamma_2 = \{\gamma(t) = (u_2(t), \dot{u}_2(t)), \ 0 \le t \le T\} \subset \mathbb{R}^2$$
(2.43)

is explored. By *T*-periodicity, this parametrization is defined for all time. During the sticking phase, the last mass rests against the foundation, $\gamma(t) = \gamma(0) = (d, 0)$, $\dot{\gamma}(t) = (0, 0)$ for $0 \le t \le \tau$ and the parametrization is then singular. Instead, a regular parametrization of Γ_2 is thus proposed in the form $\tilde{\gamma}(t) = \gamma(t - \tau)$, $\tau \le t \le T$. In other words, $\tilde{\gamma}$ is γ where the sticking phase has been removed. Also, $\tilde{\gamma}$ is defined for all time through *s*-periodicity with $s = T - \tau$. The set $\tilde{\Gamma}_2 = \tilde{\gamma}([0; s])$ is exactly Γ_2 . The curve is analytic away from the sticking point (d, 0). A precise study of $\tilde{\gamma}(t)$, $|t| < \varepsilon$ should now be undertaken for $\varepsilon > 0$ sufficiently small. To this end, the left and right derivatives are computed since the solution is left and right analytic at the sticking point [5]:

$$\frac{\mathrm{d}^k}{\mathrm{d}t^k}\tilde{\gamma}^-(0) = \frac{\mathrm{d}^k}{\mathrm{d}t^k}\gamma^-(0), \quad \frac{\mathrm{d}^k}{\mathrm{d}t^k}\tilde{\gamma}^+(0) = \frac{\mathrm{d}^k}{\mathrm{d}t^k}\gamma^+(\tau).$$
(2.44)

To compute the successive left and right derivatives, the ODE

$$m_2 \ddot{u}_2(t) = k_2(u_1(t) - u_2(t)) \tag{2.45}$$

is used just before the sticking phase and just after the sticking phase. Recall that $u_1(0) = d$ and $\dot{u}_1(0) = v > 0$, $u_2(0) = d$ and $\dot{u}_2(0) = 0$, $u_1(\tau) = d$ and $\dot{u}_1(\tau) = -v < 0$, $u_2(\tau) = d$ and $\dot{u}_2(\tau) = 0$. The ODE gives $m_2\ddot{u}_2^-(0) = k_2(d-d) = 0$, $\ddot{u}_2^+(\tau) = 0$, so $\dot{\tilde{\gamma}}^{\pm}(0) = (0,0)$. The parametrization is still singular and higher derivatives of u_2 are computed by differentiating the ODE, that is

$$m_2\ddot{u}_2(t) = k_2(\dot{u}_1(t) - \dot{u}_2(t)) \tag{2.46}$$

$$\ddot{\gamma}^{-}(0) = (0,\beta), \quad \ddot{\gamma}^{+}(0) = (0,-\beta), \quad \beta = k_2 v/m_2 > 0$$
(2.47)

$$m_2 \ddot{u}_2(t) = k_2 (\ddot{u}_1(t) - \ddot{u}_2(t)) \tag{2.48}$$

$$\ddot{\tilde{\gamma}}^{-}(0) = (\beta, \delta), \quad \ddot{\tilde{\gamma}}^{+}(0) = (-\beta, \delta), \quad \delta = -k_2 k_1 d / (m_2 m_1) < 0$$
(2.49)

where the second derivative of u_1 comes from the equation $m_1\ddot{u}_1(t) = -k_1u_1(t) - k_2(u_1(t) - u_2(t))$: $\ddot{u}_1(0) = -k_1d/m_1 < 0$. The local behaviour at t = 0 is then for $\pm t > 0$:

$$\tilde{\gamma}(t) = (d,0) + \frac{1}{2}\operatorname{sign}(t)(0,\beta)t^2 + \frac{1}{6}(\operatorname{sign}(t)\beta,\delta)t^3 + O(t^4).$$
(2.50)

There are two singularities for this parametrization: the left and right expansions for $\pm t > 0$, and the more important $\dot{\tilde{\gamma}}(0) = (0, 0)$. To clearly identify the regularity of the curve at t = 0, a last change of variable is performed [38]: $\tau = \operatorname{sign}(t)t^2$ and $\hat{\gamma}(\tau) = \tilde{\gamma}(t)$ such that:

$$\hat{\gamma}(\tau) = (d,0) + \frac{1}{2}(0,\beta)\tau + \frac{1}{6}(\beta,\operatorname{sign}(\tau)\delta)|\tau|^{1.5} + O(\tau^2).$$
(2.51)

The $C^{1.5}$ -regularity is then identified since $\dot{\hat{\gamma}}(0) \neq (0,0)$ and this is optimal.

2.5 Building 1-SPP

This Section addresses the construction of the 1-SPP developed in Section 2.2.2. An explicit formula for τ is obtained and the set of admissible initial data is derived. The initial velocity of the first mass depends on the free flight time *s* and it is proven that *s* can be found in the infinite set of roots of $h(\cdot)$. The symmetry of the solutions is also discussed.

2.5.1 Initial data

Without loss of generality, the initial data is defined in the Poincaré section $u_2 = d$. The problem is to find a periodic function **u** generated by the initial data $[d, d, v, 0]^{\top}$ such that there is one sticking phase per period. As explained previously, *T* and τ are parameterized by *s*. The sticking solution and the sticking time $\tau > 0$ are calculated explicitly in Section 2.4.2.

By denoting $\mathbf{U} = [\mathbf{u}^{\top}, \dot{\mathbf{u}}^{\top}]^{\top}$, a free flight starts with the following initial data at time τ :

$$\mathbf{U}(\tau) = [d, d, -\nu, 0]^{\top}.$$
 (2.52)

It can be written as

$$\mathbf{U}(\tau) = \mathbf{SU}(0) \quad \text{where} \quad \mathbf{S} = \begin{bmatrix} \mathbf{I} & \mathbf{0} \\ \mathbf{0} & \mathbf{L} \end{bmatrix} \quad \text{and} \quad \mathbf{L} = \begin{bmatrix} -1 & 0 \\ 0 & 1 \end{bmatrix}.$$
(2.53)

Away from the sticking phase, system (2.1a) simplifies to

$$\mathbf{M}\ddot{\mathbf{u}} + \mathbf{K}\mathbf{u} = \mathbf{0}.\tag{2.54}$$

Through the change of variable $\mathbf{u} = \mathbf{Pq}$ where \mathbf{P} is defined in (2.7), Equation (2.54) becomes

$$\mathbf{I}\ddot{\mathbf{q}} + \mathbf{\Omega}^2 \mathbf{q} = \mathbf{0} \tag{2.55}$$

which features the following block matrix solution

$$\mathbf{Q}(t) = \begin{bmatrix} \mathbf{q}(t) \\ \dot{\mathbf{q}}(t) \end{bmatrix} = \mathbf{R}(t-\tau) \begin{bmatrix} \mathbf{q}(\tau) \\ \mathbf{q}(\tau) \end{bmatrix}, \quad \forall t \in]\tau; T[$$
(2.56)

where

$$\mathbf{R}(t) = \begin{bmatrix} \cos(t\,\Omega) & \Omega^{-1}\sin(t\,\Omega) \\ -\Omega\sin(t\,\Omega) & \cos(t\,\Omega) \end{bmatrix}.$$
(2.57)

We shall find the solution \mathbf{u} and the period T such that

$$\mathbf{Q}(T) = \mathbf{Q}(0). \tag{2.58}$$

By denoting $s = T - \tau$, Equation (2.58) projected onto modal coordinates reads

$$\mathbf{R}(s)\tilde{\mathbf{S}}\mathbf{Q}(0) = \mathbf{Q}(0) \text{ where } \tilde{\mathbf{S}} = [\mathbf{B}]\mathbf{S}[\mathbf{P}] \text{ and } [\mathbf{P}] = \begin{bmatrix} \mathbf{P} & \mathbf{0} \\ \mathbf{0} & \mathbf{P} \end{bmatrix}$$
(2.59)

which can be expressed as $(\mathbf{R}(s)\mathbf{\tilde{S}} - \mathbf{I})\mathbf{Q}(0) = \mathbf{0}$ with

$$\mathbf{R}(s)\tilde{\mathbf{S}} - \mathbf{I} = \begin{bmatrix} \cos(\Omega s) - \mathbf{I} & \Omega^{-1}\sin(\Omega s)\mathbf{BLP} \\ -\Omega\sin(\Omega s) & \cos(\Omega s)\mathbf{BLP} - \mathbf{I} \end{bmatrix}$$
(2.60)

where \mathbf{B} is defined in (2.7). The computations are similar to those introduced in [25]. This

similarity will be explained later through the relationship between 1-SPP and the one-Impact-Per-Period solutions (1-IPP) detailed in [25]².

The duration *s* is assumed not to be a period of the linear differential system, $s \notin \bigcup_{j=1}^{2} T_{j}\mathbb{Z}$ where $T_{j} = 2\pi/\omega_{j}$, j = 1, 2 are frequencies of the linear system. Then, the following quantities are well defined:

$$\mathbf{\Phi}(s) = (\mathbf{I} - \cos(\mathbf{\Omega}s))^{-1} \mathbf{\Omega}^{-1} \sin(\mathbf{\Omega}s), \tag{2.61}$$

$$\mathbf{w}(s) = \mathbf{P}\mathbf{\Phi}(s)\mathbf{B}\mathbf{L}\mathbf{e}_1, \quad \mathbf{e}_1 = (1,0)^{\mathsf{T}}, \tag{2.62}$$

$$w_1(s) = \mathbf{e}_1^{\mathsf{T}} \mathbf{w}(s). \tag{2.63}$$

The solution set of initial data yielding 1-SPP, possibly "ghost" solutions if constraint (2.15) is violated, is described explicitly via the following lemma:

Lemma 2.12. If $s \notin \bigcup_{i=1}^{2} T_{i}\mathbb{Z}$ then the system

$$\mathbf{R}(s)\tilde{\mathbf{S}}\mathbf{Q}(0) = \mathbf{Q}(0) \tag{2.64}$$

defines a one dimensional vector space parametrized by $c \in \mathbb{R}$ given in variables

$$\begin{bmatrix} \mathbf{u}(0) \\ \dot{\mathbf{u}}(0) \end{bmatrix} = \mathbf{P} \begin{bmatrix} \mathbf{q}(0) \\ \dot{\mathbf{q}}(0) \end{bmatrix} = c \begin{bmatrix} \mathbf{w}(s) \\ \mathbf{e}_1 \end{bmatrix}$$
(2.65)

Proof. Compute ker($\mathbf{R}(s)\mathbf{\tilde{S}} - \mathbf{I}$) by blocks (see [25]):

$$\begin{bmatrix} \cos(\Omega s) - \mathbf{I} & \Omega^{-1} \sin(\Omega s) \mathbf{BLP} \\ -\Omega \sin(\Omega s) & \cos(\Omega s) \mathbf{BLP} - \mathbf{I} \end{bmatrix} \sim \begin{bmatrix} \cos(\Omega s) - \mathbf{I} & \Omega^{-1} \sin(\Omega s) \mathbf{BLP} \\ \mathbf{0} & (\mathbf{L} + \mathbf{I}) \mathbf{P} \end{bmatrix}$$
(2.66)

because the matrix $(\mathbf{I} - \cos(\Omega s))^{-1}\mathbf{B}$ is invertible. Since $\dot{\mathbf{u}} = \mathbf{P}\dot{\mathbf{q}}$, the right lower block in (2.66) simplifies to $(\mathbf{L} + \mathbf{I})\dot{\mathbf{u}} = \mathbf{0}$, that is $\dot{\mathbf{u}} = c\mathbf{e}_1$ with $c \in \mathbb{R}$. Similarly, the upper block provides the expression $\mathbf{q} = c(\mathbf{I} - \cos(\Omega s))^{-1}\Omega^{-1}\sin(\Omega s)\mathbf{BL}\mathbf{e}_1$.

The assumption of lemma 2.12 is always valid and proven in the next lemma: the free flight duration of any 1-SPP is never a (multiple of a) period of the linear system.

Lemma 2.13. For a 1-SPP, the duration *s* of the free-flight is not a linear period: $s \notin \bigcup_{j=1}^{2} T_{j}\mathbb{Z}$.

Proof. It is proven that there is no 1-SPP involving a free-flight duration as a linear period. In other words, if $s \in \bigcup_{j=1}^{2} T_{j}\mathbb{Z}$, i.e. there exists $k, \ell \in \mathbb{Z}$ such that $s = kT_{1}$ or $s = \ell T_{2}$, then the corresponding solutions must be linear grazing orbits.

²Note that there is a change of sign in $\mathbf{w}(s)$ due to the coefficient $a_{kj} = -P_{kj}B_{j1}$ instead of $P_{kj}B_{jN}$ in [25].

Firstly, let us emphasize that all the components of the matrix $\mathbf{P} = (P_{ij})_{i,j=1}^2$ are nonzero in our case of study, a chain of two masses. The matrix \mathbf{P} of eigenvectors can be computed explicitly and has the following formula

$$\mathbf{P} = \begin{bmatrix} \frac{a + \sqrt{a^2 + b^2}}{\sqrt{m_1((a + \sqrt{a^2 + b^2})^2 + b^2)}} & \frac{a - \sqrt{a^2 + b^2}}{\sqrt{m_1((a - \sqrt{a^2 + b^2})^2 + b^2)}} \\ \frac{b}{\sqrt{m_2((a + \sqrt{a^2 + b^2})^2 + b^2)}} & \frac{b}{\sqrt{m_2((a - \sqrt{a^2 + b^2})^2 + b^2)}} \end{bmatrix}$$
(2.67)

where

$$a = \frac{1}{2} \left(\frac{k_2}{m_2} - \frac{k_1 + k_2}{m_1} \right), \quad b = \frac{k_2}{\sqrt{m_1 m_2}}.$$
(2.68)

Hence, $P_{ij} \neq 0$ for all i, j = 1, 2.

Secondly, one shows that if $s = kT_1$ then the periodic solutions with the free-flight duration *s* must be the first linear grazing orbit. Consider $s = kT_1$, then the matrix $\mathbf{R}(s)\mathbf{\tilde{S}} - \mathbf{I}$ becomes

$$\begin{bmatrix} 0 & 0 & 0 & 0 \\ 0 & C_2 - 1 & S_2/\omega_2 A_{21} & S_2/\omega_2 A_{22} \\ 0 & 0 & A_{11} - 1 & A_{12} \\ 0 & -\omega_2 S_2 & C_2 A_{21} & C_2 A_{22} - 1 \end{bmatrix}$$
(2.69)

where $C_2 = \cos(2\pi\omega_2/\omega_1)$, $S_2 = \sin(2\pi\omega_2/\omega_1)$, and $\mathbf{A} = \mathbf{BLP}$. With the assumption of no internal resonance, $C_2 \neq 0$, $C_2 - 1 \neq 0$ and $S_2 \neq 0$. Hence, after simple calculations, the last three equations become

$$\begin{bmatrix} C_2 - 1 & S_2/\omega_2 A_{21} & S_2/\omega_2 A_{22} \\ 0 & A_{11} - 1 & A_{12} \\ -\omega_2 S_2 & C_2 A_{21} & C_2 A_{22} - 1 \end{bmatrix} \sim \begin{bmatrix} C_2 - 1 & S_2/\omega_2 A_{21} & S_2/\omega_2 A_{22} \\ 0 & A_{11} - 1 & A_{12} \\ 0 & A_{21} & A_{22} + 1 \end{bmatrix}$$
(2.70)

The last two equations can be rewritten as $(\mathbf{A} + \mathbf{L})\dot{\mathbf{q}}(0) = \mathbf{0}$ or $(\mathbf{LP} + \mathbf{PL})\dot{\mathbf{q}}(0) = \mathbf{0}$. Since $\mathbf{LP} + \mathbf{PL} = \mathbf{diag}(-2P_{11}, 2P_{22})$, where P_{11} and P_{22} are nonzero, this yields $\dot{\mathbf{q}}(0) = \mathbf{0}$. By substituting into the second equation, it follows that $q_2(0) = 0$ or $u_1(0) = -B_{22}/B_{21}d$. Therefore, the initial data is $[u_1(0), d, 0, 0]$ which corresponds to the first linear grazing orbit.

Similarly, if $s = \ell T_2$ then the periodic solutions with the free-flight duration *s* must be the second linear grazing orbit. One obtains a similar expression $(\mathbf{A}-\mathbf{L})\dot{\mathbf{q}}(0) = \mathbf{0}$ or $(\mathbf{LP}-\mathbf{PL})\dot{\mathbf{q}}(0) = \mathbf{0}$.

Since

$$\mathbf{LP} - \mathbf{PL} = \begin{bmatrix} 0 & -2P_{12} \\ -2P_{21} & 0 \end{bmatrix}$$
(2.71)

where both P_{12} and P_{21} are both non-vanishing terms, then $\dot{\mathbf{q}}(0) = \mathbf{0}$. Consequently, the initial data obtained correspond to the second linear grazing orbit.

The parameter *c* is identified from the third row of (2.65), that is $c = \dot{u}_1^+(0) = v$. By expressing the initial condition, it follows that

$$\begin{bmatrix} u_1(0) \\ u_2(0) \end{bmatrix} = \begin{bmatrix} d \\ d \end{bmatrix} = v \begin{bmatrix} w_1(s) \\ w_2(s) \end{bmatrix} = c \mathbf{P} \mathbf{\Phi}(s) \mathbf{BL} \mathbf{e}_1.$$
(2.72)

System (2.72) simplifies to $w_1(s) = w_2(s)$ or

$$h(s) = w_1(s) - w_2(s) = \sum_{j=1}^{2} \alpha_j \cot\left(\frac{\omega_j s}{2}\right) = 0,$$
(2.73)

and the initial velocity of the first mass is found from

$$v = \frac{d}{w_1(s)}, \quad w_1(s) > 0.$$
 (2.74)

If $\omega_1/\omega_2 \notin \mathbb{Q}$, the function $h(\cdot)$ exhibits a countably infinite set of roots. Moreover the set of *s* such that h(s) = 0 and v(s) > 0 is also countably infinite by Theorem 2.8. The particular case when $\omega_1/\omega_2 \in \mathbb{Q}$ is discussed in Section 2.6.

2.5.2 Symmetry

To conclude the validation of Theorem 2.4, the symmetry of 1-SPP is proven.

Proof. Through periodicity, it is sufficient to check the symmetry of the solutions on one period only. The symmetry is satisfied during the sticking phase and the free flight. Since only the first mass oscillates during the sticking phase, the solution is symmetric. Let us check the symmetry of solutions during the free flight time $t \in [\tau; T]$ where $\mathbf{u}(\tau) = \mathbf{u}(T)$ and $\dot{\mathbf{u}}^+(\tau) = -\dot{\mathbf{u}}^-(T)$. Denoting $\theta = (T + \tau)/2$, it is sufficient to show that $\mathbf{u}(\theta + t) = \mathbf{u}(\theta - t)$, $\forall t \in I = [-s/2; s/2]$. Let \mathbf{z}_+ be the function defined on I such that $\mathbf{z}_+(t) = \mathbf{u}(\theta + t)$. Then \mathbf{z}_+ is a well defined smooth function on I with $\mathbf{z}_+(s/2) = \mathbf{u}(T)$ and $\dot{\mathbf{z}}_+(s/2) = \dot{\mathbf{u}}^-(T)$. Similarly, by defining the function $\mathbf{z}_-(t) := \mathbf{u}(\theta - t)$, for $t \in I$, it can be checked that $\mathbf{z}_-(s/2) = \mathbf{u}(\tau)$ and $\dot{\mathbf{z}}_-(s/2) = -\dot{\mathbf{u}}^+(\tau)$. Furthermore, both \mathbf{z}_+ and \mathbf{z}_- are solutions to the linear differential system $\mathbf{M}\mathbf{\ddot{z}} + \mathbf{K}\mathbf{z} = \mathbf{0}$ on I. Notice that \mathbf{z}_+ and \mathbf{z}_- have the same initial data $\mathbf{z}_+(s/2) = \mathbf{z}_-(s/2)$ and $\dot{\mathbf{z}}_+(s/2) = \dot{\mathbf{z}}_-(s/2)$. Hence, by the uniqueness of the initial value problem, it is deduced that $\mathbf{z}_+(t) \equiv \mathbf{z}_-(t)$ on *I*. This completes the proof on the symmetry of solutions.

2.5.3 Relationship between 1-SPP and 1-IPP

For this two-degree-of-freedom vibro-impact system, a relationship between one-sticking-phaseper-period and one-impact-per-period solutions [25] is exhibited. It clarifies the similarities and differences of such periodic solutions.

Consider the two Figures 2.2(a) and 2.2(b) showing a single loop in Γ_2 . Figures 2.7(a) and 2.7(b) are then obtained by "deleting" the sticking phase on the whole interval $]0; \tau[$ such that a 1-IPP solution [19, 25] is identified, where the jump occurs on mass 1 (instead of mass 2) when $u_1(0) = d$ as well as $u_2(0) = d$. Take note that $u_1(t) \le d$ for all time. It is important to note that this 1-IPP is "unique" is the sense that it satisfies $u_2(0) = d$; it is denoted 1-IPP_p in the remainder. The correspondence between 1-SPP and this particular 1-IPP_p is now detailed. To



FIGURE 2.7 – (a) Is a 1-IPP or a 1-SPP without sticking phase drawn? (b) Displacements

this end, generalized 1-SPP and 1-IPP_p, *ie* G1-SPP and G1-IPP_p respectively, are first defined: they are unconstrained 1-SPP and 1-IPP_p during the free flight and u_1 as well as u_2 might exceed d^3 . By definition, a G1-SPP **u** satisfies the following requirements:

1.
$$s \in Z$$
,

- 2. $T = s + \tau$ is the fundamental period with $\tau = \tau(s)$,
- 3. a sticking phase on $]0; \tau(s)[$ with $u_1(0) = d, \dot{u}_1(0) = v = v(s), u_2(0) = d, \dot{u}_2(0) = 0,$
- 4. a free flight on $]\tau; T[$ with $u_1(\tau) = d, \dot{u}_1(\tau) = -v, u_2(\tau) = d, \dot{u}_2(\tau) = 0,$ **Mu** + **Ku** = **0**.

³The G1-IPP in this chapter has a counterpart in [25]: it is a G1-IPP where the jump in velocity affects the second mass instead of the first mass here. As such, we know that there is a unique G1-IPP for all positive periods. This is the reason why the formulas in Section 2.5 are slightly different from [25]. The condition on mass 1 in Theorem 2.17 corresponds to an elastic impact for mass 1.

The one-to-one correspondence from Figure 2.2 to Figure 2.7 is formalized as

$$\widetilde{\mathbf{u}}(t) = \mathbf{u}(\tau + t), \quad 0 < t < s, \tag{2.75}$$

where $\tilde{\mathbf{u}}$ is taken *s*-periodic so that $\tilde{\mathbf{u}}^{-}(0) = \mathbf{u}(0)$, $\tilde{\mathbf{u}}^{+}(0) = \mathbf{u}(\tau)$. As a consequence, $\tilde{\mathbf{u}}$ satisfies

- 1. *s* is the fundamental period,
- 2. $\tilde{u}_1^{\pm}(0) = d, \, \dot{\tilde{u}}_1^-(0) = \dot{u}_1(0) = v, \, \dot{\tilde{u}}_1^+(0) = \dot{u}_1(\tau) = -v,$
- 3. $\tilde{u}_{2}^{\pm}(0) = d, \dot{\tilde{u}}_{2}^{\pm}(0) = 0,$
- 4. a free flight on $]0; s[: M\tilde{u} + K\tilde{u} = 0.$

We can check that $\tilde{\mathbf{u}}$ is a G1-IPP_p. The only surprising condition is $\dot{\tilde{u}}_2(0) = 0$ but zero velocity is automatically achieved by a G1-IPP [5, 25].

Proposition 2.14 — G1-SPP \Leftrightarrow G1-IPP_p. There is a one-to-one correspondence between G1-SPP with a sticking phase for the mass 2 and G1-IPP_p.

Proof. This is a brief sketch. G1-SPP \Rightarrow G1-IPP_p was explained previously. Conversely, from a given G1-IPP_p, it is possible to build a sticking phase as in the proof of Theorem 2.17 in Section 2.4 with a free flight duration *s* to then define a unique G1-SPP. \Box

The key parameter *s* appears to be simply the period of the associated G1-IPP_p. This proposition shows that the set *Z* corresponds exactly to the set of all G1-IPP_p. Let us state briefly the correspondence between Z^+ and Z^{ad} and the corresponding subset of all G1-IPP_p.

Concerning generalized solutions with a positive velocity at the impact (v > 0), it can be said that for all $s \in Z^+$ there exists a unique G1-SPP and a corresponding unique G1-IPP_p which has a physical initial data at the impact time (no violation of the constraint near the impact time). Conversely, if a G1-IPP_p is such that, at the impact time, the incoming velocity of mass 1 is positive then the period belongs to Z^+ which corresponds to a unique G1-SPP.

Finally, a 1-SPP, *i.e.* a G1-SPP satisfying the constraint $u_2(t) \leq d$ for all time, is in a unequivocal correspondence with a G1-IPP_p satisfying the same constraint. This condition is not the constraint to be a 1-IPP since the constraint for 1-IPP is on the mass 1. Figures 2.2 and 2.7 show a perfect and rare correspondence between a 1-SPP and a 1-IPP since the associated G1-IPP_p satisfies the two constraints $u_k(t) \leq d$ for all time and k = 1, 2. As a consequence, 1-SPP are isolated solutions. The reason lies in the fact that the space of G1-IPP is a one-dimensional manifold which intersects $\tilde{u}_2 = d$ on a discrete set such that the G1-IPP become isolated. Another possible consequence of Proposition 2.14, which is not further discussed here, is to prove the existence of 1-SPP through the proof of the existence of such particular G1-IPP.

2.6 The countable set Z^{\pm}

In order to find admissible solutions, the set Z^{\pm} are defined to encompass the corresponding set of initial data of admissible solution: \mathcal{V}_0^{ad} . Admissible initial data is defined by a constraint on the associated solution: $u_2(t) \leq d$ at least locally near the sticking phase. The problem to satisfy this constraint is only at the end and the beginning of the sticking phase but not for all time. Thus, a question of interest emerges: how large is this set "admissible initial data"? The set of admissible initial data \mathcal{V}_0^+ contains \mathcal{V}_0^{ad} . In this Section, the sets Z^+ and Z^- are proven to be countably infinite if some generic assumptions are fulfilled. The proof of Theorem 2.8 is similar for the two sets and only the proof for Z^+ is provided.

2.6.1 Z^{\pm} is infinite with no resonance

Before stating the main proof with $\omega_1/\omega_2 \notin \mathbb{Q}$, we start with Lemma 2.15 below. First, the orbit O is defined in the torus $\Pi = \mathbb{R}/2\pi\mathbb{Z} \times \mathbb{R}/2\pi\mathbb{Z}$:

$$O = \{ (x, y) = (\bar{t}, \overline{\rho t}) | t > 0 \}$$
(2.76)

where $\bar{t} = t + 2\pi\mathbb{Z}$ and ρ , a constant.

Lemma 2.15 — **Transversality and density.** Let *f* be a 2π -periodic continuously differentiable function from $[0; 2\pi[$ to $[0; 2\pi[$. For any irrational number ρ , if $(x_0, y_0 = f(x_0))$ located on the curve *C* defined by the graph of *f* satisfies the transversal condition between *C* and *O*, that is

$$\dot{f}(x_0) \neq \rho \tag{2.77}$$

then $\forall \varepsilon > 0$, $\exists t > 0$ such that $\overline{\rho t} = f(\overline{t})$ and $|\overline{t} - x_0| < \varepsilon$.

In other words, every point on the curve *C* at which the tangent is transverse to the orbit *O* is an accumulation point of $O \cap C$, see Figure 2.8. Precisely, the set $O \cap C$ is dense in $\{(x, f(x)) | \dot{f}(x) \neq \rho\}$. Moreover, for all A > 0, the set $O_A = \{(x, y) = (\bar{t}, \rho \bar{t}) | t > A\}$ shares the same property.

Proof. Assume $\rho > 0$, the cases $\rho = 0$ and $\rho < 0$ follow immediately.

Since $\dot{f}(x_0) \neq \rho$ and \dot{f} is continuous, there exists $\varepsilon_0 > 0$ small enough such that $\dot{f}(x) \neq \rho$ for all $x \in [x_0 - \varepsilon_0; x_0 + \varepsilon_0]$. Without loss of generality, assume that $\dot{f}(x) > \rho$, for all $x \in [x_0 - \varepsilon_0; x_0 + \varepsilon_0]$. Since O is dense in Π , $\forall \varepsilon > 0$, there exists $t_0 > 0$ such that $z = (\overline{t_0}, \overline{\rho t_0})$ belongs to O close enough to (x_0, y_0) , *i.e.* $|\overline{t_0} - x_0| < \varepsilon$ and $|\overline{\rho t_0} - y_0| < \varepsilon$: if z is on the curve Cthen t is chosen to be t_0 , else z is above the curve C, *i.e.* $\overline{\rho t_0} > f(\overline{t_0})$.

We will show that the orbit O intersects the curve C inside the box $]x_0 - \varepsilon_0; x_0 + \varepsilon_0[\times]y_0 - k\varepsilon_0; y_0 + k\varepsilon_0[$ where k is the maximum of $|\dot{f}|$ on $[x_0 - \varepsilon_0; x_0 + \varepsilon_0]$ as shown in Figure 2.9. For



FIGURE 2.8 – Density of $\mathbf{O} \cap \mathbf{H}$ in \mathbf{H}



FIGURE 2.9 – Zoom in the box $]x_0 - \varepsilon_0; x_0 + \varepsilon_0[\times]y_0 - k\varepsilon_0; y_0 + k\varepsilon_0[$ when $\dot{\psi}(x_0) > \rho$

this purpose, we use a line d_2 under the curve on the right of (x_0, y_0) . From $p = \min_{[x_0-\varepsilon_0; x_0+\varepsilon_0]} f$, the equation of the line d_2 with slope p passing through (x_0, y_0) is $y = p(x - x_0) + y_0$. The line dwith slope ρ passing through $(\overline{t_0}, \overline{\rho t_0})$ and associated to the orbit O is defined by $y = \rho(x - \overline{t_0}) + \overline{\rho t_0}$. Let $I(x_I, y_I)$ be the intersection of those two lines. Since $p > \rho$, we have

$$x_{I} = \frac{px_{0} - y_{0} - \rho \overline{t_{0}} + \overline{\rho t_{0}}}{p - \rho}.$$
(2.78)

Choosing ε small enough such that $\varepsilon < \varepsilon_0(|p - \rho|)/(\rho + k)$ implies $x_I \in [x_0 - \varepsilon_0; x_0 + \varepsilon_0]$.

Consider the two curves d_2 and *C* intersecting at (x_0, y_0) and satisfying $\dot{f}(x) > p$ for all $x \in [x_0 - \varepsilon_0; x_0 + \varepsilon_0]$. Since $p > \rho$, *d* intersects d_2 at *I*. Hence, there exists an intersection of *C* and *d* in the interval $]x_0; x_I[$. In other words, there exists t > 0 such that $\overline{\rho t} = f(\overline{t})$ and

 $|\bar{t} - x_0| < \varepsilon_0$. The proof for the case *z* is under the curve *C* is similar.

The proof of Theorem 2.8 starts by showing that the set $Z = \{s > 0, h(s) = 0\}$ is countably infinite. It is true for the set $\{(\omega_1 s, \omega_2 s), h(s) = 0\}$ and will be useful to prove that the set Z^+ of free flight times *s* with admissible initial velocity v(s) > 0 is also countably infinite.

Proof. Set $\varphi(t) = \cot(t/2)$, then $h(s) = \alpha_1 \varphi(\omega_1 s) + \alpha_2 \varphi(\omega_2 s)$ and $w_1(s) = \beta_1 \varphi(\omega_1 s) + \beta_2 \varphi(\omega_2 s)$ where $\beta_j = b_{1j}$, and α_j , b_{kj} are defined in Equation (2.9). For every $(x, y) \in \Pi = \mathbb{R}/2\pi\mathbb{Z} \times \mathbb{R}/2\pi\mathbb{Z}$, the two functions $H(x, y) = \alpha_1 \varphi(x) + \alpha_2 \varphi(y)$ and $W(x, y) = \beta_1 \varphi(x) + \beta_2 \varphi(y)$ correspond to h(s) = $H(\omega_1 s, \omega_2 s)$, $w_1(s) = W(\omega_1 s, \omega_2 s)$. In order to simplify, the sets $\mathbf{O} = \{(\overline{\omega_1 s}, \overline{\omega_2 s}) | s > 0\}$, $\mathbf{H} = \{(x, y) \in \Pi | H(x, y) = 0\}$, and $\mathbf{W} = \{(x, y) \in \Pi | W(x, y) = 0\}$ are defined on the torus Π ; \mathbf{W}^+ , \mathbf{W}^- are denoted as the domains of Π where W(x, y) > 0 and < 0, respectively.

The set **O** is equal to *O* with $\rho = \omega_2/\omega_1$. Consider the map $\gamma : \mathbb{R}^+ \to \mathbf{O}$, $s \mapsto (\overline{\omega_1 s}, \overline{\omega_2 s})$, then γ is bijective for $\omega_2/\omega_1 \notin \mathbb{Q}$ and

$$\gamma(Z) = \mathbf{O} \cap \mathbf{H} \tag{2.79}$$

$$\gamma(Z^+) = \mathbf{O} \cap \mathbf{H} \cap \mathbf{W}^+ \tag{2.80}$$

Hence, instead of proving the set Z is countably infinite, the stronger result $\overline{\mathbf{O} \cap \mathbf{H}} = \mathbf{H}$ is proven.



FIGURE 2.10 – (a) or (b) $\mathbf{H} \cap \mathbf{W}^+$ is the half of the red curve which lies in the grey domain; (c) The set $\mathbf{O} \cap \mathbf{H} \cap \mathbf{W}^+$ is the set of all intersections between the red curve and the orange lines within the grey domain

This implies $\mathbf{O} \cap \mathbf{H}$ is countably infinite. This stronger result shows that Z^+ is countably infinite by pointing out the density of $\mathbf{O} \cap \mathbf{H} \cap \mathbf{W}^+$ in $\mathbf{H} \cap \mathbf{W}^+$ and the countable infinity of $\mathbf{H} \cap \mathbf{W}^+$.

To show that $\overline{\mathbf{O} \cap \mathbf{H}} = \mathbf{H}$, assume $\alpha_2 \neq 0$, rewrite H(x, y) = 0 to have $y = \psi(x)$ where $\psi = \varphi^{-1}(r\varphi)$ and $r = -\alpha_1/\alpha_2$.

1. We show that $\dot{\psi} \neq \rho$ almost everywhere. Since ψ is an analytic function on $I = [0; 2\pi[$, so

is $\dot{\psi}$. After simplification, the derivative of ψ becomes

$$\dot{\psi} = \frac{r(1+\varphi^2)}{1+r^2\varphi^2} = \frac{1}{r} \left(1 + \frac{r^2 - 1}{1+r^2\varphi^2} \right)$$
(2.81)

which degenerates to a constant function for $r = \pm 1$. Otherwise, $\dot{\psi}$ is not a constant function and the set $\{x \in I | \dot{\psi}(x) = \rho\}$ is empty or countable. Hence, $\dot{\psi} \neq \rho$ holds almost everywhere. It is still true if $\alpha_2 = 0$ since H(x, y) becomes a periodic function of x, and **H** then degenerates to a vertical line in the torus Π .

2. Through Lemma 2.15 where $f = \psi$ is periodic of period 2π , the set O is \mathbf{O} where ρ is the ratio ω_2/ω_1 and $\mathbf{O} \cap \mathbf{H}$ is dense in $\{(x, y) \in \mathbf{H} | \dot{\psi}(x) \neq \rho\}$ follows. In addition, it is proven above that $\dot{\psi} \neq \rho$ almost everywhere, thus $\overline{\mathbf{O} \cap \mathbf{H}} = \mathbf{H}$, since \mathbf{H} is infinite, thus $\mathbf{O} \cap \mathbf{H}$ is countably infinite. In particular, there is a countably infinite set of s > 0 such that h(s) = 0.

To complete the proof, we show that Z^+ is countably infinite by proving that $\gamma(Z^+) = \mathbf{O} \cap \mathbf{H} \cap \mathbf{W}^+$ is countably infinite. In a similar manner, it is sufficient to show that $\overline{\mathbf{O} \cap \mathbf{H} \cap \mathbf{W}^+} = \overline{\mathbf{H} \cap \mathbf{W}^+}$ and $\mathbf{H} \cap \mathbf{W}^+$ is an infinite set. If $\beta_2 \neq 0$, denote $\kappa = -\beta_1/\beta_2$, the curve \mathbf{W} then corresponds to the function $y = \tilde{\psi}(x)$ where $\tilde{\psi} = \varphi^{-1}(\kappa\varphi)$ which has the same properties with ψ . The result still holds if $\beta_2 = 0$ since \mathbf{W} degenerates to the vertical line in Π . By Assumption 2.7, $r \neq \kappa$ and it follows that H and W cannot coincide and the determinant of the coefficient matrix of the homogeneous system

$$H(x, y) = 0 \tag{2.82}$$

$$W(x, y) = 0 \tag{2.83}$$

is nonzero. Therefore, it has a trivial solution $\varphi(x) = \varphi(y) = 0$, *i.e.* $(\bar{\pi}, \bar{\pi})$ is one intersection between **H** and **W**. Assumption 2.7 is optimal to have Z^+ is infinite. Otherwise, if Assumption 2.7 does not hold, $\mathbf{H} = \mathbf{W}$, thus $\mathbf{H} \cap \mathbf{W}^+ = \emptyset$ and Z^+ is empty.

Assumption 2.7 implies that $\dot{\psi}(\bar{\pi}) \neq \dot{\tilde{\psi}}(\bar{\pi})$ since $\dot{\psi}(\bar{\pi}) = r$ and $\dot{\tilde{\psi}}(\bar{\pi}) = \kappa$, thus the curves are transverse. Moreover, since the signs of the derivatives of ψ and $\tilde{\psi}$ depend on the signs of r and κ , respectively, ψ and $\tilde{\psi}$ are monotonic functions, in which case, $\mathbf{H} \cap \mathbf{W}^+$ is a half of the curve \mathbf{H} which lies in the region \mathbf{W}^+ (see Figures 2.10(a) and 2.10(b)).

As W^+ is an open set and $\overline{\mathbf{O} \cap \mathbf{H}} = \mathbf{H}$, then $\overline{\mathbf{O} \cap \mathbf{H} \cap W^+} = \overline{\mathbf{H} \cap W^+}$. It follows that $\mathbf{O} \cap \mathbf{H} \cap W^+$ is infinite (Figure 2.10c.). Hence, Z^+ is infinite which concludes the proof. \Box

2.6.2 Internal resonances

The situation is much simpler when the ratio ω_1/ω_2 is rational. All the functions involved are periodic with the same period and the set $\mathbf{O} \cap \mathbf{H}$ is finite or empty. Thus, the set of initial velocity $\{v(s), s \in Z\}$ is finite which also means that the set of generalized 1-SPP is finite. Z^+ can be an

empty set for instance if $Z = \emptyset$: with the parameters $\alpha_1 = 1$, $\alpha_2 = -1$, and $\omega_1/\omega_2 = 2$, the graph of function h(s) is depicted in Figure 2.11. As a consequence, 1-SPP do not exist in this case.



FIGURE 2.11 – Function h(s) when $\omega_1/\omega_2 = 2 \in \mathbb{Q}$. To be compared to Figure 2.3.

2.7 Prestressed structure

In this Section, the structure of the 1-SPP when $d \le 0$ is discussed. An argument on the occurrence of the sticking phase is stated in Proposition 2.16. Precisely, sticking phases of unbounded duration can arise besides the solutions with finite sticking phases, when the initial velocity of m_1 is zero. 1-SPP for d < 0 and d = 0 are also explored and illustrated through appropriate numerical examples.

Proposition 2.16. Assume $d \le 0$. Up to a time translation, periodic solutions with a permanent sticking phase have a one-to-one correspondence to the solutions with initial data:

$$u_2(0) = d, \quad \dot{u}_2(0) = 0, \quad d \le u_1(0) \le \overline{u}_1 := \frac{k_2}{k_1 + k_2} d, \quad \text{and} \quad \dot{u}_1(0) = 0.$$
 (2.84)

Otherwise, if

$$u_2(0) = d, \quad \dot{u}_2(0) = 0, \quad u_1(0) = d, \quad \text{and} \quad \dot{u}_1(0) > 0,$$
(2.85)

then a sticking phase with finite duration occurs.

The above proposition calls for a few comments:

• for d = 0, there is only one periodic solution with infinite duration displayed below in Figure 2.14.

- for d < 0, the set of periodic solutions with infinite duration is infinite (continuous set).
- It suffices to start at time t = 0 but not necessarily when u_1 reaches its minimum. More precisely, a solution has a sticking phase for all t > 0 if and only if the minimum of u_1 during the sticking phase is greater or equal to d. An easy computation of the 1-dof dynamics of u_1 during the sticking phase can be written with the engergy E_1 (see in the proof of Lemma 3.16) as follows.

$$u_2(0) = d, \ \dot{u}_2(0) = 0, \ \text{and} \ m_1(\dot{u}_1(0))^2 + (k_1 + k_2)(u_1(0) - \overline{u}_1)^2 \le \frac{(k_1d)^2}{k_1 + k_2}.$$
 (2.86)

Proof. The proof is a consequence of the proof of Theorem 2.17 together with Remark 2.9 and the 1-dof periodic dynamics of u_1 around \overline{u}_1 during the sticking phase. The conditions given in (2.84) (or (2.86)) for u_1 are just the conditions such that $u_1(t) \ge d$ during the period of the "sticking equation" (2.31) then $u_1(t) \ge d$ for all $t \ge 0$ and mass 2 remains sticked for all time in the future. For the periodic solution, up to a time translation, it is only assumed that u_1 reaches its minimum at t = 0.

In general, the conditions $u_1(0) > d$ or $\{u_1(0) = d \text{ and } \dot{u}_1(0) > 0\}$ are necessary to enjoy the existence of a sticking phase. However, the latter implies that $u_1(t)$ gets strictly smaller than d in the past and by the periodicity in the future, the sticking phase stops.

2.7.1 Strictly prestressed structure

The dynamics is explored with d < 0.

Sticking phase of finite duration

From Proposition 2.16, at the beginning of the sticking phase, the initial data is $[\mathbf{u}(0), \dot{\mathbf{u}}(0)]^{\top} = [d, d, v, 0]^{\top}$ where v > 0. It directly follows, from Equation (2.74) when d < 0, that the admissible initial data is found in the set Z^{-} instead of Z^{+} . Z^{-} is also countably infinite as stated in Theorem 2.8. In a manner similar to the case d > 0, an infinite set of admissible initial data is expected when d < 0.

A 1-SPP is depicted in Figure 2.12 where d = -1; the positive initial velocity is $v \approx 2.26$. With a period $T \approx 5.42$, the sticking phase occurs until $\tau \approx 1.58$ and is then followed by a free flight of duration $s \approx 3.84$.

Sticking phase of infinite duration

The initial data corresponding to the periodic solution with the largest u_1 in magnitude is $[\mathbf{u}(0), \dot{\mathbf{u}}(0)]^{\top} = [d, d, 0, 0]^{\top}$. The first mass then follows the oscillation around its new equilibrium



FIGURE 2.12 – 1-SPP with finite sticking phase for d = -1 and v > 0, $m_1 = 1.0$; $m_2 = 6.0$; $k_1 = 1.0$; $k_2 = 4.0$ and initial data U(0) = [-1, -1, 2.2686626, 0]: (a) Orbits. (b) Displacements

 \overline{u}_1 . Moreover, 0 is the minimum point of u_1 , thus $u_1(t) \ge d$ for all *t*. By Theorem 2.17, it follows that the sticking phase never ends. This argument is illustrated in Figure 2.13.



FIGURE 2.13 – Sticking phase of infinite duration for d = -1 < 0 and v = 0, $m_1 = 10$; $m_2 = 6$; $k_1 = 2$; $k_2 = 3$, initial data U(0) = [-1, -1, 0, 0]. (a) Orbits. (b) Displacements

2.7.2 Statically grazing system

The dynamics is explored with d = 0.

Finite duration sticking phase

From Proposition 2.16, the sticking phase of finite duration exists if the initial data satisfies $u_2(0) = 0$, $\dot{u}_2(0) = 0$, $u_1(0) = 0$, and $\dot{u}_1(0) = v > 0$. The set of free flight time *s* is found from Equation (2.72) where d = 0, *i.e.* $vw_1(s) = 0$ and $vw_2(s) = 0$. Hence, *v* is arbitrarily positive

and s satisfies $h(s) = w_1(s) - w_2(s) = 0$ and $w_2(s) = 0$ or

$$\begin{bmatrix} \alpha_1 & \alpha_2 \\ \beta_1 & \beta_2 \end{bmatrix} \begin{bmatrix} \varphi(\omega_1 s/2) \\ \varphi(\omega_2 s/2) \end{bmatrix} = \begin{bmatrix} 0 \\ 0 \end{bmatrix}.$$
(2.87)

Through Assumption 2.7, this linear system has the unique solution

$$\varphi(\omega_1 s/2) = 0, \tag{2.88}$$

$$\varphi(\omega_2 s/2) = 0, \tag{2.89}$$

where $\varphi(t) = \cot(t/2)$. It follows that

$$\frac{\omega_1}{\omega_2} = \frac{2k+1}{2l+1} \quad \text{with} \quad k, l \in \mathbb{Z},$$
(2.90)

condition which loosely speaking represents half of the rationals. It should be satisfied to observe a sticking phase of finite duration when d = 0 while the initial velocity of mass m_1 can be chosen arbitrarily positive. Such a 1-SPP when $\omega_1/\omega_2 = 1/5$ is shown in Figure 2.14.



FIGURE 2.14 – 1-SPP with finite sticking phase for d = 0 and v = 10 (any arbitrary positive number is acceptable). (a) Orbits. (b) Displacements.

Infinite duration sticking phase

The unique corresponding initial data in this case is $[\mathbf{u}(0), \dot{\mathbf{u}}(0)]^{\top} = [0, 0, 0, 0]^{\top}$. The equilibrium $\mathbf{u} \equiv \mathbf{0}$ is a solution in which mass m_2 always grazes with the wall since the two masses stay at their equilibrium position when d = 0 and $\dot{u}_1(0) = 0$.

It should thus be noted that generically, there is no 1-SPP except the equilibrium when d = 0.

2.8 A general two-degree-of-freedom vibro-impact system

Consider a general two-dof vibro-impact system with a diagonal mass matrix \mathbf{M} and a stiffness matrix

$$\mathbf{K} = \begin{bmatrix} k_1 & k_2 \\ k_2 & k_3 \end{bmatrix}$$

with $k_1, k_3 > 0$ and $0 < |k_2| \le \sqrt{k_1 k_3}$. The 1-SPP is found in a similar way, except the condition is changed due to the change of **K**. The theorem 2.17 stating the occurrence of the sticking phase becomes:

Theorem 2.17 — Sticking contact. There exists a sticking phase exactly starting at time t_0 and persisting on its right neighbourhood if and only if:

- 1. $u_2(t_0) = d$, $\dot{u}_2(t_0) = 0$, $-k_2u_1(t_0) > k_3d$, or
- 2. $u_2(t_0) = d$, $\dot{u}_2(t_0) = 0$, $-k_2u_1(t_0) = k_3d$, $\dot{u}_1(t_0) > 0$.

The second case where $-k_2u_1(t_0) = k_3d$ and $\dot{u}_1(t_0) > 0$ corresponds to the beginning of the sticking phase.

All the conditions are kept except the one on mass 1, this is due to the second equation

$$m_2 \ddot{u}_2 = -(k_2 u_1 + k_3 u_2) + R(t). \tag{2.91}$$

The sticking phase only occurs when $u_2 = d$, $\dot{u}_2 = 0$, and $-(k_2u_1 + k_3u_2) \ge 0$ or $-k_2u_1 \ge k_3u_2$. Hence, the occurrence of sticking phase happens if $u_2 = d$, $\dot{u}_2 = 0$ and

$$\{-k_2u_1 > k_3u_2\} \text{ or } \{-k_2u_1 = k_3u_2 \text{ and } \dot{u}_1 > 0\}.$$
(2.92)

Based on the sticking equation

$$m_1\ddot{u}_1 + k_1u_1 = -k_2d, (2.93)$$

it follows that the new equilibrium of the first mass is $\bar{u}_1 = -\frac{k_2}{k_1}d$. Hence, the solution with infinite duration of sticking phase may occur only if \bar{u}_1 satisfies the condition in Theorem 2.17, i.e.

$$-k_2 \bar{u}_1 > k_3 d, \tag{2.94}$$

when d > 0, this is equivalent to $k_2^2 > k_1 k_3$, which contradicts the positive definiteness of **K**. Hence, the 1-SPP with a sticking phase of an infinite duration is expected to exist only in the prestressed case, i.e. d < 0. **Remark 2.18.** Function h(s) in theorem 2.4 becomes $h(s) = w_1(s) + \frac{k_3}{k_2}w_2(s)$ and the initial velocity of the first mass is $v = \frac{d}{w_2(s)}$.

2.9 On 1-SPP of *N*-degree-of-freedom vibro-impact systems with *N* > 2

A general N-degree-of-freedom vibro-impact system is considered,

$$\mathbf{M}\ddot{\mathbf{u}} + \mathbf{K}\mathbf{u} = \mathbf{r} \tag{2.95a}$$

$$\mathbf{u}(0) = \mathbf{u}_0, \quad \dot{\mathbf{u}}(0) = \dot{\mathbf{u}}_0 \tag{2.95b}$$

$$u_N(t) \le d, \quad R(t) \le 0, \quad (u_N(t) - d)R(t) = 0, \quad \forall t$$
 (2.95c)

$$\dot{\mathbf{u}}(t)^{\mathsf{T}}\mathbf{M}\dot{\mathbf{u}}(t) + \mathbf{u}^{\mathsf{T}}(t)\mathbf{K}\mathbf{u}(t) = E(\mathbf{u}(t), \dot{\mathbf{u}}(t)) = E(\mathbf{u}(0), \dot{\mathbf{u}}(0)),$$
(2.95d)

where $\mathbf{M} = \text{diag}(m_j)_{j=1}^N$ and $\mathbf{K} = (k_{ij})_{i,j=1}^N$ are the mass and stiffness matrices of the system, $\mathbf{u}(t) = (u_j(t))_{j=1}^N$ is the displacement of the masses, $\mathbf{r}(t) = (0, \dots, 0, R(t))^\top$, with R(t) is the reaction of the obstacle. It is always assumed that there is one degree-of-freedom in contact, which is the N^{th} mass. Similarly, the condition to have a sticking phase is then studied.

2.9.1 The occurrence of the sticking phase

Assume that $\mathbf{u}(t)$ is a solution of (2.95). A general definition of a sticking phase is given.

Definition 2.19 — Sticking phase and its duration. Let **u** be the solution to System (2.95). A *sticking phase* arises if there exist $t_0 \in \mathbb{R}$ and $\tau > 0$ such that

$$u_N(t) = d, \quad \forall t \in [t_0; t_0 + \tau].$$
 (2.96)

Moreover, when there exists $0 < \delta < 1$ such that $\forall t \in [0; \delta[$

$$u_N(t_0 - t) < d$$
 and $u_N(t_0 + \tau + t) < d$, (2.97)

then t_0 is the starting time and τ is the duration of the sticking phase.

The necessary and sufficient conditions to have sticking phase is then studied similarly as in the 2-dof system.

Remark 2.20 — *N*-dof sticking contact. There exists a sticking phase starting exactly at time $t_0 = 0$ and persisting on its right neighbourhood if and only if:

1. $u_N(t_0) = d$, $\dot{u}_N(t_0) = 0$, $F(t_0) < 0$ or 2. $u_N(t_0) = d$, $\dot{u}_N(t_0) = 0$, $F(t_0) = 0$, $F(t_0 + \delta) < 0$ for some $\delta > 0$. where $F(t) = \mathbf{e}_N^{\mathsf{T}} \mathbf{K} \mathbf{u}(t)$.

The is remark can be proven in a similar way as the proof of Theorem 2.17. The sticking system is a non-homogeneous (N - 1)-dof system

$$\underline{\mathbf{M}}\ddot{\mathbf{u}} + \underline{\mathbf{K}}\mathbf{u} = \underline{\mathbf{C}},\tag{2.98}$$

where $\underline{\mathbf{M}}$ and $\underline{\mathbf{K}}$ are $(N - 1) \times (N - 1)$ matrices obtained by neglecting the last row and last column of \mathbf{M} and \mathbf{K} , respectively; $\underline{\mathbf{C}} = d \mathbf{l}_N^{\mathsf{T}}$ with $\mathbf{l}_N = [k_{Nj}]_{j=1}^{N-1}$. It will be proven in Section 3.3 of Chapter 3, that when *d* is positive, the sticking phase ends after a finite time. Base on the occurrence of the sticking phase in Proposition 2.20, the sticking duration τ is then implicitly found from:

$$F(\tau) = 0, \ F(\tau + \delta) > 0, \ \text{for some } \delta \in]0; \ \delta_0[, \delta_0 > 0.$$

$$(2.99)$$

The feature of the 2-dof system is the 1-dof sticking system which exhibits the symmetry of the solution during the sticking phase. This is the main reason why the author have not succeeded to obtain the 1-SPP of the *N*-dof system.

2.9.2 A 3-dof chain with a non prestressed gap d > 0

Consider a chain of three masses in which the third mass is constrained by a wall at a distance d from its equilibrium. The system of interest is (2.1) where

$$\mathbf{M} = \begin{bmatrix} m_1 & 0 & 0 \\ 0 & m_2 & 0 \\ 0 & 0 & m_3 \end{bmatrix}; \ \mathbf{K} = \begin{bmatrix} k_1 + k_2 & -k_2 & 0 \\ -k_2 & k_2 + k_3 & -k_3 \\ 0 & -k_3 & k_3 \end{bmatrix}; \ \mathbf{u}(t) = \begin{pmatrix} u_1(t) \\ u_2(t) \\ u_3(t) \end{pmatrix}; \ \mathbf{r}(t) = \begin{pmatrix} 0 \\ 0 \\ R(t) \end{pmatrix}.$$
(2.100)

Let us explain in details the difficulty when looking for 1-SPP.

The occurrence of the sticking phase

There are following possibilities for a sticking phase to occurs:

• $u_3(0) = d$, $\dot{u}_3(0) = 0$, $u_2(0) > d$, or

- $u_3(0) = d$, $\dot{u}_3(0) = 0$, $u_2(0) = d$, $\dot{u}_2(0) > 0$, or
- $u_3(0) = d$, $\dot{u}_3(0) = 0$, $u_2(0) = d$, $\dot{u}_2(0) = 0$, $u_1(0) > d$, or
- $u_3(0) = d$, $\dot{u}_3(0) = 0$, $u_2(0) = d$, $\dot{u}_2(0) = 0$, $u_1(0) = d$, $\dot{u}_1(0) > 0$.

The end of sticking phase at the time τ is replaced by one of the following conditions

- $u_3(\tau) = d$, $\dot{u}_3(0) = 0$, $u_2(\tau) = d$, $\dot{u}_2(\tau) < 0$, or
- $u_3(\tau) = d$, $\dot{u}_3(\tau) = 0$, $u_2(\tau) = d$, $\dot{u}_2(\tau) = 0$, $u_1(\tau) < d$, or
- $u_3(\tau) = d$, $\dot{u}_3(\tau) = 0$, $u_2(\tau) = d$, $\dot{u}_2(\tau) = 0$, $u_1(\tau) = d$, $\dot{u}_1(\tau) < 0$.

The existence of solution with sticking phase

Sticking phase in general Assume that there exists a sticking phase at t_0 . The sticking system is then a non-homogeneous 2-dof system with the new equilibrium

$$[u_{1e}, u_{2e}] = \left[\frac{k_2 k_3}{k_1 k_2 + k_2 k_3 + k_3 k_1} d, \frac{k_1 k_3 + k_2 k_3}{k_1 k_2 + k_2 k_3 + k_3 k_1} d\right].$$
(2.101)

Hence, for d > 0, the duration of sticking phase is always finite since $u_{2e} < d$.

For d < 0, it follows that $u_{2e} > d$, hence with an energy small enough, $u_2(t)$ can be always greater than d, which follows the existence of the sticking phase with infinite duration.

Periodic solution with 1 sticking phase per period Assume that the initial state of the system is $\mathbf{U}(0) = [\mathbf{u}_0, \dot{\mathbf{u}}_0]^\top = [u_{10}, d, d, \dot{u}_{10}, v_{20}, 0]^\top$ with 3 parameters $u_{10}, \dot{u}_{10}, v_{20}$.

Assume that the sticking phase ends when $u_2(\tau) = d$ where $u_2(t)$ is the solution of the 2-dof sticking system. When τ is found, we have the data for the system at τ which is the initial data for the free flight phase after the sticking.

Then solving the system during free flight from τ to T with the condition to have a periodic solution $\mathbf{U}(T) = \mathbf{U}(0)$. It follows 5 equations of 3 parameters to solve. It seems to be over-determined.

2.10 Conclusion

The free dynamics of a two-degree-of-freedom linear oscillator subject to a unilateral constraint on one mass is investigated. Generically, a Newton-like impact law has to be incorporated in this type of formulation. In this work, periodic orbits with one sticking phase per period (1-SPP) are considered: it is shown that they are independent of the impact law. They might not always exist and whenever they exist, they are isolated as opposed to one-impact-per-period solutions (1-IPP) known to be organized on manifolds [25]. Also, they cannot be obtained through usual perturbation methods.

The whole set of 1-SPP is characterized by only one parameter belonging to a discrete set: the free flight duration. This parameter belongs to a countable set which can be empty, or even infinite in some circumstances.

A systematic numerical procedure designed to find all possible 1-SPP is expounded. It involves two numerical steps:

- 1. finding the roots of an explicit quasi-periodic function, and
- 2. checking that the corresponding closed-form trajectory satisfies the unilateral condition on the whole period of motion.

Many examples are presented but the existence proof of 1-SPP remains an open problem. In addition, parameter values leading to the non-existence of 1-SPP are provided. However, under generic assumptions on the mass and stiffness matrices, a countable infinite set of initial data including all the initial data of 1-SPP can be exhibited. The closed-forms emanating from this set (of initial data) satisfy the unilateral constraint at least near the sticking phase. The prestressed structure is also explored. The picture is similar except that 1-SPP with infinite sticking time are also found.

Extension to *N* degrees-of-freedom with N > 2 is not straightforward: the symmetry $\mathbf{u}(t) = \mathbf{u}(-t)$, property heavily used in this work, is unknown and the sticking dynamics is more complicated, the sticking system has dimension N - 1 > 1.

Chapter 3

On the first return time of *N***-degree-of-freedom vibro-impact oscillators**

Summary For an *N*-degree-of-freedom linear vibro-impact system with a unilateral contact it is known in structural mechanics that orbits near grazing contacts lead to square-root instability [7, 33]. In this chapter, the first return map is revisited in a rigorous mathematical framework to properly describe this singularity. The Poincaré section is chosen at the contact to understand the dynamics but with a loss of the usual transversality condition. The domain of the first return map is precisely explored. The square-root singularity and new singular behaviors are highlighted. In particular, the discontinuity of the first return time is the worst singular behavior of some grazing contacts. Finally, a general condition is stated to check the square-root dynamics near the linear grazing orbits which may lead to their instability.

3.1 Introduction

Recent results on nonlinear modes for vibro-impact discrete structural systems suggest that the first return map (FRM) should be defined on the hyperplane of the unilateral constraint in the phase-space [7, 33]. This approach distinguishes itself from the "discontinuity mapping" [7, 33] where the Poincaré section lies outside the contact interface. In the context of serial mass-spring systems, simplified versions of the FRM already exist; they give access to special periodic solutions with closed-form expression [25] and to reduced-order systems of nonlinear equations [45, 28] together with the companion stability analysis. In this work the FRM, which is the fundamental tool to explore periodic solutions, is extended to the investigation of other potential solutions in order to improve the understanding of the global dynamics. Since the Poincaré section is a subset of the contact interface in the phase-space, the natural transversality

condition is lost. This means that the Poincaré section can be tangent to orbits which yields the well-known square-root singularity [7, 33]. This singularity is here revisited in a rigorous mathematical framework.

The system of interest is a second-order system

$$\mathbf{M}\mathbf{\ddot{u}} + \mathbf{K}\mathbf{u} = \mathbf{r} \tag{3.1a}$$

$$\mathbf{u}(0) = \mathbf{u}_0, \quad \dot{\mathbf{u}}(0) = \dot{\mathbf{u}}_0 \tag{3.1b}$$

$$u_N(t) \le d, \quad R(t) \le 0, \quad (u_N(t) - d) R(t) = 0$$
 (3.1c)

$$\mathbf{\dot{u}}^{+}(t)^{\mathsf{T}}\mathbf{M}\mathbf{\dot{u}}^{+}(t) + \mathbf{u}^{\mathsf{T}}(t)\mathbf{K}\mathbf{u}(t) = E(\mathbf{u}(t), \mathbf{\dot{u}}^{+}(t)) = E(\mathbf{u}(0), \mathbf{\dot{u}}(0)),$$
(3.1d)

with

$$\mathbf{M} = \mathbf{diag}(m_j)_{j=1}^N; \qquad \mathbf{K} = (k_{ij})_{i,j=1}^N; \qquad \mathbf{u}(t) = (u_j)_{j=1}^N; \qquad \mathbf{r}(t) = (0, \dots, 0, R(t)) \quad (3.2)$$

where \dot{u}_j and \ddot{u}_j represent the velocity and acceleration of the mass j, j = 1, ..., N, respectively. Condition (3.1c) says that mass N is constrained on the right side by a rigid obstacle at a distance d > 0 from its equilibrium. There is only one constraint on the mass N. The other masses are not constrained. The quantity R(t) is the reaction force induced by the obstacle on mass N at the time of gap closure. In general R(t) is a measure, but for solution with sticking phases, it is a Lipschitz function which is as regular as \ddot{u}_N [28].

System (3.1) without (3.1d) is not well posed: it is known that uniqueness might not be ensured for the initial value problem. An impact law is usually incorporated into the formulation. Here, since we are interested in the non-dissipative dynamics, condition (3.1d) is enforced: the total energy is preserved during the motion. This implies the existence of a perfectly elastic impact law of the form $\dot{u}_N^+ = -e\dot{u}_N^-$ with e = 1 where \dot{u}_N^- and \dot{u}_N^+ respectively stand for the preand post-impact velocities of mass N. For the well-posedness of the initial-value problem with conserved energy, see [5, 40].

Matrices **M** and **K** are symmetric positive definite. Hence, there exists a matrix **P** of **M**-orthogonal eigenmodes which diagonalizes both **M** and **K**, that is $\mathbf{P}^{\mathsf{T}}\mathbf{M}\mathbf{P} = \mathbf{I}$ and $\mathbf{P}^{\mathsf{T}}\mathbf{K}\mathbf{P} = \mathbf{\Omega}^2 = \mathbf{diag}(\omega_j^2)|_{j=1,\dots,N}$ where **I** is the $N \times N$ identity matrix; ω_j^2 (resp., T_j) are the eigenfrequencies (resp., the associated linear periods) where $\omega_j T_j = 2\pi$, $j = 1, \dots, N$.

The chapter is organized as follows: in Section 3.2, the main results are stated including the choice of a suitable Poincaré section. Are shown also the square-root term in the first return time near the first return time of a grazing orbit and the square-root dynamics near a linear grazing orbit. The following sections detail the above results. Section 3.3 deals with the exact subset of initial data at the contact constraint $u_N = d$ where the trajectory will come back to the constraint, that is the domain of the FRM. The square-root singularity is focused on in Section 3.4 in a mathematical framework only where the implicit function theorem is used in a degenerate case.

In particular, it implies that the first return time is discontinuous. The dynamics induced by the square-root singularity is studied in Section 3.5. Some coefficients of the asymptotic expansion of the FRM activating the square-root instability are identified. The stability of grazing periodic solutions is discussed in Section 3.6.

3.2 Main results

The Poincaré section is first defined in Section 3.2.1. The theorems on the square-root singularity that comes from the grazing contact are in Section 3.2.2. The last section provides a result on the dynamics near a periodic solution with one grazing contact occurrence.

3.2.1 Poincaré section

In smooth analysis, a Poincaré section can always be locally defined transversally to the flow away from fixed-points. This section corresponds to the domain of definition of the corresponding First Return Map. In nonsmooth analysis where the vector field governing the dynamics is piecewise smooth, the hyperplane $\mathcal{H} = \{[\mathbf{u}, \dot{\mathbf{u}}]^\top \in \mathbb{R}^{2N}, u_N = d\}$ of the phase-space defined by $u_N = d$ is a natural choice for the Poincaré section when targeting trajectories with non-vanishing pre-impact velocities [25] since they hit the section transversally. However, for grazing trajectories, the transversality condition is lost. A discontinuity mapping on another suitable section for which the transversality condition is recovered shall be used instead [7, 17, 33]. In this work, the Poincaré section is still a subset of the hyperplane \mathcal{H} . This is the simplest cross section to describe the dynamics with only two phases: contact dynamics and free-flight dynamics. This does not have adverse effect of introducing a second free-flight dynamics as the discontinuity mapping does. Nevertheless, our critical choice necessitates a very careful delimitation of the domain of definition of the FRM. This is stated in the following theorem that categorizes the initial data generating orbits which will always come back to the section.

With the above Newton's impact law, two types of closing contacts can be reported: contacts with non-zero pre-impact velocity (or simply impacts) and contacts with zero pre-impact velocity. The second type can be divided as follows: a grazing contact if the mass leaves the obstacle right after the contact time, or sticking if the mass stays in contact with the obstacle for a finite time interval.

Proposition 3.1 — Finite sticking duration. When d is positive, the sticking phase of a solution to (3.1) is of finite duration.

This is proven in [28] for a 2-dof vibro-impact system. A general proof for the N-dof system is given in Section 3.3.1. This property is used to show that a solution to (3.1) has zero, one

or an infinite number of closing contacts with the hyperplane \mathcal{H} . Before stating this result, the following assumption is needed.

Assumption 3.2 — No internal resonances. The linear periods of system (3.1) satisfy $\bigcap_{i=1}^{N} T_{i}\mathbb{N} = \emptyset$.

By this assumption, the internal resonances are discarded in the current work.

Theorem 3.3 — Zero, one or infinite number of closing contacts on \mathcal{H} . Let $\mathbf{u}(t)$ be a solution to (3.1). Under Assumption 3.2 of no internal resonances, the solution is such that

- **Case 1: linear solution** the N^{th} mass never hits \mathcal{H} , i.e. $u_N(t) < d$ for all t.
- **Case 2: linear solution** the N^{th} mass experiences only one closing contact, i.e. there exists t_0 such that $u_N(t_0) = d$ and $u_N(t) < d$ for all $t \neq t_0$.
- **Case 3: nonlinear solution** the N^{th} mass experiences a countably infinite number of isolated closing contacts on \mathcal{H} .

This theorem is proven in Section 3.3.2. The affine space \mathcal{H} is of dimension 2N - 1. It is divided into three disjoint subsets:

$$\mathcal{H}^{-} = \{ [\mathbf{u}, \dot{\mathbf{u}}^{-}]^{\top} \in \mathbb{R}^{2N}, u_{N} = d \text{ and } \dot{u}_{N}^{-} > 0 \}$$

$$(3.3)$$

$$\mathcal{H}^+ = \{ [\mathbf{u}, \dot{\mathbf{u}}^+]^\top \in \mathbb{R}^{2N}, \, u_N = d \text{ and } \dot{u}_N^+ < 0 \}$$
(3.4)

$$\mathcal{H}^0 = \{ [\mathbf{u}, \dot{\mathbf{u}}]^\top \in \mathbb{R}^{2N}, \, u_N = d \text{ and } \dot{u}_N = 0 \}$$
(3.5)

By conservation of energy, it can be seen that the solution which has a contact with non-zero velocity will always experience a later closing contact. The problem of whether there will be a subsequent closing contact or not emerges only on \mathcal{H}^0 . Theorem 3.3 implies that \mathcal{H}^0 is the union of the two subsets

$$\mathcal{H}^0_{\infty} = \{ [\mathbf{u}, \dot{\mathbf{u}}]^\top \in \mathcal{H}^0, \text{ the associated orbit has an infinite number of contacts} \}$$
(3.6)

$$\mathcal{H}_{1}^{0} = \{ [\mathbf{u}, \dot{\mathbf{u}}]^{\top} \in \mathcal{H}^{0}, \text{ the associated orbit has only one grazing contact} \}$$
(3.7)

Since \mathcal{H}_1^0 includes solutions with only one grazing contact, it is sufficient to discard this set from the definition of the Poincaré section.

Definition 3.4 — **Poincaré section.** The Poincaré section $\mathcal{H}_{\mathcal{P}} \subset \mathcal{H}$ is the union of the set of initial data with non-zero velocity contacts and the set of initial data with zero velocity contact that gives rise to an infinite number of closing contacts:

$$\mathcal{H}_{\mathcal{P}} = \mathcal{H}^- \cup \mathcal{H}^0_{\infty}. \tag{3.8}$$

Remark 3.5. There are two options for the choice of the Poincaré section in \mathcal{H} depending on whether one wants to start right before or right after the contact occurrence. The former gives the Poincaré section defined as above, and the latter gives $\mathcal{H}^+_{\varphi} = \mathcal{H}^+ \cup \mathcal{H}^0_{\infty}$ as the Poincaré section.

The set \mathcal{H}^0_{∞} can also be split into two subsets: $\mathcal{H}^0_{\mathcal{S}}$ including all the initial data belonging to \mathcal{H}^0_{∞} such that the solution starts by a sticking contact and $\mathcal{H}^0_{\mathcal{G}}$ of initial data such that the solution starts by a grazing contact:

$$\mathcal{H}^0_{\infty} = \mathcal{H}^0_{\mathcal{S}} \cup \mathcal{H}^0_{\mathcal{G}}.$$
(3.9)

3.2.2 The square-root singularity

The first return time is known to be generically analytic: let $W_0 \in \mathcal{H}^-$, if the first return to \mathcal{H} , named W_1 , belongs to \mathcal{H}^- then the FRM is analytic near W_0 [7, 25]. Nonetheless, if $W_1 \in \mathcal{H}^0$, then there is a grazing contact and it is known that a square-root singularity appears [17, 33].

By definition of $\mathcal{H}_{\mathcal{P}}$, there exists a time such that the orbit emanating from $W \in \mathcal{H}_{\mathcal{P}}$ comes back to $\mathcal{H}_{\mathcal{P}}$. The first return time is then well defined as follows.

Definition 3.6 — First return time. Let $\mathbf{u}(t)$ be a solution to (3.1) with the initial data $\mathbf{W} = (W_i)_{i=1}^{2N} \in \mathcal{H}_{\mathcal{P}}$ at the initial time t = 0, i.e. $u_N(0) = \mathbf{e}_N^{\mathsf{T}} \mathbf{u}(0) = d$. The first return time $T = T(\mathbf{W}) > 0$ is defined by

$$T(\mathsf{W}) = \begin{cases} \min\{t > 0 : u_N(t) = d\} \text{ if there is no sticking phase at } t = 0 \\ \min\{t > \tau(\mathsf{W}) : u_N(t) = d\} \text{ if there is sticking phase at } t = 0 \end{cases}$$
(3.10a)
(3.10b)

where $\tau(W)$ is the sticking duration.

This definition can be shorten by saying that $T(W) = \min\{t > \tau(W) : u_N(t) = d\}$, with the convention that $\tau(W) = 0$ if there is no sticking at t = 0. For the case when a sticking phase occurs at t = 0, Proposition 3.1 ensures that the first return time is well defined since the duration of the sticking phase is finite.

Let $\mathbf{u}(t, \mathbf{W})$ be the solution associated with the initial data $\mathbf{W} \in \mathcal{H}_{\mathcal{P}}$. By the definition of $\mathcal{H}_{\mathcal{P}}, W_N = d$, hence W is viewed as a vector of 2N - 1 variables. The displacement of the N^{th} mass is $u_N(t, \mathbf{W}) = \mathbf{e}_N^\top \mathbf{u}(t, \mathbf{W})$ which is a function of time variable *t* and the initial data W. Consider the smooth function Φ defined for all $t \in \mathbb{R}$ and for all $\mathbf{W} \in \mathbb{R}^{2N}$:

$$\Phi(t, \mathsf{W}) = \mathbf{e}_N^{\mathsf{T}} \mathbf{R}(t) \mathbf{S} \mathsf{W},\tag{3.11}$$

where S = diag(1, ..., 1, -1) is a $2N \times 2N$ diagonal matrix with last entry -1 to reflect the impact

law. The operator $\mathbf{R}(t)$ describes the free-flight dynamics:

$$\mathbf{R}(t) = \begin{vmatrix} \mathbf{P}\cos(t\Omega)\mathbf{P}^{-1} & \mathbf{P}\Omega^{-1}\sin(t\Omega)\mathbf{P}^{-1} \\ -\mathbf{P}\Omega\sin(t\Omega)\mathbf{P}^{-1} & \mathbf{P}\cos(t\Omega)\mathbf{P}^{-1} \end{vmatrix}.$$
(3.12)

The dynamics between two successive closing contacts is smooth, except at the beginning and the end of the contact occurrences. The function $\Phi(t, W)$ coincides with $u_N(t, W)$ as long as $u_N(t, W) < d$.

Assumption 3.7 — Non-zero acceleration. Let W_0 be the initial data leading to an orbit which has a grazing contact at the first time T_0 . Assume that $\partial_{tt} \Phi(T_0, W_0) \neq 0$.

This is an important assumption to activate the square-root singularity near a grazing contact. With this assumption and the assumption that for some $1 \le k \le 2N$, $\partial_{W_k} \Phi(T_0, W_0) \ne 0$, the following scalar is well defined and non-zero:

$$\gamma_k = -\frac{\partial_{tt} \Phi(T_0, \mathsf{W}_0)}{2 \,\partial_{W_k} \Phi(T_0, \mathsf{W}_0)}.\tag{3.13}$$

The expression of the first return time near a grazing contact is then given in the following theorem. The implicit function theorem is applied on the smooth function Φ in place of the nonsmooth function u_N . It is a crucial ingredient of the proof.

Theorem 3.8 — Square-root singularity near a grazing contact. Let $W_0 = (W_{0i})_{i=1}^{2N} \in \mathcal{H}^$ be the initial data generating an orbit which has a grazing contact at the first return time $T_0 = T(W_0)$. Let W_k be a component of W such that $\partial_{W_k} \Phi(T_0, W_0) \neq 0$. Such a component always exists. Let $\underline{W} \in \mathbb{R}^{2N-1}$ be the reduced vector obtained from W by removing W_k . Moreover, if Assumptions 3.2 and 3.7 hold then:

1. There exist two neighborhoods V_{T_0} and $V_{W_0} \subset \mathcal{H}_{\mathcal{P}}$ of T_0 and W_0 , respectively as well as two smooth scalar functions η and α defined on $V_{W_0} = \{\underline{W}, W \in V_{W_0}\}$ such that the set

$$S_c = \{(t, \mathsf{W}) \in \mathbb{R} \times \mathbb{R}^{2N}, \ \Phi(t, \mathsf{W}) = d \text{ and } \partial_t \Phi(t, \mathsf{W}) = 0\},\tag{3.14}$$

where the square-root singularity will emerge, is locally parameterized as follows:

$$S_c \cap \{V_{T_0} \times V_{\mathsf{W}_0}\} = \{(\eta(\underline{\mathsf{W}}), \alpha(\underline{\mathsf{W}}), \underline{\mathsf{W}}), \underline{\mathsf{W}} \in V_{\mathsf{W}_0}\},$$
(3.15)

where $\eta(\underline{W}_0) = T_0$ and $\alpha(\underline{W}_0) = W_{0k}$.

2. Let $s_k = \operatorname{sign}(\gamma_k)$ and the set

$$\mathcal{B}_k = \{ \mathsf{W} \in V_{\mathsf{W}_0}, \, s_k(W_k - \alpha(\underline{\mathsf{W}})) \ge 0 \}.$$
(3.16)

Let $\sigma = \operatorname{sign}(\ddot{u}_N(T_0, W_0))$. There exists a smooth function ψ such that $\psi(0, \underline{W}_0) = 0$, $\partial_1 \psi(0, \underline{W}_0) = 1/\sqrt{|\gamma_k|}$ and for all $W \in \mathcal{B}_k$, the first return time *T* is given by

$$T(\mathsf{W}) = \eta(\underline{\mathsf{W}}) + \psi(\sigma\sqrt{s_k(W_k - \alpha(\underline{\mathsf{W}}))}, \underline{\mathsf{W}}).$$
(3.17)

Moreover, the set $\{W \in \mathcal{H}_{\mathcal{P}}, (T(W), W) \in V_{T_0} \times V_{W_0}\}$ is exactly \mathcal{B}_k .

A mathematical proof of this theorem will be given in Section 3.6.1. Take note that the square-root dependence on the initial data only appears on the set \mathcal{B}_k which is simply the region above or below the hypersurface $W_k = \alpha(\underline{W})$. The square-root term acts in one direction. This is reported by Nordmark and Fredriksson [17] and also in references [7, 33, 47].

The presence of the square-root singularity is the consequence of the non-vanishing acceleration \ddot{u}_N at T_0 . While studying the other contacts with zero-velocity, if, instead, $\dot{u}_N(T_0) = \ddot{u}_N(T_0) = 0$ and $\ddot{u}_N(T_0) \neq 0$ then a cube-root singularity is expected. More generally, if $u_N^{(l)-}(T_0) = 0$ for all 0 < l < n and $u_N^{(n)-}(T_0) \neq 0$, then a *n*th-root singularity is expected, where $u_N^{(l)-}$ refers to the left time derivative of order l^{th} of u_N .

Orbits involving grazing contacts are expected to be less frequent than the ones with non-zero velocity contacts. Studying the grazing orbits is however important since the dynamics near such a grazing contact is diversified and many criteria can be exhibited as explained further in the next section.

Remark 3.9 — **Discontinuous first return time (FRT).** Since the set \mathcal{B}_k is a one-sided set with respect to an hypersurface, the set of initial data with their first return time near T_0 is not a neighborhood of W₀. This means that, in the vicinity of W₀, there exist a lot of initial data such that the first return time is far from T_0 . In other words, the first return time T(W) is discontinuous at W₀, see Section 3.4.3.

Theorem 3.8 also has to be generalized for an initial data $W_0 \in \mathcal{H}^0_{\mathcal{G}}$ for instance the initial data of a linear grazing orbit (LGO). The neighborhood V_{W_0} has to be replaced by the one half neighborhood $V_{W_0}^+ = \{W \in V_{W_0}, \mathbf{e}_{2N}^\top W \ge 0\}$ where the initial velocity is non-positive and the set \mathcal{B}_k is replaced by the smaller set

$$\mathcal{B}_k^+ = \mathcal{B}_k \cap V_{\mathsf{W}_0}^+ = \{\mathsf{W} \in V_{\mathsf{W}_0}, \, \gamma_k(W_k - \alpha(\underline{\mathsf{W}})) \ge 0, \, W_{2N} \ge 0\}.$$
(3.18)

The following theorem is stated with the notations of Theorem 3.8.

Theorem 3.10 — Square-root singularity for two successive grazing contacts. Assume that $W_0 \in \mathcal{H}^0_{\mathcal{G}}, \ddot{u}^+_N(0, W_0) < 0$ and $W(T_0) \in \mathcal{H}^0$ where $T_0 = T(W_0)$. Let *k* belong to $\{1, \ldots, 2N\}$ such that $\partial_{W_k} \Phi(T_0, W_0) \neq 0$ and $s_k = \operatorname{sign}(\gamma_k)$ together with $\sigma = \operatorname{sign}(\ddot{u}^-_N(T_0, W_0))$.

If Assumptions 3.2 and 3.7 hold, then there exist two neighborhoods V_{T_0} and $V_{W_0} \subset \mathcal{H}_{\mathcal{P}}$ of T_0 and W_0 , respectively as well as two smooth scalar functions η and α defined on $V_{\underline{W}_0}$ containing \underline{W}_0 where $\eta(\underline{W}_0) = T_0$ and $\alpha(\underline{W}_0) = W_{0k}$. There also exists a smooth function ψ such that $\psi(0, \underline{W}_0) = 0$, $\partial_1 \psi(0, \underline{W}_0) = |\gamma_k|^{-1/2}$ and for all $W \in \mathcal{B}_k^+$, the first return time *T* is given by

$$T(\mathsf{W}) = \eta(\underline{\mathsf{W}}) + \psi(\sigma\sqrt{s_k(W_k - \alpha(\underline{\mathsf{W}})), \underline{\mathsf{W}}}).$$
(3.19)

Moreover, the set $\{W \in \mathcal{H}_{\mathcal{P}}, (T(W), W) \in V_{T_0} \times V_{W_0}^+\}$ is exactly \mathcal{B}_k^+ .

The difference with Theorem 3.8 is the smaller validity of the square-root fomula for the first return time. This is due to the fact that $\mathcal{H}_{\mathcal{G}}^0$ is at the boundary of the cross section $\mathcal{H}_{\mathcal{P}}$ so not all perturbations of W₀ are admissible. The condition $\ddot{u}_N^+(0, W_0) < 0$ insures that solutions with initial data near W₀ and with a non-negative velocity for the last mass have a first return time near T_0 . The case W₀ $\in \mathcal{H}_{\mathcal{S}}^0$ can also be considered and can add another singularity due to the sticking phase.

3.2.3 Dynamics near a linear grazing orbit

The square-root dynamics near the periodic solutions with one grazing contact per period is addressed. For an *N*-dof system without internal resonances, there are *N* such periodic solutions which are called linear grazing orbits. It is known [25, 46] that linear grazing orbits are the source of many branches of periodic solutions with *k* Impact-Per-Period (*k*-IPP).

Recall that we define the j^{th} linear grazing orbit as a periodic trajectory **u** associated to the j^{th} linear mode which satisfies $\max_{t \in \mathbb{R}} u_N(t) = d$, i.e. the contacts are at most of grazing type. A essential tool to study the dynamics near a linear grazing orbit is the first return map [7]. This map is well defined on $\mathcal{H}_{\mathcal{P}}$.

Definition 3.11 — First return map. Suppose $W \in \mathcal{H}_{\mathcal{P}}$ and T = T(W) > 0 is the first return time to $\mathcal{H}_{\mathcal{P}}$ of the orbit emanating from W. The map which associates points in $\mathcal{H}_{\mathcal{P}}$ to their first return images to $\mathcal{H}_{\mathcal{P}}$ is called the first return map \mathcal{F} . To be more precise, $\mathcal{F} : \mathcal{H}_{\mathcal{P}} \to \mathcal{H}_{\mathcal{P}}$,

$$\mathcal{F}(\mathsf{W}) = \int \mathbf{R}(T(\mathsf{W})) \, \mathbf{S} \, \mathsf{W} \qquad \text{if } \mathsf{W} \in \mathcal{H}^- \cup \mathcal{H}^0_{\mathcal{G}}, \qquad (3.20a)$$

$$\mathcal{F}(\mathsf{W}) = \begin{cases} \mathbf{R}(s(\mathbf{U}(\tau(\mathsf{W}))) \mathbf{U}(\tau(\mathsf{W})) & \text{if } \mathsf{W} \in \mathcal{H}^0_{\mathcal{S}}, \end{cases}$$
(3.20b)

where the matrix **S** describes the impact law; $\mathbf{U}(t) = [\mathbf{u}(t), \dot{\mathbf{u}}(t)]$ is the state of the system at the time *t*; τ and *s* are the duration of the sticking phase and of the free-flight phase, respectively.

Let us investigate formula (3.20b). If there is a sticking phase, i.e. $W \in \mathcal{H}_{S}^{0}$, then $\tau = \tau(W)$ is the sticking duration and $s = T(W) - \tau(W)$ denotes the duration of the free-flight after the sticking phase until the next contact. The state of the system at the end of the sticking phase is called $U(\tau(W))$. If $W \in \mathcal{H}^- \cup \mathcal{H}_{G}^{0}$, i.e. there is no sticking at t = 0, then τ is assumed to be 0, hence, $s \equiv T$, and $U(\tau(W)) \equiv W$. In other words, formula (3.20b) is valid for all the cases. An explicit formula of τ for a two-degree-of-freedom system is exposed in [28].

The square-root singularity in formula (3.17) of the first return time in the vicinity of a grazing contact is an element showing that the periodic orbits including grazing contacts yield complex dynamics [17, 33]. This square-root term may produce the so-called square-root dynamics via the coefficients C_k as defined below. In the particular framework of Section 3.5, the instability of the linear grazing orbit is expected.

Definition 3.12 — Square-root dynamics coefficients. Suppose that $W_0 \in \mathcal{H}_{\mathcal{P}}$ generates an orbit with the first contact at T_0 being a grazing contact, i.e. $u_N(T_0) = d$ and $\dot{u}_N^-(T_0) = 0$. Under Assumption 3.7, for each $k \in \{1, ..., 2N\}$ satisfying

$$\partial_{W_k} \Phi(T_0, \mathsf{W}_0) \neq 0, \tag{3.21}$$

the square-root dynamics coefficient C_k is defined by

$$C_{k} = \begin{cases} \sqrt{|\gamma_{k}|} \mathbf{e}_{k}^{\top} \mathbf{P} \dot{\mathbf{q}}(T_{0}) & \text{if } 1 \le k < N, \end{cases}$$
(3.22a)

$$\left(-\sqrt{|\gamma_k|} \mathbf{e}_{k-N}^{\mathsf{T}} \mathbf{M} \mathbf{P} \ \mathbf{\Omega}^2 \mathbf{q}(T_0) \quad \text{if } N < k \le 2N, \right.$$
(3.22b)

where $[\mathbf{q}, \dot{\mathbf{q}}]$ are modal coordinates following from $\mathbf{u} = \mathbf{P}\mathbf{q}$.

The square-root dynamics is then defined as follows.

Definition 3.13 — Square-root dynamics. System (3.1) is said to feature square-root dynamics near the grazing periodic solution associated to the initial condition W_0 if there exists at least a coefficient C_k , $1 \le k \le 2N$, $k \ne N$ which does not vanish.

Near a linear grazing orbit, the square-root dynamics is shown to exist under a generic condition as stated in the following theorem. This theorem is proven in Section 3.6.2.

Theorem 3.14 — Square-root dynamics near linear grazing orbits. Consider the j^{th} linear grazing orbit of (3.1) associated to the initial state $W_0 \in \mathcal{H}_{\mathcal{P}}$ and T_j , its period. Assumption 3.2 and 3.7 hold. If there exists an index $N < i \leq 2N$ such that $\partial_{W_i} u_N(T_j, W_0) \neq 0$ and $P_{ij} \neq 0$, then there exists a square-root dynamics near the j^{th} linear grazing orbit.

If, instead, $C_k = 0$ for all k then the square-root dynamics is not activated and the dynamics near the periodic orbit is similar to that of smooth dynamics.

This theorem gives a hint on how to study the instability of the linear grazing orbits. For instance, if after every closing contact, the orbit stays in the same regime where the square-root singularity is preserved, then the dynamics follows the framework stated in Section 3.5 where the instability of the associated fixed-point of the first return map is elaborated.

The condition $P_{ij} \neq 0$ comes from the formula of C_i , where $N < i \le 2N$. Despite the fact that $P_{ij} \neq 0$ is a generic property, it may be violated for a chain of masses. Hence, in order to

have such a $C_i \neq 0$, we need to give a condition on P_{ij} . When N = 2 (see [28]), it will be shown that $P_{ij} \neq 0$ for all i, j = 1, 2, therefore, there is always a square-root dynamics near the linear grazing orbits.

3.3 Domain of definition of the first return map

This section details a comprehensive investigation of the Poincaré section.

3.3.1 Contact times

The contact times are defined and categorized. In particular, if a sticking phase starts, then it is of finite duration. Moreover, the total energy of the system is conserved during the sticking phase. These features will be used to show that there is a countably infinite number of closing contacts after a sticking phase, further details are found in Section 3.3.2.

Definition 3.15 — Contact time. Assume that $\mathbf{u}(t)$ is the solution to system (3.1); *T* is a contact time if $u_N(T) = d$ and there exists $\delta > 0$ such that $u_N(T - t) < d$ for all $0 < t < \delta$.

In other words, a contact time T is the time when the mass N touches the rigid obstacle after a free-flight phase. Invoking [5], the contact time is well defined in the conservative case which is without chattering.

Contact times are classified into three categories:

- 1. It is a contact with non-zero pre-velocity if $u_N(T) = d$ and $\dot{u}_N(T) > 0$.
- 2. It is a contact with zero pre-velocity if $u_N(T) = d$ and $\dot{u}_N(T) = 0$, with two new possibilities:
 - (a) a grazing contact if the mass leaves the obstacle right after the contact time; or
 - (b) a sticking contact if the mass stays in contact with the obstacle.

In this chapter, the term "closing contact" indifferently refers to either an impulsive impact or a grazing contact or the beginning of a sticking contact phase. The sticking system [7] dictates the dynamics during the sticking phase. Since the last mass is at rest, the system "loses" one degree-of-freedom. The sticking system complemented by the initial data at the beginning of a sticking phase is explicitly derived as

$$\underline{\mathbf{M}}\underline{\ddot{\mathbf{u}}} + \underline{\mathbf{K}}\underline{\mathbf{u}} = -d\,\underline{\mathbf{l}}_{N}^{\mathsf{T}} \tag{3.23}$$

$$m_N \ddot{u}_N = 0, \quad u_N(0) = d, \quad \dot{u}_N(0) = 0.$$
 (3.24)

where $\underline{\mathbf{M}}$, $\underline{\mathbf{K}}$ are the mass and stiffness matrix after removing the last row and last column, $\underline{\mathbf{u}}$ is a N - 1 vector solution to the sticking dynamics, $\underline{\mathbf{l}}_N^{\mathsf{T}}$ is the last column of \mathbf{K} where the last entry k_{NN} has been removed.

Proposition 3.1 can now be proven.

Proof. Assume that the sticking phase never ends. The new equilibrium $\underline{\mathbf{u}}_{e}$ of the sticking system satisfies

$$\underline{\mathbf{K}}\underline{\mathbf{u}}_{e} = -d\,\underline{\mathbf{I}}_{N}^{\mathsf{T}}\,\mathrm{or}\,\underline{\mathbf{u}}_{e} = -d\,\underline{\mathbf{K}}^{-1}\,\underline{\mathbf{I}}_{N}^{\mathsf{T}}.$$
(3.25)

During the sticking phase, the last equation reduces to $m_N \ddot{u}_N + \underline{\mathbf{l}}_N \underline{\mathbf{u}} + k_{NN} d = R(t)$, or $m_N \ddot{u}_N = -F(t) + R(t)$, with

$$F(t) = \underline{\mathbf{l}}_{N}\underline{\mathbf{u}}(t) + k_{NN}d \le 0.$$
(3.26)

It should be understood that F(t) cannot be positive during the sticking phase. Otherwise, there exists t_0 during the sticking phase such that $F(t_0) > 0$. By continuity of the function F, it is strictly positive on an open interval including t_0 . By integrating the acceleration, $u_N(t_0) < d$ which contradicts that $u_N(t) = d$ during the sticking phase.

The solution $\underline{\mathbf{u}}$ of the sticking system is quasi-periodic and continuous and its mean value $\langle \underline{\mathbf{u}} \rangle$ is the equilibrium $\underline{\mathbf{u}}_e$. Therefore, the mean value of the scalar quasi-periodic function F is $\langle F \rangle = \underline{\mathbf{l}}_N \langle \underline{\mathbf{u}} \rangle + k_{NN}d = \underline{\mathbf{l}}_N \underline{\mathbf{u}}_e + k_{NN}d$. Since F is continuous, there exits $t_0 > 0$ such that $F(t_0) = \langle F \rangle = -d \underline{\mathbf{l}}_N \underline{\mathbf{K}}^{-1} \underline{\mathbf{l}}_N^{\top} + k_{NN}d = d\mathbf{X}^{\top} \mathbf{K} \mathbf{X}$ with $\mathbf{X} = [\underline{\mathbf{K}}^{-1} \underline{\mathbf{l}}_N^{\top}, -1]^{\top}$. $F(t_0)$ is positive because of d > 0 and the positive definiteness of the matrix \mathbf{K} , which contradicts (3.26). Therefore, the sticking duration is finite.

A priori, the conservation of energy during the sticking phase is not straightforward since the sticking system is a different system.

Lemma 3.16 — Energy during sticking phase. The solution to System (3.23)-(3.24) preserves the total energy *E* (3.1d).

Proof. Assume that t = 0 is the beginning of a sticking phase and $t = \tau$, the end. During this sticking phase on the interval $[0; \tau]$, the governing equations become $\underline{M}\ddot{\mathbf{u}} + \underline{K}\overline{\mathbf{u}} = \mathbf{0}$, with $\overline{\mathbf{u}} = \underline{\mathbf{u}} - \underline{\mathbf{u}}_e$ where $\underline{\mathbf{u}}_e = -d \underline{\mathbf{K}}^{-1} \underline{\mathbf{l}}_N^{\mathsf{T}}$ is the new equilibrium of the sticking system. This sticking system conserves the energy around the new equilibrium $\underline{\mathbf{u}}_e$:

$$\overline{E}(t) = \dot{\overline{\mathbf{u}}}^{\top}(t)\underline{\mathbf{M}}\dot{\overline{\mathbf{u}}}(t) + \overline{\mathbf{u}}^{\top}(t)\underline{\mathbf{K}}\overline{\mathbf{u}}(t) = \overline{E}(0).$$
(3.27)

Moreover, since $\underline{\mathbf{u}} = \overline{\mathbf{u}} + \underline{\mathbf{u}}_{\rho}$, an easy computation yields:

$$\underline{\underline{E}}(t) = \underline{\dot{\mathbf{u}}}^{\mathsf{T}}(t)\underline{\underline{\mathbf{M}}}\underline{\dot{\mathbf{u}}}(t) + \underline{\mathbf{u}}^{\mathsf{T}}(t)\underline{\underline{\mathbf{K}}}\underline{\mathbf{u}}(t) = \overline{\underline{E}}(t) + \underline{\underline{\mathbf{u}}}_{e}^{\mathsf{T}}\underline{\underline{\mathbf{K}}}\underline{\mathbf{u}}_{e} + 2\overline{\underline{\mathbf{u}}}^{\mathsf{T}}(t)\underline{\underline{\mathbf{K}}}\underline{\mathbf{u}}_{e}.$$
(3.28)

In particular,

$$\underline{E}(0) = \overline{E}(0) + \underline{\mathbf{u}}_{e}^{\mathsf{T}} \underline{\mathbf{K}} \underline{\mathbf{u}}_{e} + 2\overline{\mathbf{u}}^{\mathsf{T}}(0) \underline{\mathbf{K}} \underline{\mathbf{u}}_{e}.$$
(3.29)
The total energy of System (3.1) can be now calculated. Since $\dot{\mathbf{u}}$ is continuous along a sticking phase, the exponents \pm are dropped.

$$E(t) = \dot{\mathbf{u}}^{\mathsf{T}}(t)\mathbf{M}\dot{\mathbf{u}}(t) + \mathbf{u}^{\mathsf{T}}(t)\mathbf{K}\mathbf{u}(t)$$
(3.30)

$$= \underline{\mathbf{u}}^{\mathsf{T}}(t)\underline{\mathbf{M}}\underline{\mathbf{u}}(t) + \underline{\mathbf{u}}^{\mathsf{T}}(t)\underline{\mathbf{K}}\underline{\mathbf{u}}(t) + k_{NN}u_{N}^{2}(t) + 2u_{N}(t)\underline{\mathbf{l}}_{N}\underline{\mathbf{u}}(t)$$
(3.31)

$$= \underline{E}(t) + k_{NN}d^2 + 2d\underline{\mathbf{l}}_N\underline{\mathbf{u}}(t)$$
(3.32)

$$= \overline{E}(t) + \underline{\mathbf{u}}_{e}^{\mathsf{T}} \underline{\mathbf{K}} \underline{\mathbf{u}}_{e} + 2\overline{\mathbf{u}}^{\mathsf{T}}(t) \underline{\mathbf{K}} \underline{\mathbf{u}}_{e} + k_{NN} d^{2} + 2d \, \underline{\mathbf{l}}_{N}(\overline{\mathbf{u}}(t) + \underline{\mathbf{u}}_{e})$$
(3.33)

$$= \overline{E}(t) + \underline{\mathbf{u}}_{e}^{\mathsf{T}} \underline{\mathbf{K}}_{e} - 2d \,\overline{\mathbf{u}}^{\mathsf{T}}(t) \underline{\mathbf{l}}_{N}^{\mathsf{T}} + k_{NN} d^{2} + 2d \,\underline{\mathbf{l}}_{N}(\overline{\mathbf{u}}(t) + \underline{\mathbf{u}}_{e})$$
(3.34)

$$= \overline{E}(t) + \underline{\mathbf{u}}_{e}^{\mathsf{T}} \underline{\mathbf{K}} \underline{\mathbf{u}}_{e} + k_{NN} d^{2} + 2d \, \underline{\mathbf{l}}_{N} \underline{\mathbf{u}}_{e}.$$
(3.35)

Similarly,

$$E(0) = \underline{E}(0) + k_{NN}d^2 + 2d\,\underline{\mathbf{l}}_N\underline{\mathbf{u}}(0)$$
(3.36)

$$= \overline{E}(0) + \underline{\mathbf{u}}_{e}^{\mathsf{T}} \underline{\mathbf{K}} \underline{\mathbf{u}}_{e} + 2\overline{\mathbf{u}}^{\mathsf{T}}(0) \underline{\mathbf{K}} \underline{\mathbf{u}}_{e} + k_{NN} d^{2} + 2d \, \underline{\mathbf{l}}_{N} \underline{\mathbf{u}}(0)$$
(3.37)

$$= E(0) + \underline{\mathbf{u}}_{e}^{\mathsf{T}} \underline{\mathbf{K}}_{e} + k_{NN} d^{2} + 2d \, \underline{\mathbf{l}}_{N} \underline{\mathbf{u}}_{e}.$$
(3.38)

This follows that E(t) = E(0), i.e. the total energy of the system is a constant during the sticking phase.

3.3.2 Zero, one or infinity?

This section is devoted to the study of the number of closing contacts. The main results are stated in Theorem 3.3. The number of closing contacts can only be either 0, 1 or countably infinite. Consequently, if there are at least two closing contacts, then mass N will come in contact with the obstacle an infinite number of times. As a consequence, it is impossible to have a solution with 2, 3, ..., m closing contacts.

The proof of Theorem 3.3 is mainly grounded on the relative position of mass N with the wall. In case 1 of the theorem, system (3.1) is linear since the last mass moves within a free-flight and there is a unique expression for u_N . Similarly, in case 2 with only one grazing contact, the nonlinear term from the contact with zero velocity does not affect the solution of the linear system.

Between every two consecutive closing contacts, say $]t_j$; $t[_{j+1}$, system (3.1a) with r(t) = 0 is linear. There is a unique solution associated with the initial data at the exit time t_j . Hence, u_N has the unique form of a quasi-periodic function with distinct coefficients on each interval $]t_j$; $t_{j+1}[$.

The proof is divided into two steps. We first consider the quasi-periodic function φ defined on \mathbb{R} which coincides with u_N on a free-flight interval. The properties of almost periodic functions is called to study the behavior of φ . Then, the results obtained for φ are applied to obtain the

behavior of the solution $\mathbf{u}(t)$. The first step is contained in the following lemma.

Lemma 3.17 — Maximum of a quasi-periodic function. Let $\varphi(t)$ be a quasi-periodic function defined on \mathbb{R} such that

$$\varphi(t) = \sum_{j=1}^{N} (c_j \cos(\omega_j t) + s_j \sin(\omega_j t)) \quad \text{and} \quad \varphi(0) = \sum_{j=1}^{N} c_j = d > 0.$$
(3.39)

If $(\omega_1, \ldots, \omega_N)$ are \mathbb{Z} -independent, then there are two possibilities:

- 1. If $s_j = 0$ and $c_j \ge 0$ for all j = 1, ..., N then $\sup_{\mathbb{R}} \varphi = d$. Moreover, if φ is not periodic, i.e there exist at least two coefficients c_j and c_k with $j \ne k$ such that $c_j > 0$ and $c_k > 0$, then $\varphi(t) < d$ for all $t \ne 0$ and $\varphi(t) = d$ for only t = 0.
- 2. Otherwise, if there exists at least one $l \in \{1, ..., N\}$ such that $s_l \neq 0$ or $c_l < 0$ then $\sup_{\mathbb{R}} \varphi > d$.

The latter case is equivalent to saying that the converse of the first case is true. In the first case, $\varphi(t) = \sum_{j=1}^{N} c_j \cos(\omega_j t)$ and because of the \mathbb{Z} -independence of $\{\omega_j\}, \varphi$ is periodic if and only if there exists a unique $c_j > 0$ with $c_k = 0$ for all $k \neq j$. Then, the set $\{t : \varphi(t) = d\}$ is the infinite set $\{k T_j, k \in \mathbb{Z}\}$ where $T_j = 2\pi/\omega_j$. This case corresponds to the solution to system (3.1) with many grazing contacts. Discarding the periodic case, the value of φ can be very close to d but will never equal it again. This argument helps prove the case when the response of system (3.1) has only one grazing contact and never reaches the obstacle again.

Later, it is proven that the function φ with a supremum strictly greater than *d* corresponds to u_N of a solution with many closing contacts. Now, Lemma 3.17 is proven.

Proof.

- 1. Consider $s_j = 0$ and $c_j \ge 0$ for all j = 1, ..., N, then $\varphi(t) = \sum_{j=1}^N c_j \cos(\omega_j t)$ and $\sup_{\mathbb{R}} \varphi = \sum_{j=1}^N |c_j| = \sum_{j=1}^N c_j = d$. If $\exists t > 0$ such that $\varphi(t) = d$, i.e. $\sum_{j=1}^N c_j \cos(\omega_j t) = d$, then $\sum_{j=1}^N c_j(1 - \cos(\omega_j t)) = 0$ where $c_j(1 - \cos(\omega_j t)) \ge 0$ for all j = 1, ..., N. Hence, $\cos(\omega_j t) = 1$ for all j = 1, ..., N. Thus, $\omega_j t = k_j 2\pi, k \in \mathbb{Z}$, which contradicts the \mathbb{Z} -independence assumption. Hence, φ is always smaller than d for t > 0.
- 2. Otherwise, if there exists $l \in \{1, ..., N\}$ such that $s_l \neq 0$, or $c_l < 0$, then $\sup_{\mathbb{R}} \varphi = \sum_{j=1}^N \sqrt{c_j^2 + s_j^2} \ge \max(\sum_{j=1}^N \sqrt{c_j^2 + s_j^2}, \sum_{j=1}^N |c_j|) > \sum_{j=1}^N c_j = d$. Hence, there exists t > 0 such that $\varphi(t) = d$.

Now, using Lemma 3.17, Theorem 3.3 is proven.

Proof. The solution to (3.1a) when R(t) = 0 is for $t \in \mathbb{R}$

$$\Phi(t) = \mathbf{P}\cos(t\mathbf{\Omega})\mathbf{P}^{-1}\mathbf{u}(0) + \mathbf{P}\mathbf{\Omega}^{-1}\sin(t\mathbf{\Omega})\mathbf{P}^{-1}\dot{\mathbf{u}}(0),$$

$$= \sum_{j=1}^{N} (c_j\cos(\omega_j t) + s_j\sin(\omega_j t))\mathbf{P}\mathbf{e}_j,$$
(3.40)

where $\mathbf{P} \mathbf{e}_j$, j = 1, ..., N are eigenvectors corresponding to eigenvalues ω_j^2 of $\mathbf{M}^{-1}\mathbf{K}$. We can choose $\mathbf{P} \mathbf{e}_j$ such that $P_{Nj} = 1, j = 1, ..., N$. It is clear that $\mathbf{u}(t) = \mathbf{\Phi}(t)$ as long as $u_N(t) < d$.

Henceforth, consider the solution $\mathbf{u}(t)$ to the vibro-impact system (3.1). If $u_N(t) < d$, for all $t \ge 0$, then the solution of the linear problem does not impact the wall.

Since the system is autonomous, the impacting time is chosen to be t = 0, i.e. $u_N(0) = d$. Two possibilities based on the velocity of the N^{th} mass at the contact time t = 0 are considered: If $\dot{u}_N^-(0) > 0$, then $\dot{u}_N^+(0) < 0$. Let φ be a function defined on \mathbb{R} such that $\varphi(t) = u_N(t)$ on $\{t, u_N(t) < d\}$. It follows that $\varphi(0) = d$ and $\dot{\varphi}(0) < 0$, thus there exists $\tau > 0$ such that $\varphi(t) > d$ for $t \in]-\tau$; 0[. Therefore, $\sup_{\mathbb{R}^-} \varphi \ge \sup_{]-\tau; 0[} \varphi > d$. For an almost periodic function φ , the supremum taken on \mathbb{R}^- is also the one taken on \mathbb{R}^+ [13], this yields $\sup_{\mathbb{R}^+} \varphi > d$. By Lemma 3.17, the first instant $t_1 > 0$ such that $\varphi(t_1) = d$ exists, i.e. mass N will come back to the obstacle at time t_1 .

If $\dot{u}_N^-(0) = 0$ then $\dot{u}_N^+(0) = 0$ and $\sum_{j=1}^N c_j = d$, $\sum_{j=1}^N s_j \omega_j = 0$. Let φ be the function defined on \mathbb{R} so that it coincides with u_N when $u_N(t) < d$ before the contact. If $s_j = 0$ and $c_j \ge 0$ for all j = 1, ..., N, by Lemma 3.17, it follows that $\sup_{\mathbb{R}} \varphi = d$ and $\varphi(t) < d$ for all t > 0. Thus, the solution has only one grazing contact and it can be seen as the counterpart of the linear system without contact.

Otherwise, again, Lemma 3.17 shows that $\sup_{\mathbb{R}} \varphi > d$. It follows that there is a sticking phase from 0 to τ , i.e. $u_N(t) = d$ for all $t \in [0, \tau]$ and $u_N(t) < d$ for $\tau < t < \tau + \delta$, where $\delta > 0$. Assume that **w** is the solution to (3.1a) after the sticking time, associated to the new initial data at $t = \tau$. Thus, w_N has the expression

$$w_N(t) = \sum_{j=1}^N \left(\underline{c}_j \cos(\omega_j(t-\tau)) + \underline{s}_j \sin(\omega_j(t-\tau))\right), \quad t \ge \tau.$$
(3.41)

Let $\underline{\varphi}$ be the function defined on \mathbb{R} satisfying $\underline{\varphi}(t) = w_N(t)$ on $\{t : w_N(t) < d\}$. Consider the solution **u** to the vibro-impact system (3.1) just before and just after the sticking phase, $u_N(t)$ can be written as

$$u_N(t) = \begin{cases} \varphi(t) & \text{for } t \leq 0, \\ d & \text{for } t \in [0; \tau], \\ \underline{\varphi}(t) & \text{for } t \geq \tau. \end{cases}$$
(3.42)

We then show that $\sup_{[\tau;\infty[} \underline{\varphi} > d$. If not, $\sup_{\mathbb{R}} \underline{\varphi} = d$, then $\underline{\varphi}(t) < d$, for all $t \in \mathbb{R}$ and $\underline{\varphi}(t) = d$ for $t = \tau$ only. By reversibility and uniqueness of the solution, the solution to (3.1) is always without contact and there is only one grazing contact at τ . It is in contradiction with the fact that the solution involved the sticking phase. As a consequence, $\sup_{\mathbb{R}} \underline{\varphi} > d$. Hence, there exists $t > \tau$ such that $\varphi(t) = d$.

The process can be continued in both cases, thus the set of closing contacts is countably infinite. $\hfill \Box$

3.3.3 Poincaré section

This part deals with the construction of the Poincaré section on which the first return map is well-defined. Consider an orbit $[\mathbf{u}, \dot{\mathbf{u}}] \subset \mathbb{R}^{2N}$ of (3.1). To study the dynamics near such an orbit, we use the first return map (or Poincaré map). This map is defined on a Poincaré section which is classically a (2N - 1)-dimensional manifold in \mathbb{R}^{2N} that contains a point $\mathbf{U}(t) = [\mathbf{u}(t), \dot{\mathbf{u}}(t)]$ of the orbit and is transverse to the orbit at $\mathbf{U}(t)$. In the current work, the transversality is lost on \mathcal{H}^0 . An orbit starting on \mathcal{H}^0 may not intersect \mathcal{H} again. Hence, an important task is to eliminate the set of data such that the associated orbit does not intersect $\mathcal{H}_{\mathcal{P}}$ again. This is achieved by investigating the structure of \mathcal{H}^0 . By Theorem 3.3, there are two possibilities after the grazing contact: the orbit never comes in contact again (\mathcal{H}^0_1) or the orbit comes in contact an infinite number of time (\mathcal{H}^0_∞) . The explicit description of \mathcal{H}^0_1 is studied in the proposition below. Recall that $\mathbf{B} = (B_{ij})_{i,i=1}^N$ denotes \mathbf{P}^{-1} .

Proposition 3.18 — \mathcal{H}_1^0 and solutions with only one contact. The set \mathcal{H}_1^0 of initial data on \mathcal{H} such that the associated orbits have only one contact is the subset

$$\{ [\mathbf{u}^{\mathsf{T}}, \dot{\mathbf{u}}^{\mathsf{T}}]^{\mathsf{T}} \in \mathbb{R}^{2N}, \, u_N = d, \, \dot{u}_k = 0, \, k = 1, \dots, N \}$$
(3.43)

of a (N - 1)-dimensional affine subspace, such that the N - 1 components $u_1, u_2, \ldots, u_{N-1}$ satisfy N inequalities:

$$P_{Nk} \sum_{j=1}^{N-1} B_{kj} u_j \ge -P_{Nk} B_{kN} d, \qquad \forall k = 1, \dots, N$$
(3.44)

of which at least two are strict inequalities.

Note that \mathcal{H}_1^0 can be empty.

Proof. The solution to (3.1) with a unique grazing contact is analytic and $u_N(t)$ has a closed

form-expression:

$$u_N(t) = \mathbf{e}_N^{\mathsf{T}} \left(\mathbf{P} \cos(t\mathbf{\Omega}) \mathbf{P}^{-1} \mathbf{u}(0) + \mathbf{P} \mathbf{\Omega}^{-1} \sin(t\mathbf{\Omega}) \mathbf{P}^{-1} \dot{\mathbf{u}}(0) \right),$$
(3.45)

$$=\sum_{k=1}^{N} (c_k \cos(\omega_k t) + s_k \sin(\omega_k t)), \qquad (3.46)$$

where $c_k = P_{Nk} \mathbf{e}_k^{\mathsf{T}} \mathbf{P}^{-1} \mathbf{u}(0)$, $s_k = P_{Nk} \omega_k^{-1} \mathbf{e}_k^{\mathsf{T}} \mathbf{P}^{-1} \dot{\mathbf{u}}(0)$, k = 1, ..., N. Since $[\mathbf{u}^{\mathsf{T}}, \dot{\mathbf{u}}^{\mathsf{T}}]^{\mathsf{T}}$ belongs to \mathcal{H}_1^0 , this corresponds to case 1 of Lemma 3.17, thus, $u_N(0) = d$ and $\dot{u}_N(0) = 0$, and the coefficients c_k , s_k satisfy

$$c_k \ge 0, \quad \forall k = 1, \dots, N, \quad \exists l \ne m : c_l > 0, c_m > 0,$$
(3.47)

$$s_k = 0, \quad \forall k = 1, \dots, N. \tag{3.48}$$

This gives

$$P_{Nk} \mathbf{e}_k^{\mathsf{T}} \mathbf{P}^{-1} \mathbf{u}(0) \ge 0, \quad \forall k = 1, \dots, N, \quad \exists l \neq m : c_l > 0, c_m > 0,$$
 (3.49)

$$P_{Nk}\omega_k^{-1} \mathbf{e}_k^{\mathsf{T}} \mathbf{P}^{-1} \dot{\mathbf{u}}(0) = 0, \quad \forall k = 1, \dots, N.$$
(3.50)

The second condition yields a linear system $\mathbf{P}^{-1} \dot{\mathbf{u}}(0) = \mathbf{0}$, since $\det(\mathbf{P}^{-1}) \neq 0$, it follows that $\dot{\mathbf{u}}(0) = \mathbf{0}$ and u_1, \ldots, u_{N-1} satisfy the inequalities (3.49). This gives the explicit formula of \mathcal{H}_1^0 stated in Proposition 3.18.

An immediate consequence of Proposition 3.18 is that most orbits which belong to \mathcal{H}^0 are in \mathcal{H}^0_{∞} . To be precise, the set of initial data such that the associated orbits belong to \mathcal{H}^0_1 has a (2N - 2)-dimensional zero measure in \mathcal{H}^0 . As a consequence, the Poincaré section chosen in Definition 3.4 is reasonable.

Corollary 3.19 — **Domain of definition of the first return map.** The maximal subset of \mathcal{H} where the first return map \mathcal{F} is well-defined is $\mathcal{H}_{\mathcal{P}} = \mathcal{H}^- \cup \mathcal{H}^0_{\infty}$.

Some consequences of Proposition 3.18 are now stated.

Corollary 3.20. The orbits including a sticking phase have an infinite number of closing contacts.

Proof. An orbit including a sticking phase intersects \mathcal{H}^0 . As proven in Proposition 3.18, the initial data must be in \mathcal{H}^0_{∞} since \mathcal{H}^0_1 only involves the data on \mathcal{H} such that the orbits have a unique grazing contact but no sticking phase. Hence, it belongs to the set with an infinite number of closing contacts.

Corollary 3.21 — Infinite number of closing contacts. If the orbit intersects \mathcal{H}^- then it intersects \mathcal{H}^- infinitely many times in the future and in the past.

Proof. By Theorem 3.3, the orbit corresponding to the initial data in \mathcal{H}^- experiences at least one impact at t = 0. This eliminates the possibilities of case 1 and case 2 in the theorem. Hence, the orbit belongs to the third category, which means that there will be an infinite number of closing contacts. However, if the set of closing contacts involves only grazing contacts in the future, the system becomes linear with $u_N(t) \leq d$ for all t > 0. Let φ be a function defined on \mathbb{R} with $\varphi(t) = u_N(t)$, it follows that $\sup_{\mathbb{R}} \varphi = \sup_{t>0} \varphi = d$. This contradicts the fact that the supremum of φ must be greater than d since there is at least one impact at t = 0. The process can be repeated to get another closing contact and so on.

Denote by \mathcal{H}^{--} a subset of \mathcal{H}^{-} , containing all the data such that the corresponding orbits have impacts only. It seems to be dense in \mathcal{H}^{-} .

About the set \mathcal{H}_1^0 for 2-dof systems

Corollary 3.22 — \mathcal{H}_1^0 for a 2-dof system. Consider a 2-dof system (3.1), then from Proposition 3.18, \mathcal{H}_1^0 is defined as

$$\mathcal{H}_{1}^{0} = \{ [\mathbf{u}^{\top}, \dot{\mathbf{u}}^{\top}]^{\top} \in \mathbb{R}^{4} : P_{2k}B_{k1}u_{1} > -P_{2k}B_{k2}d, \ k = 1, 2, \ u_{2} = d, \ \dot{u}_{1} = \dot{u}_{2} = 0 \}$$
(3.51)
$$= D_{u_{1}} \times \{d\} \times \{0\} \times \{0\}$$
(3.52)

$$= D_{u_1} \times \{d\} \times \{0\} \times \{0\}$$
(3.52)

where D_{u_1} is the subset of \mathbb{R} including all the values u_1 satisfying the two strict inequalities:

$$P_{21}B_{11}u_1 > -P_{21}B_{12}d \text{ and } P_{22}B_{21}u_1 > -P_{22}B_{22}d.$$
 (3.53)

Therefore, the set D_{u_1} can be either void or an open interval of the form (b, ∞) , $(-\infty, a)$, or (a, b), with $a = \min\{\alpha_1, \alpha_2\}$, $b = \max\{\alpha_1, \alpha_2\}$, where $\alpha_1 = -dB_{22}/B_{21}$ and $\alpha_2 = -dB_{12}/B_{11}$.

Remark 3.23 — Linear gazing modes of a 2-dof system and the boundary of \mathcal{H}_1^0 . The scalars α_1 and α_2 are the distinct initial values of u_1 corresponding to the first and the second linear grazing orbit.

The set \mathcal{H}_1^0 is an interval in the two-dimensional space \mathcal{H}^0 . This means that the set of initial data such that the associated orbits have only one grazing contact is a very small subset of all the initial data such that orbits contain zero velocity contacts.

3.4 Implicit function theorem and power-root singularity

This section is divided into two parts. The square-root singularity is studied in the first part. This singularity implies the square-root dynamics near the linear grazing orbits as stated in Section 3.6. A general power-root singularity is investigated in the second part.

3.4.1 The square-root singularity

The first return time *T* is implicitly defined from the equation f(T, W) = 0 meaning $u_N(T) = d$ in Section 3.6. Unfortunately, if the contact at T_0 is grazing, $\partial_t f(T_0, W_0)$ vanishes. The square-root singularity is expected along the intersection of the hypersurfaces defined by f = 0 and $\partial_t f = 0$. This section gives a general approach for a general function *f* to show this emergence of the square-root singularity. The main tool is to use implicit function theorem on a variable other than *t* and then use Taylor expansion of the function obtained from the implicit function theorem to get back to the *t* variable as a function of W. We state the results in two cases corresponding to the function *f* defined in a two-dimensional and a m + 2-dimensional spaces with $m \ge 1$.

In two dimensions

Let *f* be a function of two variables *x* and *y*, where the relation of *x* with respect to *y* is implicitly given by f(x, y) = 0. Let us write *x* locally as a function of *y* when *f* satisfies some unusual conditions as follows.

Theorem 3.24—Square-root singularity in two dimensions. Suppose that $f(x, y) \in C^3(\mathbb{R}^2, \mathbb{R})$ satisfies the following conditions at (x_0, y_0) :

- 1. $f(x_0, y_0) = 0$,
- 2. $\partial_x f(x_0, y_0) = 0$,
- 3. $\partial_{xx} f(x_0, y_0) \neq 0$,
- 4. $\partial_{y} f(x_0, y_0) \neq 0$.

Denote the ratio $2\gamma = -\partial_{xx} f(x_0, y_0)/\partial_y f(x_0, y_0) \neq 0$ and $s_{\gamma} = \operatorname{sign} \gamma$. There exist two intervals *I* and *J* containing y_0 and x_0 , respectively, as well as a function $\psi \in C^3(I_{\gamma}, J)$ defined on the subinterval

$$I_{\gamma} = \{ y \in I, \, s_{\gamma}(y - y_0) \ge 0 \}$$
(3.54)

such that the set $\{(x, y) \in I \times J, f(x, y) = 0\}$ is form by the following two branches near (x_0, y_0) :

$$x = x_0 + \psi(\sqrt{s_\gamma(y - y_0)}) \text{ and } x = x_0 + \psi(-\sqrt{s_\gamma(y - y_0)}).$$
 (3.55)

Moreover, ψ satisfies $\psi(0) = 0$ and $\dot{\psi}(0) = 1/\sqrt{|\gamma|} \neq 0$.

An illustration of this theorem is given in Figure 3.1. Note that ψ is not defined on the whole open interval *I* but only on half of it which is determined by the sign of γ .

Proof. Without loss of generality, assume that (x_0, y_0) is at the origin. If not, consider the new variables $x^* = x - x_0$ and $y^* = y - y_0$, and note that at the point (0,0), the function



FIGURE 3.1 – Two branches on the right of the line $y = y_0$ when $\gamma > 0$.

 $f^*(x^*, y^*) = f(x_0 + x^*, y_0 + y^*)$ satisfies the conditions of the theorem. The conclusions are then modified accordingly.

The implicit function theorem ensures the existence of the interval V_x of $x_0 = 0$ and V_y of $y_0 = 0$, and a unique function $y = \varphi(x)$ for $x \in V_x$ and $y \in V_y$, such that $f(x, \varphi(x)) = 0$ for $x \in V_x$, $\varphi(0) = 0$, $\dot{\varphi}(0) = -\partial_x f(0, 0)/\partial_y f(0, 0) = 0$ and $\ddot{\varphi}(0) = -\partial_{xx} f(0, 0)/\partial_y f(0, 0) = 2\gamma \neq 0$. Moreover, φ is as smooth as f. The Taylor expansion of φ with an integral remainder at 0 is

$$y = \varphi(x) = x^2 r(x), \quad r(x) = \int_0^1 (1-s) \ddot{\varphi}(sx) ds$$
 (3.56)

where $r \in C^1(\mathbb{R}, \mathbb{R})$ and $r(0) = \ddot{\varphi}(0)/2 = \gamma \neq 0$. The expression of *x* with respect to *y* depends on the sign of $\ddot{\varphi}(0)$ (or the sign of γ) as follows:

- If $\gamma > 0$ then r(0) > 0 and by the continuity of r, it follows that r(x) > 0 in some neighborhood of 0. On that neighborhood, define the continuously differentiable function $\phi(x) = x\sqrt{r(x)}$: at 0, ϕ has a nonzero derivative since $r(0) \neq 0$. Through the inverse function theorem, there exists an inverse function ϕ^{-1} in some neighborhood I of $\phi(0) = 0$. From $y = \phi^2(x)$, it is required that $y \ge 0$. It entails I_{γ} is simply a right neighborhood of 0. Hence, for $y \in I_{\gamma}, x = \phi^{-1}(\sqrt{y})$ or $x = \phi^{-1}(-\sqrt{y})$.
- If $\gamma < 0$ then $y = -\phi^2(x)$ with $\phi(x) = x\sqrt{-r(x)}$. A similar proof holds for y belongs to I_{γ} which is now a left neighborhood of 0. The conclusion is obtained by replacing \sqrt{y} by $\sqrt{-y}$ and the function ψ satisfies $\dot{\psi}(0) = \dot{\phi}^{-1}(0) = 1/\sqrt{-\gamma}$.

In both cases, by denoting $\psi = \phi^{-1}$, we have shown that there exists a function ψ defined in a neighborhood I_{γ} of 0 satisfying $\psi(0) = 0$ and $\dot{\psi}(0) = \dot{\phi}^{-1}(0) = 1/\sqrt{|\gamma|}$ such that $x = \psi(\sqrt{s_{\gamma}y})$ or $x = \psi(-\sqrt{s_{\gamma}y})$. This concludes the proof.

Remark 3.25. Another way to express the square-root singularity of x is $x = \sqrt{\beta(y)}$. However, with the present expression, say $x = \lambda(\sqrt{y})$, we have a better regularity of λ since λ is smooth but β is not always smooth. For example, if $\lambda(y) = y + y^2$, $y \in \mathbb{R}^+$, then $\lambda \in C^{\infty}(\mathbb{R}^+)$. However,

 $\beta(y) = \lambda^2(\sqrt{y}) = (\sqrt{y} + y)^2 = y + 2y\sqrt{y} + y^2$ is not so smooth.

In m + 2 dimensions($m \ge 1$)

Consider a general function f of m + 2 variables with a non-zero gradient, the square-root singularity is shown.

Theorem 3.26 — Square-root singularity in m + 2 dimensions. Consider the smooth function $f(x, y, \mathbf{z}) : \mathbb{R} \times \mathbb{R} \times \mathbb{R}^m \to \mathbb{R}$ such that at point $\mathbf{X}_0 = (x_0, y_0, \mathbf{z}_0)$, it satisfies the following conditions:

- 1. $f(\mathbf{X}_0) = 0$,
- 2. $\partial_x f(\mathbf{X}_0) = 0$,
- 3. $\partial_{xx} f(\mathbf{X}_0) \neq 0$,
- 4. $\partial_{y} f(\mathbf{X}_{0}) \neq 0.$

Denote $2\gamma = -\partial_{xx} f(\mathbf{X}_0) / \partial_y f(\mathbf{X}_0) \neq 0$ and $s_{\gamma} = \operatorname{sign} \gamma$. Then, there exist three neighborhoods V_{x_0}, V_{y_0} , and $V_{\mathbf{z}_0}$ of x_0, y_0 and \mathbf{z}_0 respectively, and two smooth scalar functions $\eta : V_{\mathbf{z}_0} \to V_{x_0}$ and $\alpha : V_{\mathbf{z}_0} \to V_{y_0}$ satisfying $\eta(\mathbf{z}_0) = x_0$ and $\alpha(\mathbf{z}_0) = y_0$ such that the set

$$S_c = \{ (x, y, \mathbf{z}) \in \mathbb{R} \times \mathbb{R} \times \mathbb{R}^m, \ f(x, y, \mathbf{z}) = 0 \text{ and } \partial_x f(x, y, \mathbf{z}) = 0 \}$$
(3.57)

is parameterized by $S_c \cap \Omega = \{(\eta(\mathbf{z}), \alpha(\mathbf{z}), \mathbf{z}), \mathbf{z} \in V_{\mathbf{z}_0}\}$ where $\Omega = V_{x_0} \times V_{y_0} \times V_{\mathbf{z}_0}$. Let B_{γ} be a



FIGURE 3.2 – Square-root singularity in dimension 3.

subset of $V_{y_0} \times V_{\mathbf{z}_0}$ where

$$B_{\gamma} = \{ (y, \mathbf{z}) \in V_{y_0} \times V_{\mathbf{z}_0}, \, s_{\gamma}(y - \alpha(\mathbf{z})) \ge 0 \}.$$
(3.58)

There exists a smooth scalar function ψ defined on B_{γ} such that there are two graphs:

$$x = \eta(\mathbf{z}) + \psi(\sqrt{s_{\gamma}(y - \alpha(\mathbf{z}))}, \mathbf{z}), \quad (y, \mathbf{z}) \in B_{\gamma},$$
(3.59)

and

$$x = \eta(\mathbf{z}) + \psi(-\sqrt{s_{\gamma}(y - \alpha(\mathbf{z}))}, \mathbf{z}), \quad (y, \mathbf{z}) \in B_{\gamma}.$$
(3.60)

In particular, $\psi(0, \mathbf{z}_0) = 0$ and $\partial_y \psi(0, \mathbf{z}_0) = 1/\sqrt{|\gamma|}$.

An illustration of the square-root singularity in 3 dimensions is depicted in Figure 3.2.

The function ψ is not defined on the whole neighborhood $V_{y_0} \times V_{z_0}$, but only in one part of it, which is B_{γ} . The subset B_{γ} defined in the theorem gives the one-sided condition from which the square-root singularity arises. When the sign of γ is known, B_{γ} is simply a region above or below and including the hypersurface $y = \alpha(\mathbf{z})$, if γ is positive or negative, respectively. A more general result when the condition (3) above does not hold is discussed in Remark 3.27.

Proof. Let S_c be the manifold of *m* dimensions which is the intersection of the two hypersurfaces of dimension m + 1 near (x_0, y_0, \mathbf{z}_0) , $S_c = S^0 \cap S^1$ where $S^0 = \{(x, y, \mathbf{z}) \in \Omega, f(x, y, \mathbf{z}) = 0\}$ and $S^1 = \{(x, y, \mathbf{z}) \in \Omega, G(x, y, \mathbf{z}) = 0\}$. Moreover, it can be parameterized in a neighborhood of \mathbf{z} by two parametric functions η and α as follows.

From (1) and (4), by the implicit function theorem, there exist two neighborhoods V of (x_0, \mathbf{z}_0) and W of y_0 , and a function $\varphi : V \to W$, $(x, \mathbf{z}) \mapsto y = \varphi(x, \mathbf{z})$ such that $f(x, \varphi(x, \mathbf{z}), \mathbf{z}) = 0$ for $(x, \mathbf{z}) \in V$, $\varphi(x_0, \mathbf{z}_0) = y_0$. In particular, $\partial_x \varphi(x_0, \mathbf{z}_0) = 0$ and $\partial_{xx} \varphi(x_0, \mathbf{z}_0) =$ $-\partial_{xx} f(x_0, y_0, \mathbf{z}_0) / \partial_y f(x_0, y_0, \mathbf{z}_0) = 2\gamma \neq 0$. Moreover, φ has the same smoothness as f.

For $(x, \mathbf{z}) \in V$, once again we apply the implicit function theorem for the function $G(x, \mathbf{z}) = \partial_x f(x, \varphi(x, \mathbf{z}), \mathbf{z})$ which satisfies:

$$G(x_0, \mathbf{z}_0) = \partial_x f(x_0, y_0, \mathbf{z}_0) = 0,$$

$$\partial_x G(x_0, \mathbf{z}_0) = \partial_{xx} f(x_0, y_0, \mathbf{z}_0) + \partial_{yx} f(x_0, y_0, \mathbf{z}_0) \partial_x \varphi(x_0, \mathbf{z}_0) = \partial_{xx} f(x_0, a_0, \mathbf{z}_0) \neq 0.$$
(3.61)
(3.62)

This gives the existence of the neighborhoods $B_{\mathbf{z}_0}$ of \mathbf{z}_0 and B_{x_0} of x_0 such that there is a smooth function $\eta : B_{\mathbf{z}_0} \to B_{x_0}$, $\mathbf{z} \mapsto x = \eta(\mathbf{z})$ satisfying $x_0 = \eta(\mathbf{z}_0)$, $G(\eta(\mathbf{z}), \mathbf{z}) = 0$ for all $\mathbf{z} \in B_{\mathbf{z}_0}$. It follows that $\partial_x f(\eta(\mathbf{z}), \alpha(\mathbf{z}), \mathbf{z}) = 0$ for all $\mathbf{z} \in B_{\mathbf{z}_0}$ where $\alpha(\mathbf{z}) := \varphi(\eta(\mathbf{z}), \mathbf{z})$ belongs to a neighborhood of y_0 , denoted V_{y_0} . In particular, $\alpha(\mathbf{z}_0) = \varphi(x_0, \mathbf{z}_0) = y_0$.

As a consequence, there exists a local parameterization of S_c :

$$S_{c} = \{ (x, y, \mathbf{z}) = (\eta(\mathbf{z}), \, \alpha(\mathbf{z}), \, \mathbf{z}), \, \mathbf{z} \in B_{\mathbf{z}_{0}} \}.$$
(3.63)

This parameterization can be seen directly by using the implicit function theorem for the vector

function $H = (f, \partial_x f)$ with the invertible matrix

$$D_{x,y}H(\mathbf{X}_0) = \begin{bmatrix} \partial_x f & \partial_y f \\ \partial_{xx} f & \partial_{yx} f \end{bmatrix} (\mathbf{X}_0) = \begin{bmatrix} 0 & \neq 0 \\ \neq 0 & \partial_{yx} f(\mathbf{X}_0) \end{bmatrix}.$$
(3.64)

We will now show that the square-root singularity arises along the hypersurface $y = \alpha(\mathbf{z})$. For each fixed $\mathbf{z} \in B_{\mathbf{z}_0}$, the Taylor expansion with an integral remainder of φ with respect to x near $\eta(\mathbf{z})$ reads

$$\varphi(x, \mathbf{z}) = \varphi(\eta(\mathbf{z}), \mathbf{z}) + \partial_x \varphi(\eta(\mathbf{z}), \mathbf{z}) (x - \eta(\mathbf{z}))$$
(3.65)

$$+ (x - \eta(\mathbf{z}))^2 \int_0^1 (1 - s)\partial_{xx}\varphi\left(s(x - \eta(\mathbf{z})) + \eta(\mathbf{z}), \mathbf{z}\right) ds.$$
(3.66)

Since $y = \varphi(x, \mathbf{z}), \partial_x \varphi(\eta(\mathbf{z}), \mathbf{z}) = 0, \varphi(\eta(\mathbf{z}), \mathbf{z}) = \alpha(\mathbf{z})$, this yields

$$y - \alpha(\mathbf{z}) = (x - \eta(\mathbf{z}))^2 r(x - \eta(\mathbf{z}), \mathbf{z}), \tag{3.67}$$

where $r(x-\eta(\mathbf{z}), \mathbf{z})$ is a smooth function with respect to x and z, satisfying $r(0, \mathbf{z}) = \partial_{xx} \varphi(\eta(\mathbf{z}), \mathbf{z})/2$.

The remainder $r(x - \eta(\mathbf{z}), \mathbf{z})$ has the sign of γ for all \mathbf{z} in a neighborhood of \mathbf{z}_0 . Consider the function $\gamma(\mathbf{z}) = r(0, \mathbf{z}), \gamma(\mathbf{z})$: it is as smooth as r and satisfies $\gamma(\mathbf{z}_0) = \partial_{xx}\varphi(x_0, \mathbf{z}_0)/2 = \gamma \neq 0$. Without loss of generality, assume that $\gamma > 0$. By continuity, $\gamma(\mathbf{z}) > 0$ in some neighborhood of \mathbf{z}_0 , say $V_{\mathbf{z}_0}$. It follows that $r(0, \mathbf{z}) > 0$ for $\mathbf{z} \in V_{\mathbf{z}_0}$. Similarly, by the continuity of r with respect to $x, r(x, \mathbf{z}) > 0$ for all x in a neighborhood of $\eta(\mathbf{z})$.

Let $B_{\gamma} \subset V_{y_0} \times V_{\mathbf{z}_0}$ be the region adjacent to and including the hypersurface $y = \alpha(\mathbf{z})$ such that

$$B_{\gamma} = \{ (y, \mathbf{z}) \in V_{y_0} \times V_{\mathbf{z}_0}, \, \gamma(y - \alpha(\mathbf{z})) \ge 0 \}.$$
(3.68)

Denote by ϕ the function $\phi(x - \eta(\mathbf{z}), \mathbf{z}) = (x - \eta(\mathbf{z}))\sqrt{r(x - \eta(\mathbf{z}), \mathbf{z})}$, (3.67) then becomes

$$y - \alpha(\mathbf{z}) = (\phi(x - \eta(\mathbf{z}), \mathbf{z}))^2.$$
(3.69)

This follows that $y - \alpha(\mathbf{z}) \ge 0$. The subset B_{γ} is then a region above and including the hypersurface $y = \alpha(\mathbf{z})$. Since ϕ satisfies $\phi(0, \mathbf{z}) = 0$, $\partial_x \phi(0, \mathbf{z}) = \sqrt{r(0, \mathbf{z})} = \sqrt{\gamma(\mathbf{z})} \ne 0$, there exists an inverse function ϕ^{-1} in some neighborhood of 0. The inverse function theorem is used uniformly with respect to the parameter \mathbf{z} and eventually reduces the neighborhood of \mathbf{z}_0 and x_0 . Therefore, for $(y, \mathbf{z}) \in B_{\gamma}$, (3.69) yields $\phi(x - \eta(\mathbf{z}), \mathbf{z}) = \sqrt{y - \alpha(\mathbf{z})}$ or $\phi(x - \eta(\mathbf{z}), \mathbf{z}) = -\sqrt{y - \alpha(\mathbf{z})}$. This gives $x = \eta(\mathbf{z}) + \phi^{-1}(\sqrt{y - \alpha(\mathbf{z})}, \mathbf{z})$ or $x = \eta(\mathbf{z}) + \phi^{-1}(-\sqrt{y - \alpha(\mathbf{z})}, \mathbf{z})$. In particular, $\phi^{-1}(0, \mathbf{z}) = 0$ and $\partial_y \phi^{-1}(0, \mathbf{z}) = 1/\sqrt{\gamma(\mathbf{z})}$.

Similarly, for the case $\gamma < 0$, $r(x - \eta(\mathbf{z}), \mathbf{z}) < 0$ for x in some neighborhood of $\eta(\mathbf{z})$, it follows

that $y - \alpha(\mathbf{z}) = -\phi^2(x - \eta(\mathbf{z}), \mathbf{z})$, where $\phi(x - \eta(\mathbf{z}), \mathbf{z}) = (x - \eta(\mathbf{z}))\sqrt{-r(x - \eta(\mathbf{z}), \mathbf{z})}$. The subset B_{γ} is now the region below and including the hypersurface $y = \alpha(\mathbf{z})$. The same conclusions hold with a change of $\sqrt{y - \alpha(\mathbf{z})}$ in $\sqrt{\alpha(\mathbf{z}) - y}$ and $\dot{\phi}^{-1}(0, \mathbf{z}) = 1/\sqrt{-\gamma(\mathbf{z})}$.

Consequently, by denoting $\psi = \phi^{-1}$, we have shown that ψ is defined in B_{γ} such that

$$x = \eta(\mathbf{z}) + \psi(\sqrt{s_{\gamma}(y - \alpha(\mathbf{z}))}, \mathbf{z}) \quad \text{or} \quad x = \eta(\mathbf{z}) + \psi(-\sqrt{s_{\gamma}(y - \alpha(\mathbf{z}))}, \mathbf{z}).$$
(3.70)

In particular, $\psi(0, \mathbf{z}_0) = 0$, $\partial_y \psi(0, \mathbf{z}_0) = 1/\sqrt{\gamma(\mathbf{z}_0)} = 1/\sqrt{|\gamma|}$. This ends the proof.

3.4.2 Power-root singularity

The square-root singularity shown in Theorem 3.26 relies on the non-vanishing second derivative $\partial_{xx} f(x_0, y_0, \mathbf{z}_0)$. If it vanishes to zero, then higher order derivatives of f must be considered. The following remark gives analogous conditions from which we define power-root singularity.

Remark 3.27. Let $n \le m + 1$ and $f(x, y, \mathbf{z}) : \mathbb{R} \times \mathbb{R} \times \mathbb{R}^m \to \mathbb{R}$ be a smooth function, such that at point $\mathbf{X}_0 = (x_0, y_0, \mathbf{z}_0)$, it satisfies the following conditions:

1. $f(\mathbf{X}_0) = \partial_x f(\mathbf{X}_0) = \partial_{xx} f(\mathbf{X}_0) = \dots = \partial_{x^{n-1}}^{n-1} f(\mathbf{X}_0) = 0, \ \partial_{x^n}^n f(\mathbf{X}_0) \neq 0, \ n \ge 3,$ 2. $\partial_y f(\mathbf{X}_0) \neq 0,$

3. linear independance of vectors $\nabla_{(x,y,\mathbf{z})} \partial_x^{\ell} f(x_0, y_0, \mathbf{z}_0), 0 \leq \ell < n$,

then a n^{th} -root singularity emerges.

Let us explain this remark with more details. Denote $\gamma = -f^{(n)}(\mathbf{X}_0)/n!\partial_y f(\mathbf{X}_0) \neq 0$ and $s_{\gamma} = \operatorname{sign} \gamma$. There exist three neighborhoods V_{x_0} , V_{y_0} , and $V_{\mathbf{z}_0}$ of x_0 , y_0 and \mathbf{z}_0 respectively, and smooth scalar functions $\eta : V_{\mathbf{z}_0} \to V_{x_0}$, $\alpha : V_{\mathbf{z}_0} \to V_{y_0}$ such that the critical set

$$S_c = \{ (x, y, \mathbf{z}) \in \mathbb{R} \times \mathbb{R} \times \mathbb{R}^m, \ \partial_{x^\ell}^\ell f(x, y, \mathbf{z}) = 0, \forall \ell = 0, \dots, n-1 \}$$
(3.71)

can be parametrized under the classical assumption (3). The set S_c has dimension m + 2 - n. Therefore the functions η and α do not depend on all components of \mathbf{z} . Two cases arise when solving the equation $f(x, y, \mathbf{z}) = 0$ on a small box $\Omega = V_{x_0} \times V_{y_0} \times V_{\mathbf{z}_0}$:

1. if $n = 2\ell$, $\ell \in \mathbb{N}$: let B_{γ} be subset of $V_{y_0} \times V_{z_0}$ defined by

$$B_{\gamma} = \{ (y, \mathbf{z}) \in V_{y_0} \times V_{\mathbf{z}_0}, \, \gamma(y - \alpha(\mathbf{z})) \ge 0 \}.$$
(3.72)

Then there exist a smooth real function ψ such that $\forall \mathbf{z} \in B_{\gamma}$ there are only two branches in Ω :

$$x = \eta(\mathbf{z}) + \psi(\pm(s_{\gamma}(y - \alpha(\mathbf{z})))^{\frac{1}{n}}, \mathbf{z}),$$
(3.73)

2. Otherwise, if $n = 2\ell + 1$, $\ell \in \mathbb{N}$, then there exist an interval *I* containing $\alpha(\mathbf{z})$, and a smooth function ψ defined on $I \times V_{\mathbf{z}_0}$ such that,

$$x = \eta(\mathbf{z}) + \psi((\mathbf{y} - \alpha(\mathbf{z}))^{\frac{1}{n}}, \mathbf{z}).$$
(3.74)

In both cases, ψ is not degenerate at $(0, \mathbf{z}_0)$ with respect to the first variable: $\psi(0, \mathbf{z}_0) = 0$ and $\partial_y \psi(0, \mathbf{z}_0) = 1/\gamma^{\frac{1}{n}} \neq 0$.

The existence of a cube-root or fourth-root singularity is studied in [10]. In the present work, the so-called power-root singularity is defined for any positive integer n.

Definition 3.28 — Power-root singularity. A function *F* defined in a subset of \mathbb{R}^m is said to have an *n*th-root singularity at 0 if there exists n > 0 such that, by a change of variable if needed, *F* can be written as

$$F(\mathbf{X}) = f(X_1^{\frac{1}{n}}, X_2, \dots, X_m)$$
(3.75)

where f is a smooth function.

Using this definition, Theorem 3.27 states the conditions to have a power-root singularity. This singularity is associated to the multiplicity of a root of a function whose definition is recalled below.

Definition 3.29 — Multiplicity. Given a smooth real function f, a positive integer n is said to be the multiplicity of a root r of f, denoted by mult(f)(r) = n, if it satisfies

$$f(r) = \dot{f}(r) = \dots = f^{(n-1)}(r) = 0$$
 and $f^{(n)}(r) \neq 0.$ (3.76)

By convention, we denote $\operatorname{mult}(f)(r) = 0$ when $f(r) \neq 0$. Before stating the main results of this section, recall that $\underline{\mathbf{M}} = \operatorname{diag}(m_j)_{j=1}^{N-1}$ and $\underline{\mathbf{K}} = (k_{ij})_{i,j=1}^{N-1}$ are the matrices obtained from \mathbf{M} and \mathbf{K} by removing their last row and column, with $\mathbf{l}_N = \mathbf{e}_N^{\mathsf{T}} \mathbf{K}$ as the last row of \mathbf{K} and $\underline{\mathbf{l}}_N$ is the last row of \mathbf{K} removing the last entry k_{NN} . Lemma 3.32 requires the following assumptions on the matrix $\underline{\mathbf{D}} = \underline{\mathbf{M}}^{-1} \underline{\mathbf{K}}$ and the matrix \mathbf{P} of eigenvectors of $\mathbf{M}^{-1} \mathbf{K}$ (see Section 3.1).

Assumption 3.30. Assume that rank $(\underline{\mathbf{l}}_N, \underline{\mathbf{l}}_N \underline{\mathbf{D}}, \dots, \underline{\mathbf{l}}_N \underline{\mathbf{D}}^{N-2}) = N - 1$ is maximal.

This assumption states that the vectors $\underline{\mathbf{l}}_N \underline{\mathbf{D}}^i$ are linearly independent and therefore constitute an invertible matrix. Such an assumption is well-known in controllability, for instance, the so-called Kalman's criterion to ensure controllability of a linear system. Another application is refered to Krylov subspace which is used in modern iterative methods for finding one (or a few) eigenvalues of large sparse matrices or solving large systems of linear equations. This assumption is also related to the Frobenius decomposition with the block companion matrix [14]. Assumption 3.31. Assume that all the eigenvalues of **K** are distinct, and none of the last components of the eigenvectors \mathbf{Pe}_k shall vanish, i.e. $P_{Nk} \neq 0$, for all k = 1, ..., N.

This assumption is natural to have N linear grazing orbits [25]. The following lemma shows that for a non-zero solution to a linear differential system to exist, the multiplicity of the last entry of the solution must be bounded and related to the dimension of the solution.

Lemma 3.32. Assume that either Assumption 3.30 or Assumption 3.31 holds. Let $\mathbf{x}(t) \in \mathbb{R}^N$ be a solution to the linear differential system

$$\mathbf{M}\ddot{\mathbf{x}} + \mathbf{K}\mathbf{x} = \mathbf{0}.\tag{3.77}$$

Then, either $mult(x_N)(0) \le 2N - 1$ or $\mathbf{x} \equiv \mathbf{0}$.

In other words, for a non-zero solution to (3.77) to exist, $mult(x_N)(t) < 2N$, for all *t*. We first prove this Lemma by using Assumption 3.30; the proof under Assumption 3.31 is provided after.

Proof. With Assumption 3.30: The statement of Lemma 3.32 is equivalent to saying that if $mult(x_N)(0) > 2N - 1$, that is

$$x_N(0) = \dot{x}_N(0) = \dots = x_N^{(2N-1)}(0) = 0,$$
 (3.78)

then $\mathbf{x} \equiv \mathbf{0}$. From the *N*th equation of (3.77), $m_N \ddot{x}_N + \mathbf{l}_N \mathbf{x} = 0$. Differentiating this equation 2N - 3 times and using (3.78) implies that, for each k = 1, ..., 2N - 3, the following relation holds

$$(A_k): \quad \underline{\mathbf{l}}_N \, \underline{\mathbf{x}}^{(k)}(0) = 0. \tag{3.79}$$

The other N - 1 equations form a reduced system $\underline{\mathbf{M}}\ddot{\mathbf{x}} + \underline{\mathbf{K}}\mathbf{x} = \underline{\mathbf{C}}$ where $\underline{\mathbf{x}} = (x_j)_{j=1}^{N-1}$ and $\underline{\mathbf{C}} = -x_N \underline{\mathbf{I}}_N^{\mathsf{T}}$. Differentiating this system with respect to *t* gives $\underline{\mathbf{M}}\mathbf{x}^{(3)}(0) + \underline{\mathbf{K}}\dot{\mathbf{x}}(0) = \mathbf{0}$, or $\underline{\mathbf{x}}^{(3)}(0) + \underline{\mathbf{D}}\dot{\mathbf{x}}(0) = \mathbf{0}$. Similarly, differentiating this system 2N - 5 times yields

$$(B_k): \underline{\mathbf{x}}^{(k+2)}(0) + \underline{\mathbf{D}}\underline{\mathbf{x}}^{(k)}(0) = \mathbf{0}, \text{ for each } k = 1, \dots, 2N - 5.$$
(3.80)

Since $x_N(0) = \dot{x}_N(0) = 0$, it is sufficient to prove that $\underline{\mathbf{x}}(0) = \mathbf{0}$ and $\underline{\dot{\mathbf{x}}}(0) = \mathbf{0}$. To show the former, a linear system with $\underline{\mathbf{x}}(0)$ as the unknown is constructed as follows. The relation (A_1) from (3.79) gives the first equation of the linear system: $\underline{\mathbf{l}}_N \underline{\mathbf{x}}(0) = 0$. Multiplying (B_1) with $\underline{\mathbf{l}}_N$, and using the relation (A_3) to eliminate the term $\underline{\mathbf{l}}_N \underline{\mathbf{x}}^{(3)}(0)$ results in the second equation: $\underline{\mathbf{l}}_N \underline{\mathbf{Dx}}(0) = 0$. The third equation is obtained after the following steps. First, multiply (B_3) with $\underline{\mathbf{l}}_N$, then use the relation (A_5) to get $\underline{\mathbf{l}}_N \underline{\mathbf{Dx}}^{(3)}(0) = 0$. This vanishing term appears when (B_1) is multiplied by $\underline{\mathbf{l}}_N \underline{\mathbf{D}}$, which gives rise to the third equation: $\underline{\mathbf{l}}_N \underline{\mathbf{D}}^2 \underline{\mathbf{x}}(0) = 0$. The same recursive process can be used for each k = 1, ..., N - 1: the use of the relations of odd indices $(A_{2k-1}, B_{2k-3}, B_{2k-5}, ..., B_1)$ gives the k^{th} equation of the system

$$\underline{\mathbf{l}}_{N}\underline{\mathbf{D}}^{k-1}\underline{\mathbf{x}}(0) = 0.$$
(3.81)

Combining the N - 1 equations (3.81), k = 1, ..., N - 1, it follows that $\underline{\mathbf{x}}(0)$ satisfies $\underline{\mathbf{l}}_N \underline{\mathbf{x}}(0) = 0$, $\underline{\mathbf{l}}_N \underline{\mathbf{D}} \underline{\mathbf{x}}(0) = 0, ..., \underline{\mathbf{l}}_N \underline{\mathbf{D}}^{N-2} \underline{\mathbf{x}}(0) = 0$. By Assumption 3.30, the unique solution of this linear system is $\underline{\mathbf{x}}(0) = \mathbf{0}$. Together with the hypothesis that $x_N(0) = 0$, then $\mathbf{x}(0) = \mathbf{0}$.

Similarly, $\underline{\dot{\mathbf{x}}}(0) = \mathbf{0}$ is shown by constructing another linear system with $\underline{\dot{\mathbf{x}}}(0)$ as the variable. The k^{th} equation $\underline{\mathbf{l}}_N \underline{\mathbf{D}}^{k-1} \underline{\dot{\mathbf{x}}}(0) = 0$, k = 1, ..., N - 1, of the system can be derived by using a similar recursive process now involving the even-indexed relations $(A_{2k}, B_{2k-2}, B_{2k-4}, ..., B_2)$, whence, $\underline{\mathbf{l}}_N \underline{\dot{\mathbf{x}}}(0) = 0$, $\underline{\mathbf{l}}_N \underline{\mathbf{D}} \underline{\dot{\mathbf{x}}}(0) = 0$, \cdots , $\underline{\mathbf{l}}_N \underline{\mathbf{D}}^{N-2} \underline{\dot{\mathbf{x}}}(0) = 0$. It then follows that $\underline{\dot{\mathbf{x}}}(0) = \mathbf{0}$, and together with the assumption $\dot{x}_N(0) = 0$, eventually gives $\dot{\mathbf{x}}(0) = \mathbf{0}$.

It was shown that $\mathbf{x}(0) = \mathbf{0}$ and $\dot{\mathbf{x}}(0) = \mathbf{0}$. A solution of (3.77) associated to this initial data is then identically zero: $\mathbf{x}(t) \equiv \mathbf{0}$. This ends the proof by using Assumption 3.30.

With Assumption 3.31: Consider the N^{th} component of $\mathbf{x}(t)$:

$$x_{N}(t) = \mathbf{e}_{N}^{\top} \mathbf{x}(t) = \mathbf{e}_{N}^{\top} \left(\mathbf{P} \cos(t\mathbf{\Omega}) \mathbf{P}^{-1} \mathbf{x}(0) + \mathbf{P} \mathbf{\Omega}^{-1} \sin(t\mathbf{\Omega}) \mathbf{P}^{-1} \dot{\mathbf{x}}(0) \right),$$
(3.82)
$$= \sum_{k=1}^{N} \left(\alpha_{k} \cos(\omega_{k}t) + \beta_{k} \sin(\omega_{k}t) \right) \mathbf{e}_{N}^{\top} \mathbf{P} \mathbf{e}_{k} = \sum_{k=1}^{N} \left(\alpha_{k} \cos(\omega_{k}t) + \beta_{k} \sin(\omega_{k}t) \right) v_{k},$$

where α_k and β_k are coefficients depending on the initial data $[\mathbf{x}(0), \dot{\mathbf{x}}(0)], v_k = \mathbf{e}_{\mathbf{N}}^{\top} \mathbf{P} \mathbf{e}_k$ are the components of the last row of the matrix \mathbf{P} .

Assume that mult(x_N)(0) > 2N - 1, using (3.78) for x_N and N - 1 first even-order derivatives of x_N yields $\sum_{k=1}^{N} \alpha_k v_k = 0$, $\sum_{k=1}^{N} \alpha_k \omega_k^2 v_k = 0$, ..., $\sum_{k=1}^{N} \alpha_k \omega_k^{2N-2} v_k = 0$. Denote $\lambda_k = \omega_k^2$, this can be rewritten as a linear system where α_k is the unknown:

$$\begin{bmatrix} 1 & 1 & \cdots & 1 \\ \lambda_1 & \lambda_2 & \cdots & \lambda_N \\ & & \ddots & \\ \lambda_1^{N-1} & \lambda_2^{N-1} & \cdots & \lambda_N^{N-1} \end{bmatrix} \begin{bmatrix} v_1 & 0 & \cdots & 0 \\ 0 & v_2 & \cdots & 0 \\ & & \ddots & \\ 0 & 0 & \cdots & v_N \end{bmatrix} \begin{bmatrix} \alpha_1 \\ \alpha_2 \\ \vdots \\ \alpha_N \end{bmatrix} = \begin{bmatrix} 0 \\ 0 \\ \vdots \\ 0 \end{bmatrix}.$$
 (3.83)

By Assumption 3.31, it follows that λ_k , k = 1, ..., N are distinct, and $v_k = P_{Nk} \neq 0$ for all k = 1, ..., N, then there is a unique solution $\alpha_k = 0$, for all k = 1, ..., N.

Similarly, using (3.78) for N first odd-order derivatives of x_N yields Similarly, using (3.78)

for N first odd-order derivatives of x_N yields

$$\sum_{k=1}^{N} \beta_k \omega_k v_k = 0, \quad \sum_{k=1}^{N} \beta_k \omega_k^3 v_k = 0, \quad \dots, \quad \sum_{k=1}^{N} \beta_k \omega_k^{2N-1} v_k = 0.$$
(3.84)

This can be rewritten as a linear system where the β_k are the unknowns:

$$\begin{bmatrix} 1 & 1 & \dots & 1 \\ \lambda_1 & \lambda_2 & \dots & \lambda_N \\ & \ddots & \\ \lambda_1^{N-1} & \lambda_2^{N-1} & \dots & \lambda_N^{N-1} \end{bmatrix} \begin{bmatrix} \omega_1 v_1 & 0 & \dots & 0 \\ 0 & \omega_2 v_2 & \dots & 0 \\ & \ddots & & \\ 0 & 0 & \dots & \omega_N v_N \end{bmatrix} \begin{bmatrix} \beta_1 \\ \beta_2 \\ \vdots \\ \beta_N \end{bmatrix} = \begin{bmatrix} 0 \\ 0 \\ \vdots \\ 0 \end{bmatrix}.$$
 (3.85)

With the same argument, the unique solution of this sytem is $\beta_k = 0$, for all k = 1, ..., N. As a consequence, all the components of $\mathbf{x}(t)$ are zero, hence, $\mathbf{x}(t) \equiv \mathbf{0}$ for all *t*. Lemma 3.32 is then proven.

Remark 3.33. The main idea of the proof using Assumption 3.31 is to show that the set of functions $\{\cos(\omega_k t), \sin(\omega_k t)\}_{k=1,...,N}$ is linearly independent. This is a very classical proof in linear algebra.

This result can be proven to be true for all component x_k of \mathbf{x} , k = 1, ..., N with suitable Assumption 3.30 or 3.31. Let us apply this general result to the solution to (3.1) in order to show that at the contact time, u_N has at most 2N - 1 derivatives which vanish. Between the successive closing contacts, the system is linear and hence the solution $\mathbf{u}(t)$ is analytic. Therefore, for the sake of simplicity, assume that all the derivatives are taken on the left of 0, $u_N^{(n)-}$ is denoted for the left n^{th} -derivative of u_N at 0 with respect to t.

Proposition 3.34. Assume that $\mathbf{u}(t)$ is a solution to (3.1) which has a closing contact at t = 0, i.e. $u_N(0) = d$. Then, under Assumption 3.30 or 3.31,

$$0 \le \text{mult}(\dot{u}_N)(0) \le 2N - 1. \tag{3.86}$$

Moreover, if there is a sticking phase after t = 0 of duration τ and with one similar assumption as Assumption 3.30 or 3.31 for the $(N - 1) \times (N - 1)$ sticking system rewritten in a suitable basis where $\underline{\mathbf{l}}_N \underline{\mathbf{u}}$ corresponds to the last entry of the solution, then

$$0 \le \operatorname{mult}(\underline{\mathbf{l}}_N \underline{\dot{\mathbf{u}}})(\tau) \le 2N - 3. \tag{3.87}$$

Proof. The first part of the proposition is obtained by applying Lemma 3.32 to $\mathbf{x} = \dot{\mathbf{u}}^-$. Notice that the system of motion is linear outside the successive closing contacts, i.e.

$$\mathbf{M}\ddot{\mathbf{u}}^{-} + \mathbf{K}\mathbf{u} = \mathbf{0}. \tag{3.88}$$

Hence, $\dot{\mathbf{u}}$ is the solution to $\mathbf{M}\ddot{\mathbf{u}}^- + \mathbf{K}\dot{\mathbf{u}} = \mathbf{0}$. By Lemma 3.32, it follows that $\operatorname{mult}(\dot{u}_N^-)(0) \le 2N - 1$ or $\dot{\mathbf{u}}(t) \equiv \mathbf{0}$. However, if $\dot{\mathbf{u}}(t) \equiv \mathbf{0}$, then substituting this into the linear system, it follows that $\mathbf{u}(t) \equiv \mathbf{0}$ which is in contradiction with the hypothesis that $u_N(0) = d > 0$.

If there is a sticking phase at t = 0, then $\dot{u}_N(0) = 0$ and $\dot{u}_N(\tau) = 0$ where τ is the end of sticking phase. During the sticking phase from t = 0 to $t = \tau$, the Nth mass lies on the obstacle,

$$m_N \ddot{u}_N(t) + \mathbf{l}_N \mathbf{u}(t) = R(t), \tag{3.89}$$

where $R(t) \le 0$, and $F(t) = \mathbf{l}_N \mathbf{u}(t)$. The duration of the sticking phase τ is also the end of the sticking phase, it is implicitly presented by

$$F(\tau) = \mathbf{l}_N \mathbf{u}(\tau) = \underline{\mathbf{l}}_N \underline{\mathbf{u}}(\tau) + k_{NN} d = 0,$$
(3.90)

$$F(\tau + \delta) > 0, \quad \forall \delta \in]0; \delta_0[, \quad \delta_0 > 0, \tag{3.91}$$

where $\underline{\mathbf{u}}$ is the solution of the sticking system $\underline{\mathbf{M}}\mathbf{\ddot{u}} + \underline{\mathbf{K}}\mathbf{u} = \underline{\mathbf{C}}$, with $\underline{\mathbf{C}} = -d \underline{\mathbf{l}}_N^{\top}$. Differentiating this system with respect to *t* yields $\underline{\mathbf{M}}\mathbf{\ddot{u}} + \underline{\mathbf{K}}\mathbf{\dot{u}} = \mathbf{0}$. A change of variables $\underline{\mathbf{\dot{u}}} = \mathbf{Q}\mathbf{v}$ using an $\underline{\mathbf{M}}$ -orthogonal matrix \mathbf{Q} , i.e. $\mathbf{Q}^{\top}\underline{\mathbf{M}}\mathbf{Q} = \mathbf{I}$, such that $\underline{v}_{N-1} = c \underline{\mathbf{l}}_N \underline{\dot{\mathbf{u}}}$, with $c \neq 0$ yields

$$\mathbf{\underline{\ddot{v}}} + \mathbf{Q}^{\mathsf{T}} \mathbf{\underline{K}} \mathbf{Q} \mathbf{\underline{v}} = \mathbf{0}. \tag{3.92}$$

Invoking Theorem 3.32 for this $(N-1) \times (N-1)$ reduced system, it follows that $\operatorname{mult}(\underline{\mathbf{l}}_N \underline{\dot{\mathbf{u}}})(\tau) \le 2N-3$.

In the following proposition, the singularity of the duration of the free-flight and of the duration of sticking phase is shown. These durations have a relationship with the first return time. More precisely, if $W \in \mathcal{H}^- \cup \mathcal{H}^0_{\mathcal{G}}$, then the first return time of an orbit generating from W coincides with the duration of the free-flight s(W). Otherwise, if $W \in \mathcal{H}^0_S$, then the first return time involves the duration of sticking phase and the duration of the free-flight: $T(W) = \tau(W) + s(W)$. The bounded multiplicity shown in Proposition 3.34 is the main ingredient to prove the singularity.

Proposition 3.35 — Power-root singularity. Suppose that $W_0 \in \mathcal{H}_{\mathcal{P}}$ generates an orbit $[\mathbf{u}(t), \dot{\mathbf{u}}(t)]$ of (3.1) which has a closing contact at $t = T_0$. Assumption 3.30 or 3.31 and a similar assumption for the sticking system hold.

1. If $W_0 \in \mathcal{H}^-$ and $W(T_0) \in \mathcal{H}^0$, then the duration of the free-flight of the orbit generated from a nearby W until the next closing contact, denoted by s(W), has an n^{th} -root singularity at W_0 with

$$n = 1 + \operatorname{mult}(\dot{u}_N)(T_0), \quad 1 \le n \le 2N.$$
 (3.93)

2. If $W_0 \in \mathring{\mathcal{H}}^0_{\mathcal{S}}$, the interior of the set $\mathcal{H}^0_{\mathcal{S}}$ in the topology of \mathcal{H}^0 , i.e. there is a sticking phase

right after t = 0 with the duration $\tau_0 = \tau(W_0)$, and then for all $W \in \mathcal{H}^0_S$, W near W_0 , the duration of the sticking phase $\tau = \tau(W)$ has an m^{th} -root singularity at W_0 with

$$m = 1 + \text{mult}(l_N \dot{\mathbf{u}})(\tau(\mathbf{W}_0)), \quad 1 \le m \le 2N - 2.$$
 (3.94)

Remark 3.36. The first return time can involve two power-root singularities: one from the duration of the sticking phase, and one from the duration of the free-flight phase.

Remark 3.37. When $W_0 \in \mathcal{H}^0_{\mathcal{G}} \cap \overline{\mathcal{H}}^0_{\mathcal{S}}$, the behaviour of the first return time is more complicated. It depends whether W belongs to $\mathcal{H}^0_{\mathcal{S}}$ or not. In the former, there are two power-root singularities while in the latter, there is a single power-root singularity.

Proof. The idea of the proof is to use the implicit function theorem for the duration of the free-flight or the duration of the sticking phase.

- 1. Consider the duration T_0 of the free-flight between the two closing contacts at t = 0 and $t = T_0$. For (s, W) in the neighborhood of (T_0, W_0) , consider the function $f(s, W) = \mathbf{e}_N^\top \mathbf{R}(s)\mathbf{S}W$. It is smooth and defined for all arguments and corresponds to the nonsmooth function (due to the closing contacts) $u_N(s, W)$. A main point is to apply the implicit function theorem to f and then to interpret the result for the nonsmooth function u_N . By Proposition 3.34, it follows that $0 \le \text{mult}(\dot{u}_N)(T_0) \le 2N 1$. Using Remark 3.27 for the function f in the neighborhood of (T_0, W_0) , it follows that there exists a subset containing W_0 such that s has an n^{th} -root singularity at W_0 with $n = 1 + \text{mult}(\dot{u}_N)(T_0)$, and thus, $1 \le n \le 2N$.
- 2. Recall that $\mathbf{U}(t, \mathbf{W}) = [\mathbf{u}(t, \mathbf{W}), \dot{\mathbf{u}}(t, \mathbf{W})]$ is the solution to (3.1) associated to the initial data W. During the sticking phase, $u_N(t, W) = d$ and $\dot{u}_N(t, W) = 0$ for all $t \in [0; \tau]$. By denoting $F(t, W) = \mathbf{l}_N \mathbf{u}(t, W) = \mathbf{l}_N \mathbf{u}(t, W) + k_{NN}d$, the last equation of (3.1) yields $m_N \ddot{u}_N(t, W) =$ -F(t, W) + R(t, W), where the reaction from the obstacle is $R(t, W) \le 0, \forall t \in [0; \tau]$. Hence, the sticking phase starts at t = 0 when F(0, W) = 0 and $F(\delta, W) < 0, \forall \delta \in [0; \delta_0[$, $\delta_0 > 0$. The second condition ensures that the total force acting on the Nth mass is strictly positive right after t = 0 and therefore the mass stays on the obstacle. The sticking phase holds as long as $F(t, W) \leq 0$, it ends at $t = \tau$ when $F(\tau, W) = 0$ and $F(\tau + \delta, W) > 0$, $\forall \delta \in [0; \delta_0[, \delta_0 > 0]$. The latter condition makes sure that the total force acting on the Nth mass becomes strictly negative right after $t = \tau$ and therefore the mass leaves the obstacle. Similarly, denote by $F^*(t) = \mathbf{l}_N \mathbf{u}(t) + k_{NN}d$, where \mathbf{u} is the solution of the smooth system <u>Mü</u> + <u>Ku</u> = <u>C</u>, with <u>C</u> = $-d\underline{I}_N^{\top}$. By using Remark 3.27 for the smooth function F^* in the neighborhood of (τ_0, W_0) , it follows that there is a subset of \mathcal{H}^0_S containing W_0 such that τ has an mth-root singularity at W₀ with $m = 1 + \text{mult}(\dot{F})(\tau(W_0))$. Proposition 3.34 implies that $0 \leq \text{mult}(\mathbf{I}_N \dot{\mathbf{u}})(\tau(\mathbf{W}_0)) \leq 2N - 3$, therefore $1 \leq m \leq 2N - 2$.

Remark 3.38. The lower bound of this multiplicity is optimal for a chain, i.e. 1/2N is the greatest lower bound of the power-root singularity of the duration *s*. Let us verify this by

showing that, if $u_N(T_0) = d$, $\dot{u}_N(T_0) = \ldots = u_N^{(2N-1)}(T_0) = 0$ then there is a unique data of (3.1) corresponding to the solution with the maximal 2*N*-root singularity of *s*.

Notice that in this case, the symmetric matrix **K** has the form

with $k_{i+1,i} \neq 0$ for all i = 1, ..., N - 1. Outside the closing contacts, **u** is the solution of the linear system

$$\mathbf{M}\ddot{\mathbf{u}} + \mathbf{K}\mathbf{u} = \mathbf{0}.\tag{3.96}$$

From the last equation of (3.96): $m_N \ddot{u}_N + k_{N,N-1} u_{N-1} + k_{NN} u_N = 0$, together with $k_{N,N-1} \neq 0$, it follows that

$$u_{N-1}(0) = C_{N-1}d$$
 and $\dot{u}_{N-1}(0) = \dots = u_{N-1}^{(2N-3)}(0) = 0$, $C_{N-1} = -k_{NN}/k_{N,N-1}$. (3.97)

Similarly, the recursive process can be used until the (k + 1)th equation, k = 2, ..., N - 1, which together with the assumption $k_{k,k-1} \neq 0$ give

$$u_k(0) = C_k d \text{ and } \dot{u}_k(0) = \dots = u_k^{2k-1}(0) = 0,$$
 (3.98)

where $C_k = -(k_{kk}C_k + k_{k+1,k}C_{k+1})/k_{k,k-1}$. As a consequence, an initial data is obtained:

$$\mathbf{u}(0) = [C_1 d, \dots, C_{N-1} d, d], \ \dot{\mathbf{u}}(0) = \mathbf{0}.$$
(3.99)

Hence, there exists a unique solution to (3.1) associated to this initial data.

However, it is not sure that 1/(2N - 2) is the greatest lower bound of the singularity of τ since it depends on the admissibility condition of the end of the sticking phase which is $\underline{\mathbf{l}}_{N}\underline{\mathbf{u}}^{(2N-2)}(\tau) > 0$.

Remark 3.39 — Power root singularity for the 2-dof chain. Let us clarify the power-root singularity when N = 2. In this case, the duration of the free-flight may have a square-root, cube-root, or at most fourth-root singularity, while the duration of the sticking phase is analytic.

An illustration of these power-root singularities is given in Figure 3.3.

Let $W_0 = [u_1(0), d, \dot{u}_1(0), \dot{u}_2(0)] \in \mathcal{H}_{\mathcal{P}}$ be the state of the system at t = 0. The interesting criteria happens when the first contact $(u_2(T_0) = d)$ is with zero velocity, $\dot{u}_2(T_0) = 0$. That means $W(T_0) \in \mathcal{H}^0$.



FIGURE 3.3 – Power-root singularity in the plane (u_1, \dot{u}_1) which is isomorphic to the set \mathcal{H}^0 since $u_2 = d$ and $\dot{u}_2 = 0$. The left part, $u_1 < d$ or $u_1 = d$, $\dot{u}_1 = 0$, corresponds to the grazing contact. The orange branch $u_1 = d$ and $\dot{u}_1 > 0$ corresponds to the beginning of the sticking phase. The other parts do not coincide with closing contacts. The gray part $u_1 > d$ corresponds to a state during the sticking phase. The dashed line $u_1 = d$ and $\dot{u}_1 < 0$ corresponds to the end of a sticking phase.

If $W_0 \in \mathcal{H}^- \cup \mathcal{H}^0_{\mathcal{G}}$, i.e there is no sticking phase after t = 0, three possibilities are considered: 1. $u_1(T_0) < d$: This gives $m_2 \ddot{u}_2(T_0) = k_2(u_1(T_0) - u_2(T_0)) < 0$. The duration of the

- 1. $u_1(I_0) < a$: This gives $m_2 u_2(I_0) = k_2(u_1(I_0) u_2(I_0)) < 0$. The duration of the free-flight s = s(W) has a square-root singularity at W₀.
- 2. $u_1(T_0) = d$ and $\dot{u}_1(T_0) > 0$: this is the beginning of a sticking phase, $\ddot{u}_2(T_0) = 0$ and $m_2\ddot{u}_2^-(T_0) = k_2(\dot{u}_1(T_0) \dot{u}_2(T_0)) > 0$. Hence, s = s(W) has a cube-root singularity at W₀.
- 3. $u_1(T_0) = d$ and $\dot{u}_1(T_0) = 0$: this gives $\ddot{u}_2(T_0) = \ddot{u}_2(T_0) = 0$. However, $m_2 u_2^{(4)-}(T_0) = k_2(\ddot{u}_1(T_0) \ddot{u}_2(T_0)) = k_1 k_2 d > 0$. This case corresponds to a grazing contact and the fourth-root singularity of s(W) arises.

If $W_0 \in \mathcal{H}_S^0$, i.e. there is a sticking phase of duration τ_0 at t = 0 and there are two possibilities: $u_1(0) > d$ or $(u_1(0) = d$ and $\dot{u}_1(0) > 0)$. During a sticking phase, u_1 is a solution of $m_1\ddot{u}_1 + (k_1 + k_2)u_1 = k_2d$. The end of the sticking phase is at $t = \tau_0$, when $u_1(\tau_0) = d$ and $\dot{u}_1(\tau_0) < 0$. This shows that $\operatorname{mult}(\dot{u}_1)(\tau_0) \le 2N - 3 = 1$. This is a particular case where the duration of the sticking phase $\tau(W)$ is analytic for $W \in \mathcal{H}_S^0$.

An example of a cubic-root singularity is depicted in Figure 3.4.

3.4.3 Discontinuous first return time

A consequence of Theorem 3.8 is the discontinuity of the return time near many grazing orbits. The simplest case is stated in the next corollary. This result may gives many consequences: possible



FIGURE 3.4 – First return time $T(v_1)$ with respect to \dot{u}_1 (near a periodic solution with one sticking phase per period [28]). A cubic-root singularity appears near $\dot{u}_1(0) = 5.86$.

discontinuous first return map, more important singularity than the square-root singularity, immediate instability, complex patterns... These consequences are postponed to forthcoming works.

Corollary 3.40 — **Discontinuous first return time.** If $W_0 \in \mathcal{H}^-$ and $W(T_0) \in \mathcal{H}^0$ where $T_0 = T(W_0) > 0$ and $W(T_0)$ satisfies Assumption 3.7 which says that the grazing contact is not degenerate, then the first return time T(W) is discontinuous at W_0 .

Proof. Theorem 3.8 states that T(W) is near $T_0 = T(W_0)$ only on at most a half neighborhood of W_0 : $\mathcal{B}_k = \{W \in V_{W_0}, s_k(W_k - \alpha(\underline{W})) \ge 0\}$. Thus, T(W) is not near T_0 for $s_k(W_k - \alpha(\underline{W})) < 0$. \Box

3.5 The square-root instability

In this section, the so-called square-root instability is introduced for a fixed-point of the map. Loosely speaking, the appearance of the square-root singularity may affect the dynamics and the fixed-point may become unstable.

Let $\mathbf{F} : \mathbb{R}^n \to \mathbb{R}^n$ be a map defined by $\mathbf{F}(\mathbf{X}) = \mathbf{G}(\sqrt{|x_1|}, x_2, ..., x_n)$ where $\mathbf{G} : \mathbb{R}^n \to \mathbb{R}^n$ is at least a C^2 function. **F** has a fixed-point $\mathbf{0} \in \mathbb{R}^n$. Consider the dynamical system obtained by iterating **F**:

$$\mathbf{X}^{m+1} = \mathbf{F}(\mathbf{X}^m), \quad \mathbf{X}^m = (x_1, x_2, \dots, x_n)^m \in \mathbb{R}^n, \quad m = 1, 2, \dots$$
 (3.100)

With the presence of the square-root term $\sqrt{x_1}$, a question on the stability of the fixed-point **0** then arises. This section provides a generic condition for which the fixed-point **0** of the map **F** is unstable. Besides, counterexamples show the dynamical complexity of the map in the vicinity of its fixed-point even in small dimensions.

3.5.1 A nonlinear *n* dimensional map

In the following theorem, it is shown under a specific condition that the square-root term acting on the component x_1 of **X** yields the instability of the fixed-point **0**. Moreover, this instability occurs along the direction of x_1 .

Theorem 3.41 — Unstable fixed-point. Suppose $\mathbf{F}, \mathbf{G} : \mathbb{R}^n \to \mathbb{R}^n$ are functions such that $\mathbf{F}(\mathbf{X}) = \mathbf{G}(\underline{\mathbf{X}})$, where $\mathbf{X} = (x_1, \dots, x_n)$ and $\underline{\mathbf{X}} = (\sqrt{|x_1|}, x_2, \dots, x_n)$. If the function \mathbf{G} belongs to C^2 , $\mathbf{G}(\mathbf{0}) = \mathbf{0}$, and the Jacobian $\mathbf{DG}(\mathbf{0}) = (a_{ij})_{i,j=1}^n$ satisfies

$$a_{11} \neq 0 \tag{3.101}$$

then **0** is an unstable fixed-point of **F**.

Remark 3.42. More generally, if the square-root is activated on the component x_k , then the instability of the fixed-point **0** can be obtained by evaluating the value of $a_{kk} = \partial_{x_k} g_k(\mathbf{0})$.

Proof. Denote $\mathbf{G}(\mathbf{X})$ by $(g_1, g_2, \dots, g_n)(\mathbf{X}) \in \mathbb{R}^n$. Our goal is to show that there is a neighborhood of **0** such that many points arbitrarily close to **0** will go out of that neighborhood.

The Taylor expansion with an integral remainder of each g_i near $\mathbf{0}, i = 1, ..., n$ is

$$g_{i}(\underline{\mathbf{X}}) = a_{i1}\sqrt{|x_{1}|} + \sum_{j=2}^{n} a_{ij}x_{j} + |x_{1}|r_{i,11}(\underline{\mathbf{X}}) + 2\sum_{j=2}^{n} \sqrt{|x_{1}|}x_{j}r_{i,1j}(\underline{\mathbf{X}}) + 2\sum_{n\geq k>l>1}^{n} x_{k}x_{l}r_{i,kl}(\underline{\mathbf{X}}) + \sum_{j=2}^{n} x_{j}^{2}r_{i,jj}(\underline{\mathbf{X}}),$$
(3.102)

where $r_{i,k\ell}(\underline{\mathbf{X}}) = \int_0^1 (1-s)\partial_{k\ell}^2 g_i(s\underline{\mathbf{X}}) ds$, for all $k, \ell = 1, ..., n$. On a compact set which will be chosen later, there exists a constant M > 0 such that $|r_{i,k\ell}(\mathbf{X})| \le M$ for all $k, \ell = 1, ..., n$ and for each i = 1, ..., n. Hence,

$$|a_{i1}|\sqrt{|x_1|} - \sum_{j=2}^n |a_{ij}||x_j| - M\,\theta(\mathbf{X}) \le |g_i(\underline{\mathbf{X}})| \le |a_{i1}|\sqrt{|x_1|} + \sum_{j=2}^n |a_{ij}||x_j| + M\,\theta(\mathbf{X}),$$
(3.103)

where $\theta(\mathbf{X}) = |x_1| + 2\sqrt{|x_1|} \sum_{j=2}^n |x_j| + 2\sum_{n \ge k > \ell > 1} |x_k| |x_\ell| + \sum_{j=2}^n x_j^2$. A suitable neighborhood, denoted by D_{ϵ} , is constructed so that a sequence starting from any point in D_{ϵ} will eventually go away from **0** in the direction of x_1 . To define D_{ϵ} , the following notations are needed. Let α_i , i = 2, ..., n, be:

$$\alpha_{i} = \begin{cases} \frac{|a_{11}|}{2n |a_{1i}|} & \text{if } a_{1i} \neq 0, \\ 1 & \text{if } a_{1i} = 0. \end{cases}$$
(3.104a)
(3.104b)

Through this definition, it follows that $\alpha_j |a_{1j}| \leq |a_{11}|/2n$, the equality occurs if $a_{1j} \neq 0$, otherwise it is a strict inequality. For $\mathbf{X} \in \mathbb{R}^n$ such that $|x_j| \leq \alpha_j \sqrt{|x_1|}$, $\forall j = 2, ..., n$, we have $|a_{1j}||x_j| \leq |a_{1j}|\alpha_j \sqrt{|x_1|}$, and hence,

$$\sum_{j=2}^{n} |a_{1j}| |x_j| \le \sum_{j=2}^{n} |a_{1j}| \alpha_j \sqrt{|x_1|} \le (n-1) \frac{|a_{11}|}{2n} \sqrt{|x_1|}.$$
(3.105)

Moreover, the following inequality holds

$$\theta(\mathbf{X}) \le |x_1| \left(1 + 2\sum_{j=2}^n \alpha_j + 2\sum_{n \ge k > l > 1} \alpha_k \alpha_l + \sum_{j=2}^n \alpha_j^2 \right).$$
(3.106)

Denote $\alpha = 1+2\sum_{j=2}^{n} \alpha_j + 2\sum_{n\geq k>l>1} \alpha_k \alpha_l + \sum_{j=2}^{n} \alpha_j^2$, the above inequality becomes $\theta(\mathbf{X}) \leq \alpha |x_1|$. Denote also $\gamma_0 = a_{11}^2/(2n\alpha M)^2$. Then, for $\mathbf{X} \in \mathbb{R}^n$ such that $|x_1| \leq \gamma_0$, we have

$$M\alpha|x_1| \le \frac{|a_{11}|}{2n}\sqrt{|x_1|}.$$
(3.107)

Since $M\theta(\mathbf{X}) \leq M\alpha |x_1|$, it follows that

$$M\theta(\mathbf{X}) \le \frac{|a_{11}|}{2n} \sqrt{|x_1|}.$$
(3.108)

Let C > 0 be a constant such that

$$C \ge \frac{2}{|a_{11}|} \left(|a_{i1}| + \sum_{j=2}^{n} |a_{ij}| \alpha_j + M\alpha \right), \, \forall i = 2, \dots, n.$$
(3.109)

Let $\gamma > 0$ such that $\gamma = \min_{i=2,...,n} \{\gamma_i > 0 \mid C\gamma_i \le \alpha_i \sqrt{\gamma_i}\}$. Therefore, for any $\mathbf{X} \in \mathbb{R}^n$ such that $|x_1| \le \gamma$, we have $C|x_1| \le \alpha_i \sqrt{|x_1|}$, for all i = 2, ..., n. We now can define D_{ϵ} as follows. Let $\epsilon = \min\{\gamma_0, \gamma, a_{11}^2/8\} > 0$. Consider

$$D = \{ \mathbf{X} = (x_1, \dots, x_n) \in \mathbb{R}^n, |x_i| \le C |x_1|, \forall i = 2, \dots, n \},$$
(3.110)

$$D_{\epsilon} = \{ \mathbf{X} = (x_1, \dots, x_n) \in \mathbb{R}^n, |x_1| \le \epsilon \text{ and } |x_i| \le C |x_1|, \forall i = 2, \dots, n \}.$$
 (3.111)

This choice of D_{ϵ} avoids the criteria in which the instability of **0** is hidden by starting at a point near **0** but the sequence comes back at **0** after one step. See Example 3.44 for more information. An illustration of D_{ϵ} is provided in Figure 3.5. Any $\mathbf{X} \in D_{\epsilon}$ satisfies the important inequalities as stated next.



FIGURE 3.5 – Neighborhood D_{ϵ}

Lemma 3.43. For any $\mathbf{X} \in D_{\epsilon}$, the following inequalities hold:

$$\frac{|a_{11}|}{2}\sqrt{|x_1|} \le |g_1(\underline{\mathbf{X}})| \le \frac{3|a_{11}|}{2}\sqrt{|x_1|},\tag{3.112}$$

$$|g_i(\underline{\mathbf{X}})| \le C_i |g_1(\underline{\mathbf{X}})|, \ \forall i = 2, \dots, n.$$
(3.113)

Proof. Suppose that $\mathbf{X} \in D_{\epsilon}$ then $|x_j| \leq C|x_1|$ for all j = 2, ..., n, and $|x_1| \leq \epsilon \leq \gamma$. By the definition of γ , it follows that $C|x_1| \leq \alpha_j \sqrt{|x_1|}$. Hence, $|x_j| \leq C|x_1| \leq \alpha_j \sqrt{|x_1|}$. Thus (3.105) holds. Substituting (3.105) and (3.108) into (3.103) when i = 1, the right hand side (RHS) of (3.103) becomes

$$\operatorname{RHS}(3.103) \le |a_{11}|\sqrt{|x_1|} + (n-1)\frac{|a_{11}|}{2n}\sqrt{|x_1|} + \frac{|a_{11}|}{2n}\sqrt{|x_1|} \le \frac{3|a_{11}|}{2}\sqrt{|x_1|}.$$
 (3.114)

Similarly,

LHS(3.103)
$$\ge |a_{11}|\sqrt{|x_1|} - (n-1)\frac{|a_{11}|}{2n}\sqrt{|x_1|} - \frac{|a_{11}|}{2n}\sqrt{|x_1|} \ge \frac{|a_{11}|}{2}\sqrt{|x_1|}.$$
 (3.115)

Therefore, **X** satisfies (3.112).

Let us prove inequality (3.113). For i = 2, ..., n, expressions (3.103) and (3.108) imply

$$|g_i(\underline{\mathbf{X}})| \le \left(|a_{i1}| + \sum_{j=2}^n |a_{ij}|\alpha_j + M\alpha\right) \sqrt{|x_1|}.$$
(3.116)

Via inequality (3.112), it is shown that $|a_{11}|\sqrt{|x_1|} \le 2|g_1(\tilde{\mathbf{X}})|$. Hence, to prove (3.113), it is sufficient to show that

$$\frac{2}{|a_{11}|} \left(|a_{i1}| + \sum_{j=2}^{n} |a_{ij}| \alpha_j + M\alpha \right) \le C.$$
(3.117)

This is true by the choice of C given in (3.109). Inequality (3.113) is then proven.

Back to the proof of Theorem 3.41, the idea is to show that the recurrence goes away from $\mathbf{0}$ in the direction of the first component. In other words, the square-root singularity acting on the first component plays an important role via the inequalities (3.112) and (3.113).

Let us show that, for any $0 < \delta < \epsilon$, if the sequence $(\mathbf{X}^m)_{m \ge 1}$ in D_{ϵ} , where $\mathbf{X}^m = (x_1^m, \dots, x_n^m)$, is defined by

$$\mathbf{X}^{0} = (\delta/2, 0, \dots, 0) \in D_{\epsilon}, \quad \mathbf{X}^{m+1} = \mathbf{F}(\mathbf{X}^{m}), \quad m \ge 1,$$
(3.118)

then there exists a $N_0 > 0$ such that $\mathbf{X}^{N_0} \notin D_{\epsilon}$. From (3.113), $\mathbf{X}^1 = \mathbf{F}(\mathbf{X}^0) = \mathbf{G}(\underline{\mathbf{X}}^0) \in D$. If $|x_1^1| > \epsilon$ then $\mathbf{X}^1 \notin D_{\epsilon}$. This shows the instability of **0**. Otherwise, if $|x_1^1| \leq \epsilon$ and $|x_i^1| = |g_i(\mathbf{X}^0)| \leq C|x_1^1|$ by inequality (3.113), then $\mathbf{X}^1 \in D_{\epsilon}$, and \mathbf{X}^2 is considered. If there exists $\mathbf{X}^m \in D_{\epsilon}$ for all $m \geq 1$ then

$$\frac{a_{11}}{2}\sqrt{|x_1^m|} \le |x_1^{m+1}| \le \frac{3a_{11}}{2}\sqrt{|x_1^m|}.$$
(3.119)

Consider the sequence z_m defined by $z_{m+1} = a_{11}/2\sqrt{|z_m|}$, $z_0 = x_1^0 = \delta/2 \le \epsilon \le a_{11}^2/8$. Then, inequality (3.119) yields $|x_1^m| \ge z_m$. It is known that z_m increasingly converges to $a_{11}^2/4$ in an interval $(0, a_{11}^2/4)$. Hence, there exists $N_0 > 0$ such that $|z_{N_0}| > a_{11}^2/8 \ge \epsilon$, and thus $|x_1^{N_0}| \ge |z_{N_0}| > \epsilon$. That means, $\mathbf{X}^{N_0} \notin D_{\epsilon}$ and hence **0** is unstable.

Example 3.44. Consider $\mathbf{F}(x, y) = \mathbf{G}(\sqrt{|x|}, y)$ when **G** is a linear map

$$\mathbf{G}(\mathbf{X}) = \mathbf{G}(x, y) = \mathbf{A}\mathbf{X} = \begin{bmatrix} a & b \\ c & d \end{bmatrix} \begin{bmatrix} x \\ y \end{bmatrix}, \ a \neq 0.$$
(3.120)

Figure 3.6(a) shows that the fixed-point (0,0) of **F** is unstable and the recurrence goes away from this point along the line y = cx/a. An interesting case is when $c = \alpha a$, $d = \alpha b$ with $\alpha \neq 0$. The instability of (0,0) is hidden if the sequence starts at a point belonging to the curve (*C*) : $y = -a\sqrt{|x|}/b$ because the sequence stays at (0,0) as soon as the second step (see Figure 3.6(b)). To show the instability, it is required to start at a point (x_0, y_0) which does not lie on *C*. This is the reason why the set D_{ϵ} is chosen as specified in the previous proof.

3.5.2 Two-dimensional maps with critical instability

As proven in Theorem 3.41, the square-root instability of the fixed-point appears when $a_{11} \neq 0$. Otherwise, when a_{11} vanishes, the square-root singularity may appear under additional assumptions on the other components of **A**.



FIGURE 3.6 – Instability of the fixed-point (0, 0): (a) when a = 1, c = 2, and b = d = 0, the recurrence goes away from (0, 0) along the line y = cx/a; (b) when $c = \alpha a$ and $d = \alpha b$, the instability is hidden if starting at a point on the curve $C : y = -a\sqrt{|x|}/b$; (c) when b = c = 1 and a = d = 0. The gradient color scale [______] shows initial iterates in blue to final iterates in red irrespective of the magnitude.

Proposition 3.45. Suppose $\mathbf{F}, \mathbf{G} : \mathbb{R}^2 \to \mathbb{R}^2$ such that $\mathbf{F}(x, y) = \mathbf{G}(\sqrt{|x|}, y)$ where \mathbf{G} is a linear map satisfying $\mathbf{G}(x, y) = (by, cx + dy)$ with $bc \neq 0$. Then, (0, 0) is an unstable fixed-point of \mathbf{F} .



FIGURE 3.7 – Instability of the fixed-point (0,0): (a) when 0 < d = 0.5 < 1 and c = 1; (b) when -1 < d = -0.5 < 0 and c = 1; (c) when d = 0 and c = 1. The gradient color scale shows initial iterates in blue to final iterates in red irrespective of the magnitude.

Proof. By considering the map $\mathbf{F}^2(x, y) = \mathbf{F}(\mathbf{F}(x, y)) = (bc\sqrt{|x|} + bdy, cd\sqrt{|x|} + d^2y + c\sqrt{|by|})$, it is seen that $a_{11} = bc \neq 0$, then the same technique as in the proof of Theorem 3.41 is used to show the instability of the fixed-point (0,0) of \mathbf{F}^2 .

Denote

$$\mathbf{DG}(x, y) = \begin{bmatrix} 0 & b \\ c & d \end{bmatrix},\tag{3.121}$$

Then, in the case of a linear map **G**, the fixed-point (0, 0) of **F** is stable if and only if bc = 0 and |d| < 1. The square-root term in *x* disappears by the condition bc = 0, the stability of the 2-dimensional map becomes the one of 1-dimensional linear map, and hence |d| < 1 is needed to have the stability.

Example 3.46. A simple example of this instability is shown in Fig. 3.6(c) where b = c = 1 and a = d = 0. It is seen that, starting at a point closed to (0, 0), the recurrence goes away from this point.

Example 3.47. There are also the cases where (0, 0) is stable for the linear map **G** but unstable for the nonlinear map **G**. Suppose $\mathbf{F}(x, y) = \mathbf{A}X = (0, \sqrt{|x|} + dy)$ with

$$\mathbf{A} = \begin{bmatrix} 0 & 0\\ 1 & d \end{bmatrix},\tag{3.122}$$

where |d| < 1. Thus, (0,0) is a stable fixed-point of **F**. However, by adding a nonlinear term $\sqrt{|x|}y$ into **F** to have $\underline{\mathbf{F}}(x, y) = (\sqrt{|x|}y, \sqrt{|x|} + dy)$, then (0,0) becomes unstable numerically.

More precise, there are three cases:

- for 0 < d < 1, the recurrence oscillates for several steps then goes away from (0, 0) along a curve like y = k√|x| (see Figure 3.7a).
- 2. for -1 < d < 0, the dynamics is different, it seems that the ocurrence always oscillates around (0, 0). This case is harder to see if it is stable or unstable (see Figure 3.7b).
- 3. For d = 0, the fixed-point (0,0) of the linear map **F** is asymptotically stable. It is not asymptotically stable if there is the nonlinear term $\sqrt{|x|y}$, but it might be still stable. In some cases the recurrence may tend to another fixed-point (see Figure 3.7c).

3.6 Dynamics in the vicinity of the grazing orbits

The aim of this section is first to prove the square-root singularity near a grazing contact: Theorems 3.8 and 3.10 and then to study the possible square-root dynamics near periodic solutions with one grazing contact per period. Such periodic solutions of the *N*-dof system are limited to *N* linear grazing orbits and there is no other solutions. Periodic solutions with sticking contacts are excluded in this current work. Take note that $\mathcal{H}_{\mathcal{P}}$ ensures that the first return time exists and finite. A challenging point in building the FRT is to check the nonlocal admissibility condition: $u_N(t) < d$ before the closing contact. Moreover, data may lead to the first return time not in the vicinity of T_0 but instead in the vicinity of $2T_0$, $3T_0$ and so on. We pay attention to studying the class of the initial data near W_0 in \mathcal{H}^- or $\mathcal{H}_{\mathcal{G}}^0$ which lead to some orbits having their FRT near T_0 .

3.6.1 The first return time

This section deals with the proof of Theorem 3.8. It will be shown that, in a class of initial data, the FRT has a particular form containing a square-root term. First, a lemma is stated, then a

long proof is proposed. Recall that $\Phi(t, W) = \mathbf{e}_N^\top \mathbf{R}(t)\mathbf{S}W$ is smooth. It coincides with u_N in the neighborhood of (T_0, W_0) as long as $u_N(t) \le d$.

Lemma 3.48. The set \mathcal{K} of indices $\{i \in \{N + 1, \dots, 2N\}, \partial_{W_i} \Phi(T_0, W_0) \neq 0\}$ is not empty.

In other words, there exists at least one non-vanishing partial derivative of Φ with respect to at least one initial velocity. This means also that a component W_k stated in Theorem 3.8 always exits.

Proof. This is proved by contradiction. The smooth function $\Phi(t, W)$ can be written as

$$\Phi(t, \mathsf{W}) = \mathbf{e}_N^{\mathsf{T}} \mathbf{R}(t) \mathbf{S} \mathsf{W} = \mathbf{e}_N^{\mathsf{T}} \left(\mathbf{P} \cos(t\Omega) \mathbf{P}^{-1} \mathbf{u} + \mathbf{P} \Omega^{-1} \sin(t\Omega) \mathbf{P}^{-1} \mathbf{v} \right).$$
(3.123)

If $\mathcal{K} = \emptyset$ then $\partial_{W_i} \Phi(T_0, W_0) = 0$ for all $i = N + 1, \dots, 2N$. It follows that

$$\mathbf{e}_{N}^{\top} \mathbf{P} \, \mathbf{\Omega}^{-1} \sin(T_{0} \mathbf{\Omega}) \, \mathbf{P}^{-1} \, \mathbf{e}_{i} = 0 \text{ for all } i = 1, \dots, N.$$
(3.124)

In other words, the N^{th} row of the matrix $\mathbf{A} = \mathbf{P} \mathbf{\Omega}^{-1} \sin(T_0 \mathbf{\Omega}) \mathbf{P}^{-1}$ is zero. Therefore, rank $(\mathbf{A}) < N$, this is impossible. Hence $\mathcal{K} \neq \emptyset$.

Now, the implicit equation $\Phi(T, W) = d$ is solved with a function $T = \theta(W)$ on a half neighborhood of W₀. At the end, it is proven that $\theta(W)$ is the first return time T(W).

Proof. Consider the initial condition $W_0 \in \mathcal{H}^- \cup \mathcal{H}^0_{\mathcal{G}}$ at initial instant t = 0, the associated orbit has the first grazing contact at $T_0 = T(W_0)$. The first return time T is implicitly given by the equation $u_N(T, W) = d$. The nonsmooth function u_N is replaced by the smooth function Φ to can apply the implicit function theorem to the equation:

$$\Phi(T, \mathsf{W}) = d,\tag{3.125}$$

where $\Phi(T_0, W_0) = u_N(T_0, W_0) = d$, $\partial_t \Phi(T_0, W_0) = \dot{u}_N^-(T_0, W_0) = 0$. In order to apply Theorem 3.26, the following lemma is needed. As a consequence of Lemma 3.48, there exists $k \in \mathcal{K}$ where $\partial_{W_k} \Phi(T_0, W_0) \neq 0$. Denote \underline{W} is the reduced vector obtained from W by removing W_k . Together with Assumption 3.7, function $\Phi(t, W)$ can be seen as $\Phi(t, W_k, \underline{W})$ where W_0 corresponds to $(W_{0k}, \underline{W}_0)$, and f satisfies the following conditions: $\Phi(T_0, W_{0k}, \underline{M}_0) = d$, $\partial_t \Phi(T_0, W_{0k}, \underline{M}_0) = 0$, $\partial_{W_k} \Phi(T_0, W_{0k}, \underline{M}_0) \neq 0$, $\partial_{tt} \Phi(T_0, W_{0k}, \underline{M}_0) \neq 0$. By applying Theorem 3.26, there exist neighborhoods $V_{\underline{W}_0}$, V_{T_0} and $V_{W_{0k}}$ of \underline{W}_0 , T_0 and W_{0k} respectively and smooth scalar functions η , α such that

$$\eta: V_{\mathsf{W}_0} \to V_{T_0}, \, \underline{\mathsf{W}} \mapsto T = \eta(\underline{\mathsf{W}}) \tag{3.126}$$

$$\alpha: V_{\mathsf{W}_0} \to V_{W_{0k}}, \, \underline{\mathsf{W}} \mapsto W_k = \alpha(\underline{\mathsf{W}}) \tag{3.127}$$

satisfying $\eta(\underline{W}_0) = T_0$ and $\alpha(\underline{W}_0) = W_{0k}$. The set S_c defined as the intersection of the two hypersurfaces $\Phi = d$ and $\partial_t \Phi = 0$ can be parameterized as follows

$$S_c = \{(t, W_k, \underline{\mathsf{W}}) \in V_{T_0} \times V_{W_{0k}} \times V_{\underline{\mathsf{W}}_0}, \ \Phi(t, W_k, \underline{\mathsf{W}}) = d \text{ and } \partial_t \Phi(t, W_k, \underline{\mathsf{W}}) = 0\},$$
(3.128)

$$= \{ (\eta(\underline{\mathsf{W}}), \alpha(\underline{\mathsf{W}}), \underline{\mathsf{W}}), \underline{\mathsf{W}} \in V_{\underline{\mathsf{W}}_0} \}.$$
(3.129)

Recall that $\mathcal{B}_k = \{W = (W_k, \underline{W}) \in V_{W_{0k}} \times V_{W_{0k}}, s_k(W_k - \alpha(\underline{W})) \ge 0\}$ where $s_k = \operatorname{sign}(\gamma_k)$. It is the region adjacent to and including the hypersurface $W_k = \alpha(\underline{W})$. By applying Theorem 3.26, there exists a smooth function ψ such that there are two graphs to solve equation (3.125):

$$T = \theta(\mathsf{W}) = \eta(\underline{\mathsf{W}}) + \psi(\pm \sqrt{s_k(W_k - \alpha(\underline{\mathsf{W}}))}, \underline{\mathsf{W}}), \, \mathsf{W} \in \mathcal{B}_k,$$
(3.130)

where $\psi(0, \underline{W}_0) = 0$ and $\partial_{W_k} \psi(0, \underline{W}_0) = |\gamma_k|^{-1/2}$. We choose the branch of $\theta(W)$ corresponding to the admissibility condition for the velocity at the contact, which is

$$\dot{\Phi}(\theta(\mathsf{W}),\mathsf{W}) \ge 0. \tag{3.131}$$

Denote $F(W) = \mathbf{R}(\theta(W)) \mathbf{S} W$. Using the asymptotic expansion up to the first order of $\sqrt{s_k(W_k - \alpha(\underline{W}_0))}$, in the direction of the k^{th} component of W, and take note that $\eta(\underline{W}_0) = T_0$, one has $\theta(W) = T_0 \pm |\gamma_k|^{-1/2} \sqrt{s_k(W_k - \alpha(\underline{W}_0))} + O(h^2) = T_0 + h + O(h^2)$ where $h = \pm |\gamma_k|^{-1/2} \sqrt{s_k(W_k - \alpha(\underline{W}_0))}$. This also follows that

$$W_k = \alpha(\underline{W}_0) \pm \gamma_k h^2 = W_{0k} + O(h^2), \text{ since } \alpha(\underline{W}_0) = W_{0k} \text{ (see Theorem 3.26)}, \quad (3.132)$$
$$W = W_0 + h^2 \mathbf{e}_k, \ \mathbf{e}_k^\top = [0, \dots, 1, \dots, 0] \in \mathbb{R}^{2N}. \quad (3.133)$$

Similarly,

$$\cos(\theta(\mathsf{W})\,\mathbf{\Omega}) = \cos\left(\left(T_0 + h + O(h^2)\right)\mathbf{\Omega}\right) = \cos(T_0\,\mathbf{\Omega}) - h\,\mathbf{\Omega}\sin(T_0\,\mathbf{\Omega}) + O(h^2), \quad (3.134)$$

$$\sin(\theta(\mathsf{W})\,\mathbf{\Omega}) = \sin\left((T_0 + h + O(h^2))\,\mathbf{\Omega}\right) = \sin(T_0\,\mathbf{\Omega}) + h\,\mathbf{\Omega}\cos(T_0\,\mathbf{\Omega}) + O(h^2). \quad (3.135)$$

Hence, $\mathbf{R}(\theta(\mathbf{W}))$ can be written as $\mathbf{R}(T_0 + h + O(h^2)) = \mathbf{R}(T_0) + h\dot{\mathbf{R}}(T_0) + O(h^2)$ where $\dot{\mathbf{R}}(t)$ denotes the matrix whose elements are the derivatives of the elements of \mathbf{R} :

$$\dot{\mathbf{R}}(t) = \begin{bmatrix} -\mathbf{P}\,\Omega\,\sin(t\,\Omega)\mathbf{P}^{-1} & \mathbf{P}\cos(t\,\Omega)\mathbf{P}^{-1} \\ -\mathbf{P}\,\Omega^{2}\cos(t\,\Omega)\mathbf{P}^{-1} & -\mathbf{P}\,\Omega\sin(t\,\Omega)\mathbf{P}^{-1} \end{bmatrix}.$$
(3.136)

Inserting these into F and neglecting the higher order term of h leads to

$$\mathsf{F}(\mathsf{W}) \approx \left(\mathbf{R}(T_0) + h \, \dot{\mathbf{R}}(T_0) \right) \, \mathbf{S} \, \mathsf{W}_0 \approx \begin{bmatrix} \mathbf{u}(T_0, \mathsf{W}_0) \\ \mathbf{v}(T_0, \mathsf{W}_0) \end{bmatrix} + h \begin{bmatrix} \mathbf{v}(T_0, \mathsf{W}_0) \\ -\mathbf{K}\mathbf{u}(T_0, \mathsf{W}_0) \end{bmatrix}.$$
(3.137)

Thus, $\dot{\Phi}(\theta(\mathsf{W}), \mathsf{W}) = \mathbf{e}_{2N}^{\top} P(\mathsf{W}) \approx \mathbf{e}_N^{\top} \left(\mathbf{v}(T_0, \mathsf{W}_0) - h \, \mathbf{K} \, \mathbf{u}(T_0, \mathsf{W}_0) \right) = -h \, \mathbf{e}_N^{\top} \mathbf{K} \, \mathbf{u}(T_0, \mathsf{W}_0).$ where \mathbf{e}_{2N} is the vector in \mathbb{R}^{2N} such that all coordinates are 0 except the $2N^{\text{th}}$ one equal to 1, similarly for the vector \mathbf{e}_N in \mathbb{R}^N . Based on the N^{th} equation of (3.1) taken at $t = T_0$, $m_N \ddot{u}_N^{-}(T_0) + \mathbf{e}_N^{\top} \mathbf{K} \, \mathbf{u}(T_0) = 0$ it follows that $m_N \ddot{u}_N^{-}(T_0, \mathsf{W}_0) = -\mathbf{e}_N^{\top} \mathbf{K} \, \mathbf{u}(T_0, \mathsf{W}_0)$. Thus,

$$\dot{\Phi}(\theta(\mathsf{W}),\mathsf{W}) \approx h \, m_N \ddot{u}_N^-(T_0,\mathsf{W}_0). \tag{3.138}$$

Therefore, since $\ddot{u}_N(T_0, W_0) < 0$ then, in order to have $\dot{\Phi}(\theta(W)) \ge 0$, *h* is chosen with the negative sign. Hence, $\theta(W)$ satisfies (3.131).

The proof of Theorem 3.8 ends by showing that $\theta(W) = T(W)$. This means that $\Phi(t, W) < d$ for all $t \in [0; \theta(W)[$.

Proof. First, $\ddot{\Phi}(T_0, W_0) < 0$ since it is not zero and $\dot{\Phi}(T_0, W_0) = 0$, so it is mandatory to have $\Phi(t, W_0) < d$ for $t < T_0$ and t near T_0 . By the smoothness of Φ with respect to (t, W), there exists a neighborhood $[T_-; T_+] \times V_1$ of (T_0, W_0) and $\delta_1 > 0$ such that $\ddot{\Phi}(T, W) < -\delta_1 < 0$ in this neighborhood and $\theta(W) \in [T_-; T_+]$ for $W \in V_1$. This will be the crucial point to conclude at the end that $\Phi(t, W) < d$ for all $t \in [T_-; \theta(W)]$.

Second, $\dot{\Phi}(0, W_0) < 0$ because W_0 belongs to \mathcal{H}^- . Thus there exists a neighborhood $[0; T_2] \times V_2$ of $(0, W_0)$ with $V_2 \subset V_1$ and $\delta_2 > 0$ such that $\dot{\Phi}(T, W) < -\delta_2 < 0$ in this neighborhood. This implies that $\Phi(t, W) < d - \delta_2 t < d$ for all $t \in [0; T_2]$.

Let $\delta_3 = 2^{-1}\delta_2 T_2 > 0$, there exists a neighborhood of W_0 denoted by $V_3 \subset V_2$ such that $|\Phi(t, W) - \Phi(t, W_0)| < \delta_3$ for all $(t, W) \in [T_2; T_-] \times V_3$. This yields $\Phi(t, W) < d - \delta_3 < d$ for all $(t, W) \in [T_2; T_-] \times V_3$.

Finally, on $[T_-; \theta(W)]$ for $W \in V_3$, the function $\Phi(t, W)$ is concave and the velocity at time $\theta(W)$ is nonnegative obtained from (3.131), so $\Phi(t, W) < d$ on $[T_-; \theta(W)] \times V_3$ which concludes the proof.

This ends the proof of main Theorem 3.8.

The proof of Theorem 3.10 is the same except two points: the neighborhood of W_0 is smaller and the proof that $\Phi(t, W) < d$ for 0 < t small enough. The difference is that the initial velocity of the last mass is zero. Hopefully, the acceleration is negative: $\ddot{u}_N^+(0, W_0) < 0$ which is enough to get $\Phi(t, W) < d$ for 0 < t small enough and conclude the proof of Theorem 3.10.

3.6.2 Square-root dynamics in the vicinity of a linear grazing orbit (LGO)

This subsection explores the possible square-root dynamics near the LGO. One feature of the LGO is that the sticking phase does not occur near such a mode. This property is proven in the coming Proposition. Then, the square-root dynamics coefficients are computed. At least one of them, non-vanishing, will activate the square-root dynamics near the LGOs. If the first return map has a particular expression with the square-root term in a class of initial data then one may expect the instability of the LGOs.

Throughout this section, let us consider the j^{th} LGO (the notation LGO_j is also used below) with period $T_j = T(W_0)$ [25] associated to the initial data

$$\mathbf{W}_0 = [\mathbf{u}_0^{\mathsf{T}}, \mathbf{v}_0^{\mathsf{T}}]^{\mathsf{T}}, \quad \mathbf{u}_0 = \frac{d}{P_{Nj}} \mathbf{P} \mathbf{e}_j, \quad \mathbf{v}_0 = \mathbf{0}.$$
(3.139)

Proposition 3.49. The sticking phase does not occur near the j^{th} LGO, j = 1, ..., N.

Proof. From the *N*th equation in (3.1), $m_N \ddot{u}_N(t) + \mathbf{e}_N^\top \mathbf{K} \mathbf{u}(t) = R(t)$, $R(t) \le 0$, the sticking phase does not occur if $u_N(t) = d$ and $F(t) = \mathbf{e}_N^\top \mathbf{K} \mathbf{u}(t) > 0$. By the periodicity of LGO, it is sufficient to show that F(0) > 0. The initial data of the LGO yields

$$F(0) = \frac{d}{P_{Nj}} \mathbf{e}_N^{\mathsf{T}} \mathbf{K} \mathbf{P} \mathbf{e}_j = \frac{d}{P_{Nj}} \mathbf{e}_N^{\mathsf{T}} \mathbf{M} \mathbf{P} \, \mathbf{\Omega}^2 \, \mathbf{P}^{-1} \mathbf{P} \mathbf{e}_j = dm_N \omega_j^2 > 0 \text{ for } d > 0.$$
(3.140)

In some cases when d > 0 and for some initial data in \mathcal{B}_k^+ defined in Theorem 3.10, the associated orbit takes less time to come back to the Poincaré section. This is the consequence of Theorem 3.10 and a feature of the LGO.

Corollary 3.50. Consider a LGO with period T_j and an initial perturbation of its initial data $W = W_0 + w \mathbf{e}_k$ with $k \neq 2N$ (or k = 2N and $\sigma > 0$ where $\sigma = sign(\gamma_k)$), then there exists $\delta > 0$ such that $0 < \sigma w < \delta$ and $T(W) \le T_j$.

Such one-sided condition on the first return time is already known for nonlinear modes with one impact per period near a LGO [25]. It is expected that this inequality is valid in a larger set near W₀ but of course not in the whole neighborhood if $k \neq 2N$.

Proof. To apply Theorem 3.10, Assumption 3.7 must be verified. The second time derivative of u_N at the grazing point is

$$\ddot{u}_N(T_j, \mathbf{W}_0) = -\frac{1}{m_N} \mathbf{e}_N^{\mathsf{T}} \mathbf{K} \mathbf{u}(T_j) = -\frac{1}{m_N} \mathbf{e}_N^{\mathsf{T}} \mathbf{K} \mathbf{u}(0) = -d\omega_j^2 < 0.$$
(3.141)

Hence, by Theorem 3.8, the first return T takes the form given by (3.17). Moreover, since



FIGURE 3.8 – First return time (red lines) with respect to the initial displacement of the first mass: (a) near the first LGO, (b) near the second LGO.

 $\ddot{u}_N(T_0, W_0) < 0$, it follows that in the direction of the k^{th} component of W,

$$T = \eta(\underline{\mathbf{W}}_0) + \psi(-\sqrt{\sigma w}, \underline{\mathbf{W}}_0), \quad \eta(\underline{\mathbf{W}}_0) = T_j, \quad \partial_{W_k}\psi(0, \underline{\mathbf{W}}_0) = 1/\sqrt{|\gamma_k|}.$$
(3.142)

Hence, $T = T_j - 1/\sqrt{|\gamma_k|}\sqrt{\sigma w} + O(w) \le T_j$.

Let us name $u_{1LGO_1}(0)$ and $u_{1LGO_2}(0)$, the initial displacement of the first mass along LGO₁ and LGO₂, respectively. The square-root singularity of the FRT with respect to the initial u_1 , when $u_1(0) \ge u_{1LGO_1}(0)$ or $u_1(0) \le u_{1LGO_2}(0)$ is shown in Figure 3.8. The red lines are computed numerically, while the dashed lines illustrate the Taylor expansion of ψ . It is seen that in both cases, $T \le T_j$, j = 1, 2.

The following is the proof of Theorem 3.14 which gives the computation of the coefficients C_k and the generic condition that causes the square-root dynamics near the LGO.

Proof. As proven in Proposition 3.49, the sticking phase does not occur near the LGOs, thus the first return map takes the form (3.20a), $\mathcal{F}(W) = \mathbf{R}(T(W))\mathbf{S}W$ where the FRT has the square-root dependence (3.17) in the subset \mathcal{B}_k^+ . Using the proof in Subsection 3.6.1, the first return map \mathcal{F} can be rewritten as $\mathcal{F}(W) = \mathbf{G}(\bar{W}) = (g_1, \ldots, g_{2N})(\bar{W})$ where $\bar{W} = [\bar{W}_i]_{i=1}^{2N}$, with a change of variables:

$$\bar{W}_k = \sqrt{s_k(W_k - \alpha(\underline{W}_0))} \text{ and } \bar{W}_i = \underline{W}_i, \quad \forall i \neq k.$$
 (3.143)

Note that $g_N(\bar{W}) = d$. From the proof in Subsection 3.6.1, it follows that $\bar{W}_k = \sqrt{|\gamma_k|}h$, together with (3.137), \mathcal{F} is written as

$$\mathcal{F}(\mathsf{W}) = \mathbf{G}(W_1, \dots, \sqrt{|\gamma_k|}h, \dots, W_{2N}) \approx \begin{bmatrix} \mathbf{u}(T_j, \mathsf{W}_0) \\ \mathbf{v}(T_j, \mathsf{W}_0) \end{bmatrix} + h \begin{bmatrix} \mathbf{v}(T_j, \mathsf{W}_0) \\ -\mathbf{K}\mathbf{u}(T_j, \mathsf{W}_0) \end{bmatrix}.$$
(3.144)

For $k \in \{1, ..., 2N\}$, $k \neq N$ such that $\partial_{W_k} u_N(T_j, W_0) \neq 0$, the coefficient associated with the square-root term *h* is $C_k = a_{kk} = \partial_{\bar{W}_k} g_k(T_j, W_0) = \sqrt{|\gamma_k|} \partial_h g_k(T_j, W_0)$. The expression of C_k is then obtained via (3.144),

$$C_k = \begin{cases} \sqrt{|\gamma_k|} \mathbf{e}_k^{\mathsf{T}} \mathbf{v}(T_j, \mathsf{W}_0) & \text{if } 1 \le k < N, \end{cases}$$
(3.145a)

$$\int \left(-\sqrt{|\gamma_k|} \mathbf{e}_{k-N}^{\mathsf{T}} \mathbf{K} \mathbf{u}(T_j, \mathbf{W}_0) \right) \qquad \text{if } N < k \le 2N.$$
(3.145b)

Since $\mathbf{v}(T_j) = \mathbf{v}_0 = \mathbf{0}$, it follows that $C_k = 0$, $\forall 1 \le k < N$. Another way to write C_k is by using the modal coordinates $[\mathbf{q}, \dot{\mathbf{q}}]$ where $\mathbf{u} = \mathbf{P}\mathbf{q}$, as follows:

$$C_{k} = \begin{cases} \sqrt{|\gamma_{k}|} \mathbf{e}_{k}^{\mathsf{T}} \mathbf{P} \dot{\mathbf{q}}(T_{j}, \mathsf{W}_{0}) & \text{if } 1 \le k < N, \\ \sqrt{|\gamma_{k}|} \mathbf{e}_{k}^{\mathsf{T}} \mathbf{P} \dot{\mathbf{q}}(T_{j}, \mathsf{W}_{0}) & \text{if } N \le k < 2N \end{cases}$$
(3.146a)

$$\left(-\sqrt{|\gamma_k|}\mathbf{e}_{k-N}^{\mathsf{T}}\mathbf{M}\mathbf{P}\,\mathbf{\Omega}^2\,\mathbf{q}(T_j,\mathsf{W}_0)\quad\text{if }N< k\le 2N.\right.$$
(3.146b)

It is shown in Lemma 3.48 that $\mathcal{K} \neq \emptyset$. For each $k \in \mathcal{K}$, the coefficient C_k is

$$C_k = -\sqrt{|\gamma_k|} \mathbf{e}_{k-N}^{\top} \mathbf{M} \mathbf{P} \, \mathbf{\Omega}^2 \, \mathbf{q}(T_j) = -\sqrt{|\gamma_i|} dm_i \omega_j^2 \frac{P_{kj}}{P_{Nj}}.$$
(3.147)

By the hypothesis that there exists $i \in \mathcal{K}$ where $P_{ij} \neq 0$, it follows that $C_i \neq 0$. This non-vanishing coefficient then facilitates the square-root dynamics near the LGO.

The square-root dynamics is however activated near W_0 and in a particular class of initial data. If the orbits stay in that regime of initial data, the dynamics will follows the framework of Section 3.5 and one may be able to determine the instability of the LGOs.

Conclusions

The existence of the solutions to the *N*-degree-of-freedom vibro-impact system with a unilateral contact condition and without source terms is given in chapter 1. The difficulty while giving a simple proof for the uniqueness of the solution of (1.1) is also discussed.

In chapter 2, a class of periodic solutions involving sticking phases is investigated on a two-degree-of-freedom linear oscillator subjected to a unilateral contact condition. In particular, periodic orbits with one sticking phase per period are considered. The set of such orbits is characterized by only one parameter, the free flight duration, which belongs to a discrete set. Hence, solutions with 1-SPP might or might not exist, and when they do exist, they are isolated. This property can also be seen via the one-to-one correspondence between the solutions with 1-SPP and the ones with 1-IPP. A numerical procedure is given in order to seek for all possible 1-SPP and examples of 1-SPPs are presented. The prestressed structure is also explored; the solutions with 1-SPP with finite duration of sticking phase as well as with infinite duration of sticking phase are found. For *N*-degree-of-freedom with N > 2, the symmetry $\mathbf{u}(t) = \mathbf{u}(-t)$ which is heavily used in the two-degree-of-freedom is unknown, and the sticking system is of N - 1 > 1 dimensions. For these reasons, the results on the periodic solutions to *N*-degree-of-freedom vibro-impact systems involving sticking phases remain to be done.

The First Return Map which is used in seeking for periodic solutions is investigated in order to study the stability or instability of the periodic orbits. This map is determined based on its domain of definition- the Poincaré section and the First Return Time of the orbits to this section. In chapter 3, the well-known square-root singularity of the First Return Time in the vicinity of grazing orbits is carefully studied in a rigorous mathematical framework. In this work, the Poincaré section, that is the domain of definition of the First Return Map, is chosen to be a subset of the impact hyperplane. This critical choice ensures that the FRM is well defined as the orbit comes back to the Poincaré section an infinite number of times. The square-root singularity is shown to appear and it is only defined on a subset of the neighborhood of the initial data leading to a grazing orbit. This implies the discontinuity of the FRT. This result may give rise to a possibly discontinuous FRM as well as immediate instability of the periodic grazing orbits. The square-root dynamics is shown to emerge from the square-root singularity of the FRT if one of the coefficients related to a linear map does not vanish. Under a generic condition of the matrix

of eigenvectors, the square-root dynamics near a linear grazing orbit is proven to exist. However, this square-root dynamics is only activated in a particular class of initial data. Hence, the theory on the instability of the fixed point of a map exhibiting the square-root term is not applied directly. Further study on the FRM is needed in order to understand better the global dynamics. The continuity of the FRM remains an open question. When this is clear then the stability of the linear grazing orbits and 1-SPP periodic solutions follow as the next steps. Besides, challenges on the results of the First Return Time arise with the presence of internal resonances. It is however interesting since there are many periodic solutions involving one grazing contact per period other than the N linear grazing modes (LGM).

Bibliography

- Vincent ACARY and Bernard BROGLIATO. Numerical Methods for Nonsmooth Dynamical Systems. Vol. 35. Lecture Notes in Applied and Computational Mechanics. Springer, 2008. ISBN: 9783540753926. [inria-00423530].
- [2] Robert ADAMS. *Sobolev Spaces*. Vol. 65. Pure and Applied Mathematics. Academic Press, 1975. ISBN: 9780120441433.
- [3] Marat AKHMET and Aysegul KIVILCIM. "Periodic motions generated from non-autonomous grazing dynamics". *Communications in Nonlinear Science and Numerical Simulation* 49 2017, pp. 48–62. [arxiv-1608.07790].
- [4] Ricardo ALZATE, Mario di BERNARDO, Umberto MONTANARO, and Stefania SANTINI. "Experimental and numerical verification of bifurcations and chaos in cam-follower impacting systems". *Nonlinear Dynamics* 50(3) 2007, pp. 409–429. [hal-01304369].
- [5] Patrick BALLARD. "The dynamics of discrete mechanical systems with perfect unilateral constraints". *Archive for Rational Mechanics and Analysis* 154(3) 2000, pp. 199–274. [hal-00111308].
- [6] Jérôme BASTIEN, Frédéric BERNARDIN, and Claude-Henri LAMARQUE. Systèmes dynamiques discrets non réguliers déterministes ou stochastiques : applications aux modèles avec frottement ou impact. Mécanique des structures. Hermes Science, 2012. ISBN: 9782746239081.
- [7] Mario di BERNARDO, Chris BUDD, Alan CHAMPNEYS, and Piotr KOWALCZYK. *Piecewise-Smooth Dynamical Systems. Theory and Applications*. Applied Mathematical Sciences. Springer Science & Business Media, 2008. ISBN: 9781846280399. DOI: 10.1007/978-1-84628-708-4.
- [8] Chris Budd and Felix Dux. "Chattering and related behaviour in impact oscillators". *Philosophical Transactions of the Royal Society London A* 347(1683) 1994, pp. 365–389. [hal-01304424].
- [9] David Chillingworth. "Discontinuity geometry for an impact oscillator". *Dynamical Systems* 17(4) 2002, pp. 389–420. [hal-01544324].
- [10] David CHILLINGWORTH. "Dynamics of an impact oscillator near a degenerate graze". *Nonlinearity* 23(11) 2010, pp. 2723–2748. [hal-01390258].
- [11] Wai CHIN, Edward OTT, Helena NUSSE, and Celso GREBOGI. "Grazing bifurcation in impact oscillators". *Physical Review E* 50(6) 1994, pp. 4427–4444. [hal-01386141].
- [12] Earl Alexander CODDINGTON and Norman LEVINSON. *Theory of Ordinary Differential Equations*. International Series in Pure and Applied Mathematics. McGraw-Hill, 1955. ISBN: 9780898747553.
- [13] Constantin CORDUNEANU. *Almost Periodic Functions*. Interscience tracts in pure and applied mathematics. Interscience Publishers, 1968. ISBN: 9780828403313.
- [14] Jean-Michel CORON. Control and Nonlinearity. Vol. 136. Mathematical Surveys and Monographs. American Mathematical Society Providence RI, 2007. ISBN: 9780821849187.
- [15] Harry DANKOWICZ and Jenny JERRELIND. "Control of near-grazing dynamics in impact oscillators". Proceeding of The Royal Society 461(2063) 2005, pp. 3365–3380. [hal-01544507].
- [16] David DOYEN, Alexandre ERN, and Serge PIPERNO. "Time-integration schemes for the finite element dynamic Signorini problem". SIAM Journal of Scientific Computing 33(1) 2011, pp. 223–249. [hal-00440128].
- [17] Mats FREDRIKSSON and Arne NORDMARK. "Bifurcations caused by grazing incidence in many degrees of freedom impact oscillators". *Proceedings of the Royal Society* 453(1961) 1997, pp. 1261– 1276. [hal-01297285].
- [18] John GUCKENHEIMER and Phillip HOLMES. Nonlinear Oscillations, Dynamical Systems and Bifurcations of Vector Fields. Vol. 42. Applied Mathematical Sciences. Springer, 1989. ISBN: 9781461211402. DOI: 10.1007/978-1-4612-1140-2.
- [19] Sokly HENG. Nonlinear normal modes of impact oscillators. 2013.
- [20] Raouf IBRAHIM. Vibro-Impact Dynamics. Modeling, Mapping and Applications. Vol. 43. Lecture Notes in Applied and Computational Mechanics. Springer, 2009. ISBN: 9783642002755. DOI: 10.1007/978-3-642-00275-5.
- [21] Alexander IVANOV. "The stability of periodic solutions of discontinuous systems that intersect several surfaces of discontinuity". *Journal of Applied Mathematics and Mechanics* 62 1998, pp. 677–685. [hal-01354035].
- [22] Gatëan KERSCHEN, Jean-Claude GOLINVAL, and Alexander VAKAKIS. "Nonlinear normal modes, Part I: A useful framework for the structural dynamicist". *Mechanical Systems and Signal Processing* 23 2009, pp. 170–194. [hal-01357931].
- [23] Claude-Henri LAMARQUE and Olivier JANIN. "Modal analysis of mechanical systems with impact non-linearities: limitations to a modal superposition". *Journal of Sound and Vibration* 235(4) 2000, pp. 567–609. [hal-01510826].
- [24] Denis LAXALDE and Mathias LEGRAND. "Nonlinear Modal Analysis of Mechanical Systems with Frictionless Contact Interfaces". *Computational Mechanics* 2011. [hal-00492775].
- [25] Mathias LEGRAND, Stéphane JUNCA, and Sokly HENG. "Nonsmooth modal analysis of a *N*-degreeof-freedom system undergoing a purely elastic impact law". *Communications in Nonlinear Science and Numerical Simulation* 45 2017, pp. 190–219. [hal-01185980].
- [26] Remco LEINE and Henk NIJMEIJER. Dynamics and Bifurcations of Non-Smooth Mechanical Systems. Vol. 18. Lecture Notes in Applied and Computational Mechanics. Springer, 2004. ISBN: 9783642060298. DOI: 10.1007/978-3-540-44398-8.
- [27] Remco LEINE and Nathan van de WOUW. Stability and Convergence of Mechanical Systems with Unilateral Constraints. Vol. 36. Lecture Notes in Applied and Computational Mechanics. Springer, 2008. ISBN: 9783540769750. DOI: 10.1007/978-3-540-76975-0.
- [28] Huong LETHI, Stéphane JUNCA, and Mathias LEGRAND. "Periodic solutions of a two-degree-of-freedom autonomous vibro-impact oscillator with sticking phases". *preprint* 2016. [hal-01305719].
- [29] G.-W. Luo. "Hopf-flip bifurcations of vibratory systems with impacts". *Nonlinear Analysis: Real World Applications* 7(5) 2006, pp. 1029–1041. [hal-01352124].
- [30] G.-W. Luo and J.-H. XIE. "Hopf bifurcation of a two-degree-of-freedom vibro-impact system". *Journal of Sound and Vibration* 213(3) 1998, pp. 391–408. [hal-01544342].
- [31] Jean-Jacques MOREAU. "Standard inelastic shocks and the dynamics of unilateral constraints". Unilateral Problems in Structural Analysis. Ed. by G. Del PIERO and F. MACERI. CISM Courses and Lectures 288. Springer-Verlag, 1983, pp. 173–221. [hal-01544442].

- [32] El Hadi Moussi, Sergio Bellizzi, Bruno Cochelin, and Ionel Nistor. "Nonlinear normal modes of a two-degree-of-freedom piecewise linear system". *Mechanical Systems and Signal Processing* 64-65 2015, pp. 266–281. [hal-00783088].
- [33] Arne NORDMARK. "Existence of periodic orbits in grazing bifurcations of impacting mechanical oscillators". *Nonlinearity* 14(6) 2001, pp. 1517–1542. [hal-01297283].
- [34] Arne NORDMARK and Petri PIIROINEN. "Simulation and stability analysis of impacting systems with complete chattering". *Nonlinear Dynamics* 58(1-2) 2009, pp. 85–106. [hal-01304375].
- [35] Madeleine PASCAL. "Dynamics and stability of a two-degree-of-freedom oscillator with an elastic stop". *Journal of Computational and Nonlinear Dynamics* 1(1) 2005, pp. 94–102. [hal-00342874].
- [36] Danilo PERCIVALE. "Uniqueness in the elastic bounce problem, I". *Journal of Differential Equations* 56(2) 1985, pp. 206–215. [hal-01294061].
- [37] Danilo PERCIVALE. "Uniqueness in the elastic bounce problem, II". *Journal of Differential Equations* 90 1991, pp. 304–315. [hal-01544491].
- [38] Valery PILIPCHUK. "Impact modes in discrete vibrating systems with rigid barriers". *International Journal of Non-Linear Mechanics* 36(6) 2001, pp. 999–1012. [hal-01295245].
- [39] Reinhardt ROSENBERG. "On nonlinear vibrations of systems with many degrees of freedom". *Advances in Applied Mechanics* 9 1966, pp. 155–242. [hal-01544496].
- [40] Michelle SCHATZMAN. "A class of nonlinear differential equations of second order in time". *Nonlinear Analysis: Theory, Methods & Applications* 2(3) 1978, pp. 355–373. [hal-01294058].
- [41] Michelle SCHATZMAN. "Uniqueness and continuous dependence on data for one dimensional impact problems". *Mathematical and Computational Modelling* 28 1998, pp. 1–18. [hal-01544389].
- [42] Laurent SCHWARTZ. *Méthodes mathématiques pour les sciences physique*. Vol. 65. Collection Enseignement des sciences 3. Hermann, 1965. ISBN: 9782730208413.
- [43] Steven SHAW and Christophe PIERRE. "Nonlinear normal modes and invariant manifolds". *Journal of Sound and Vibration* 150(1) 1991, pp. 170–173. [hal-01310674].
- [44] Steven SHAW and Christophe PIERRE. "Normal modes for non-linear vibratory systems". *Journal of Sound and Vibration* 164(1) 1993, pp. 85–124. [hal-01544500].
- [45] Anders THORIN, Pierre DELEZOIDE, and Mathias LEGRAND. "Nonsmooth modal analysis of piecewise-linear impact oscillators". *Journal on Applied Dynamical Systems* 2017. [hal-01298983].
- [46] Anders THORIN, Mathias LEGRAND, and Stéphane JUNCA. "Nonsmooth modal analysis: investigation of a two-dof spring-mass system subject to an elastic impact law". *Proceedings of the ASME IDETC/CIE. 11th International Conference on Multibody Systems, Nonlinear Dynamics, and Control.* Vol. 6. 2015. [hal-01185973].
- [47] Phanikrishna THOTA. "Analytical and Computational Tools for the Study of Grazing Bifurcations of Periodic Orbits and Invariant Tori". PhD thesis. Virginia Polytechnic Institute and State University, 2007. [tel-01330429].
- [48] Alexander VAKAKIS. "Nonlinear normal modes and their applications in vibration theory: An overview". *Mechanical Systems and Signal Processing* 11(1) 1997, pp. 3–22. [hal-01354037].
- [49] Alexander VAKAKIS, Leonid MANEVITCH, Yuri MIKHLIN, Valery PILIPCHUK, and Alexandr ZEVIN. Normal modes and localization in nonlinear systems. John Wiley & Sons, 1996. ISBN: 9783527617876.
- [50] David WAGG. "Periodic sticking motion in a two-degree-of-freedom impact oscillator". *International Journal of Non-Linear Mechanics* 40(8) 2005, pp. 1076–1087. [hal-01509345].

- [51] Carlos YOONG, Anders THORIN, and Mathias LEGRAND. *Nonsmooth modal analysis of an elastic bar subject to a unilateral contact constraint*. preprint. 2017. [hal-01471341].
- [52] Carlos Yoong, Anders THORIN, and Mathias LEGRAND. "The Wave Finite Element Method applied to a one-dimensional linear elastodynamic problem with unilateral constraints". *Proceedings of the ASME 2015 IDETC/CIE. 11th International Conference on Multibody Systems, Nonlinear Dynamics, and Control.* Vol. 6. Boston, United States, 2015. [hal-01194922].

Appendix A

Power-root singularity for a mass-spring chain

In the case of a chain, i.e. the stiffness matrix **K** has the form as in (3.95), there is a simpler proof for Proposition 3.34 as stated below.

Lemma A.1. Suppose $\mathbf{u}(t)$ is a solution to (3.1) which models a chain of N masses and has a closing contact at t = 0. Under the assumption that $k_{j+1,j} \neq 0$ for all j = 1, ..., N - 1, then $\operatorname{mult}(\dot{u}_N)(0) \leq 2N - 1$. Moreover, if there is sticking phase starting at t = 0, then $\operatorname{mult}(\dot{u}_{N-1})(\tau) \leq 2N - 3$.

Proof. The proof includes two parts. The first part shows that $\operatorname{mult}(\dot{u}_N^-)(0) \le 2N - 1$. Otherwise, $\operatorname{mult}(\dot{u}_N^-)(0) \ge 2N$, i.e. $\dot{u}_N^-(0) = \cdots = u_N^{(2N-1)-}(0) = u_N^{(2N)-}(0) = 0$ and it is shown that $\mathbf{u}(0) = \mathbf{0}$ and $\dot{\mathbf{u}}(0) = \mathbf{0}$.

Outside the closing contacts, the system is linear:

$$\mathbf{M}\ddot{\mathbf{u}} + \mathbf{K}\mathbf{u} = \mathbf{0}.\tag{A.1}$$

Let $\mathbf{v} = \dot{\mathbf{u}}$, then outside the closing contacts, \mathbf{v} is the solution of the linear system

$$\mathbf{M}\ddot{\mathbf{v}} + \mathbf{K}\mathbf{v} = \mathbf{0}.\tag{A.2}$$

Moreover, $v_N = \dot{u}_N$, from the assumption, v_N satisfies $v_N^{(\ell)}(0) = 0$ for all $1 \le \ell \le 2N - 1$. From the last equation of (A.2): $m_N \ddot{v}_N + k_{N,N-1} v_{N-1} + k_{NN} v_N = 0$, and since $k_{N,N-1} \ne 0$, it follows that $v_{N-1}(0) = \cdots = v_N^{(2N-3)}(0) = 0$. Similarly, from the $(N-1)^{\text{th}}$ equation and by the assumption that $k_{N-1,N-2} \ne 0$, $v_{N-2}(0) = \cdots = v_{N-2}^{(2N-5)}(0) = 0$. In the end, the second equation and $k_{21} \ne 0$ give $v_1(0) = \dot{v}_1(0) = 0$. As a consequence, we have $\mathbf{v}(0) = \mathbf{0}$, $\dot{\mathbf{v}}(0) = \mathbf{0}$ and the corresponding solution is $\mathbf{v} \equiv \mathbf{0}$, $\dot{\mathbf{v}} \equiv \mathbf{0}$. Thus, $\dot{\mathbf{u}} \equiv \mathbf{0}$ and $\ddot{\mathbf{u}} \equiv \mathbf{0}$. Substitution of these identities into the linear system (A.1) yields $\mathbf{Ku} = \mathbf{0}$. This induces $\mathbf{u} \equiv \mathbf{0}$ since det $(\mathbf{K}) \ne 0$. However, this contradicts the fact that $u_N(0) = d > 0$, i.e. the solution cannot rest at its equilibrium $\mathbf{0}$.

The next part shows that $\operatorname{mult}(\dot{u}_{N-1})(\tau) \leq 2N - 3$ when a sticking phase arises after t = 0. Based on the last equation of (A.1), $m_N \ddot{u}_N + k_{N,N-1} u_{N-1} + k_{NN} u_N = 0$, the end of the sticking phase at $t = \tau$ satisfies

$$k_{N,N-1}u_{N-1}(\tau) = -k_{NN}d,$$
(A.3)

$$k_{N,N-1}u_{N-1}(\tau+\delta) > -k_{NN}d, \quad \forall \, 0 < \delta < 1.$$
(A.4)

The sticking system is then a (N-1) degree-of-freedom non-homogeneous system $\underline{\mathbf{M}}\mathbf{\ddot{u}} + \underline{\mathbf{K}}\mathbf{u} = \underline{\mathbf{C}}$ where $\mathbf{C} = [0, \dots, 0, k_{N-1,N}d]^{\top}$. Now, assume that $\operatorname{mult}(\dot{u}_{N-1})(\tau) \ge 2N-2$, then $\dot{u}_{N-1}(\tau) = \cdots$ $= u_{N-1}^{(2N-2)}(\tau) = 0$. By a similar proof for the reduced $(N-1) \times (N-1)$ system with the last entry of the solution is u_{N-1} instead of u_N , it follows that $\operatorname{mult}(\dot{u}_{N-1})(\tau) \le 2N-3$. \Box

Appendix B

One-impact-per-period dynamics with internal resonances near grazing orbits

Summary In this appendix, we discuss the behaviour of periodic solutions with one-impactper-period (1-IPP) where the period is near linear natural frequencies in the case of internal resonances [25].

Consider the *N*-dof system (3.1) with the assumption of internal resonances, i.e. $(w_1, w_2, ..., w_N)$ are \mathbb{Z} -dependent. That is $w_j = k_j w_1$ for $k_j \in \mathbb{Q}^+$, j = 2, ..., N. Let $\alpha(N)$ be the set of all the non empty subsets of $\{1, 2, ..., N\}$. Suppose T_0 belongs to $\bigcap_{j \in \alpha(N)} T_j \mathbb{N}$.

Remark B.1. For each $T \notin \bigcup_{j=1}^{N} T_j \mathbb{N}$, it is proven in [25] that there is a unique periodic solution with one impact-per-period with *T* as the period. This unique periodic solution corresponds to the initial data

$$\mathbf{u}(0;T) = \dot{u}_N^+(0;T)\mathbf{w}(T)$$
 and $\dot{\mathbf{u}}^+(0;T) = \dot{u}_N^+(0;T)\mathbf{e}_N,$ (B.1)

where $\dot{u}_N^+(0;T) = d/w_N(T)$. Recall that $w_N(T) = \sum_{j=1}^N a_j \Phi_j(T)$ with the coefficients

$$a_j = P_{Nj} P_{jN}^{-1}$$
 and $\Phi_j(t) = \frac{\sin(\omega_j t)}{\omega_j (1 - \cos(\omega_j t))}, \quad j = 1, ..., N.$ (B.2)

Remark B.2. When $T_0 \in \bigcap_{j \in \alpha(N)} T_j \mathbb{N}$, a linear grazing periodic solution of (3.1) is the periodic solution with period T_0 associated with the initial data

$$\mathbf{u}(0;T_0) = [u_k(0;T_0)]_{k=1}^N, \quad u_k(0;T_0) = d \frac{\sum_{j \in \alpha(N)} a_{kj} \omega_j^{-2}}{\sum_{j \in \alpha(N)} a_j \omega_j^{-2}}, \quad k = 1,\dots,N,$$
(B.3)

$$\dot{\mathbf{u}}^+(0;T_0) = \mathbf{0},\tag{B.4}$$

where $w_k(T) = \sum_{j=1}^{N} a_{kj} \Phi_j(T)$ and $a_{kj} = P_{kj} P_{jN}^{-1}$.

In particular, when $\alpha(N)$ is a singleton, then T_0 is kT_j , $k \in \mathbb{N}$, we recover the linear grazing modes or linear subharmonics corresponding to the linear natural frequency T_j or the subharmonic frequency kT_j in the non internal resonance case.

Proposition B.3. Suppose $T \notin \bigcup_{j=1}^{N} T_j \mathbb{N}$, *T* is near $T_0 \in \bigcap_{j \in \alpha(N)} T_j \mathbb{N}$. Then the initial data $[\mathbf{u}(0; T), \dot{\mathbf{u}}^+(0; T)]$ in (B.1) converges to the initial data (B.3) of the corresponding grazing orbit.

Proof. If $T_0 \in \bigcap_{j \in \alpha(N)} T_j \mathbb{N}$ then $T_0 = m_j T_j$, for $j \in \alpha(N)$. Hence, for every $j \in \alpha(N)$, the function Φ_j can be expanded as

$$\Phi_{j}(T) = \frac{\sin(\omega_{j}T)}{\omega_{j}(1 - \cos(\omega_{j}T))} \sim \frac{2}{\omega_{j}^{2}(T - m_{j}T_{j})} \sim \frac{2}{\omega_{j}^{2}(T - T_{0})}.$$
(B.5)

Thus

$$w_N(T) = \sum_{j=1}^N a_j \Phi_j(T) \sim \sum_{j \in \alpha(N)} a_j \Phi_j(T) \sim \sum_{j \in \alpha(N)} \frac{2a_j}{\omega_j^2} \frac{1}{T - T_0} \quad \text{when} \quad T \to T_0.$$
(B.6)

From the condition $u_N(0;T_0) = d = \dot{u}_N^+(0;T)w_N(T)$,

$$\dot{u}_{N}^{+}(0;T) \sim \frac{d}{\sum_{j \in \alpha(N)} a_{j} \Phi_{j}(T)} \sim \frac{d}{\sum_{j \in \alpha(N)} 2a_{j} \omega_{j}^{-2}} \left(T - T_{0}\right).$$
(B.7)

It follows that $\dot{u}_N^+(0;T) \to 0$ when $T \to T_0$. The study of the initial displacements is similar. Precisely, the initial displacement is $u_k(0;T) = dw_k(T)/w_N(T)$ where

$$w_k(T) = \sum_{j=1}^N a_{kj} \Phi_j(T) \sim \sum_{j \in \alpha(N)} \frac{2a_{kj}}{\omega_j^2} \frac{1}{T - T_0}.$$
(B.8)

Hence,

$$u_k(0;T) \to d \frac{\sum_{j \in \alpha(N)} \frac{2a_{kj}}{\omega_j^2} \frac{1}{T - T_0}}{\sum_{j \in \alpha(N)} \frac{2a_j}{\omega_j^2} \frac{1}{T - T_0}} = d \frac{\sum_{j \in \alpha(N)} a_{kj} \omega_j^{-2}}{\sum_{j \in \alpha(N)} a_j \omega_j^{-2}}.$$
(B.9)

As a consequence, $u_k^+(0;T) \to u_k(0;T_0)$ when $T \to T_0$. In the end, we obtain that the initial data of the periodic solution with period *T* converges to the initial data (B.3) of the grazing mode corresponding to T_0 when $T \to T_0$.

The above proposition shows that there exists a family of 1-IPP orbits with period $T \notin \bigcup_{j=1}^{N} T_j \mathbb{N}$, T is near a linear period or a subharmonic $T_0 \in \bigcap_{j \in \alpha(N)} T_j \mathbb{N}$ and these 1-IPP stay near the new linear grazing periodic solutions. Numerical examples for two-dof and three-dof systems are given in Figures (B.1) and (B.2).



FIGURE B.1 – A 2-dof system when $T_1 = 2T_2$: (a) The 1-IPP solutions are not near to the LGO; (b) they are near to the new periodic solution with grazing.



FIGURE B.2 – A 3-dof system when T near T_1 : (c) the associated 1-IPPs are not near the usual linear grazing mode with fundamental period T_1 ; (d) the associated 1-IPPs are near the grazing orbit with initial data (B.3).

Contents

Abstract								
Acknowledgments Table of contents Introduction								
					1	On	solutions to vibro-impact oscillators	5
						1.1	Introduction	5
	1.2	On the existence	7					
	1.3	On the uniqueness	9					
2	Peri	odic solutions with sticking phases of a 2-dof oscillator	13					
	2.1	Introduction	13					
	2.2	Main results	15					
		2.2.1 Sticking phase	15					
		2.2.2 Periodic solutions with one sticking phase per period (1-SPP)	18					
	2.3	Examples	21					
	2.4	Sticking contact	23					
		2.4.1 Occurrence of a sticking phase	24					
		2.4.2 Sticking system	26					
		2.4.3 1.5-singularity at the sticking point	27					
	2.5	Building 1-SPP	29					
		2.5.1 Initial data	29					
		2.5.2 Symmetry	33					
	2 (2.5.3 Relationship between 1-SPP and 1-IPP	34					
	2.6	The countable set Z^{\perp}	36					
		2.6.1 Z^{\perp} is infinite with no resonance	36					
	27	2.6.2 Internal resonances	39					
	2.7	Prestressed structure	40					
		2.7.1 Strictly prestressed structure	41					
	20	2.7.2 Statically grazing system	42					
	2.8	A general two-degree-of-freedom vibro-impact system	44 15					
	2.9	On 1-SPF of <i>N</i> -degree-of-freedom vibro-impact systems with $N > 2$	45 15					
		2.9.1 The occurrence of the sucking phase	45					
		2.9.2 A 5-dol chain with a non prestressed gap $a > 0$	40					

	2.10	Conclusion	47		
3	On t	he first return time of N-dof vibro-impact oscillators	49		
	3.1	Introduction	49		
	3.2	Main results	51		
		3.2.1 Poincaré section	51		
		3.2.2 The square-root singularity	53		
		3.2.3 Dynamics near a linear grazing orbit	56		
	3.3	Domain of definition of the first return map	58		
		3.3.1 Contact times	58		
		3.3.2 Zero, one or infinity?	60		
		3.3.3 Poincaré section	63		
	3.4	Implicit function theorem and power-root singularity	65		
		3.4.1 The square-root singularity	66		
		3.4.2 Power-root singularity	71		
		3.4.3 Discontinuous first return time	79		
	3.5	The square-root instability	80		
		3.5.1 A nonlinear <i>n</i> dimensional map	81		
		3.5.2 Two-dimensional maps with critical instability	84		
	3.6	Dynamics in the vicinity of the grazing orbits	86		
		3.6.1 The first return time	86		
		3.6.2 Square-root dynamics in the vicinity of a linear grazing orbit (LGO)	90		
Co	onclus	sions	93		
Bi	bliog	raphy	95		
A	Powe	er-root singularity for a mass-spring chain	99		
B	B 1-IPP dynamics with internal resonances near grazing orbits				
Co	Contents				

On some periodic solutions of discrete vibro-impact oscillators with a unique unilateral contact condition

Abstract

The purpose of this thesis is to investigate *N* degree-of-freedom vibro-impact oscillators with an unilateral contact. The dynamics is linear in the absence of contact; it is governed by an impact law otherwise. Trajectories that display a sticking phase are identified. The First Return Map is a fundamental tool to explore such periodic solutions. The Poincaré section is tangent to grazing orbits and thus yields the well-known square-root singularity, as already reported in Mechanics, which is here revisited in a rigorous mathematical framework. Another important singularity is exhibited: the discontinuity of the First Return Time. Finally, the square-root dynamics near the linear grazing orbits which may lead to the instability of these linear grazing orbits is studied. It is found that the square-root dynamics emerges from the square-root singularity of the First Return Time if one of the coefficients related to a linear map does not vanish. Under a generic condition of the matrix of eigenvectors, the square-root dynamics near a linear grazing orbit is proven to exist.

Keywords: nonsmooth analysis, vibro-impact systems, unilateral contact, periodic solutions, sticking phase, linear grazing orbit, poincaré map

Sur des solutions périodiques d'oscillateurs discrets à vibro-impact avec une unique condition de contact unilatéral

Résumé

Le but de cette thèse est d'étudier des oscillateurs à vibro-impact à *N* degrés de liberté avec une condition de contact unilatéral. La dynamique est linéaire en l'absence de contact ; elle est régie par une loi d'impact autrement. Des trajectoires présentant une phase de contact collant sont identifiées. L'application de premier retour de Poincaré est un outil fondamental pour étudier la dynamique près de solutions périodiques. La section de Poincaré est tangente aux orbites rasantes et conduit à une singularité en « racine carrée », déjà connue en Mécanique, singularité revisitée dans un cadre mathématique rigoureux. Elle implique la discontinuité du temps de premier retour. Enfin, la dynamique en racine carrée près des modes linéaires rasants, qui peut conduire à l'instabilité de ces modes, est abordée. On constate que la dynamique en racine carrée émerge de la singularité en racine carrée du temps de premier retour si l'un des coefficients liés à une carte linéaire ne s'annule pas. Pour une condition générique de la matrice des vecteurs propres, la dynamique en racine carrée près d'une mode linéaire rasants est prouvée.

Mots clés : analyse non lisse, systèmes discrets à vibro-impact, contact unilatéral, solutions périodiques, mode linéaire rasant, l'application de poincaré